Methods and Protocols in Food Science

## **Springer Protocols**

## José Manuel Lorenzo · Rubén Domínguez Mirian Pateiro · Paulo E. S. Munekata *Editors*

# Methods to Assess the Quality of Meat Products



### Methods and Protocols in Food Science

Series Editor Anderson Sant'Ana University of Campinas Campinas, Brazil

For further volumes: http://www.springer.com/series/16556 Methods and Protocols in Food Science series is devoted to the publication of research protocols and methodologies in all fields of food science.

Volumes and chapters will be organized by field and presented in such way that the readers will be able to reproduce the experiments in a step-by-step style. Each protocol will be characterized by a brief introductory section, followed by a short aims section, in which the precise purpose of the protocol will be clarified.

More information about this series at https://link.springer.com/bookseries/16556

## Methods to Assess the Quality of Meat Products

Edited by

## José Manuel Lorenzo

Centro Tecnológico de la Carne de Galicia, Ourense, Spain; Universidade de Vigo, Área deTecnoloxía dos Alimentos, Facultade de Ciencias, Ourense, Spain

## Rubén Domínguez

Centro Tecnológico de la Carne de Galicia, Ourense, Spain

## **Mirian Pateiro**

Centro Tecnológico de la Carne de Galicia, Ourense, Spain

## Paulo E. S. Munekata

Centro Tecnológico de la Carne de Galicia, Ourense, Spain



*Editors* José Manuel Lorenzo Centro Tecnológico de la Carne de Galicia Ourense, Spain

Universidade de Vigo, Área deTecnoloxía dos Alimentos, Facultade de Ciencias Ourense, Spain

Mirian Pateiro Centro Tecnológico de la Carne de Galicia Ourense, Spain Rubén Domínguez Centro Tecnológico de la Carne de Galicia Ourense, Spain

Paulo E. S. Munekata Centro Tecnológico de la Carne de Galicia Ourense, Spain

ISSN 2662-950X ISSN 2662-9518 (electronic) Methods and Protocols in Food Science ISBN 978-1-0716-2001-4 ISBN 978-1-0716-2002-1 (eBook) https://doi.org/10.1007/978-1-0716-2002-1

© The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

This work is subject to copyright. All rights are solely and exclusively licensed by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, expressed or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

This Humana imprint is published by the registered company Springer Science+Business Media, LLC, part of Springer Nature.

The registered company address is: 1 New York Plaza, New York, NY 10004, U.S.A.

#### Preface

The meat industry is a sector with great importance worldwide. However, current consumer trends as well as regulatory changes make essential the determination of the constituents and the additives. The different analytical techniques that can be applied arise from the need to provide correct nutritional information in accordance with the legislation (labeling), as well as to determine the nutritional quality and chemical composition of meat and meat products. On the other hand, the physicochemical and degradative changes that can take place can also be a source of significant economic losses or even lead to the appearance of toxic substances, which would put the health of the consumer at risk. Therefore, the determination of these changes should be also carried out.

This book arises from the collaboration between different institutions and centers at an international level, which are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568). Therefore, the publication of this book is a very useful tool in order to standardize the analytical processes, as well as to serve as a basis for the development and validation of these techniques in those research groups that do not have them implemented.

Several analytical techniques have been developed for each of the determinations described in this book. However, the variations between them as well as the use of different techniques could lead to the results obtained not being reproducible or comparable. With this in mind, the current book examines the techniques used for the analysis of meat and meat products. A complete and comprehensive description of the methods and materials used is made so that each technique can be reproduced by other researchers. In addition, the techniques include explanatory notes and elucidate the possible points to take into account for the correct achievement of the determination. The ultimate goal is to support the scientific community, professionals, laboratories, food companies, and regulatory bodies in their aim to identify and measure meat and meat products composition and properties. The book consists of 14 chapters.

Chapter 1 provides the complete material and methods description for the determination of the different chemical parameters, including moisture, fat, protein, and ash. In addition to being mandatory to include part of this information on the products labeling, these determinations also offer nutritional information on the final product.

Chapter 2 discusses the techniques used for the measurement of pH and color. In this case, color is considered one of the most important parameters that influence consumer acceptance. Thus, the correct determination of physical (CIELab) and also chemical (haem pigments) color is important to control the meat and meat products quality.

Not only color but also texture is important in consumer preferences. With this in mind, Chapter 3 provides different procedures for the correct determination of texture parameters in different meat and meat products.

Chapter 4 includes the complete method for fat extraction, transesterification, and determination of fatty acids with the aim to have complete nutritional information. This is important to quantify saturated fatty acids (necessary for products labeling) and also for nutritional claims (e.g., source of omega-3, low saturated fats, etc.). Additionally, the fatty acid determination also provides information about the lipid stability of the product.

In Chapter 5, a complete description of free amino acids and hydrolyzed amino acids extraction is explained, followed by the derivatization and chromatographic analysis. The

free amino acids content is related to the proteolytic phenomena during meat products manufacturing and processing, while hydrolyzed amino acids indicate the nutritional quality of meat and meat products (e.g., essential amino acids).

Chapter 6 deals with the analysis of total cholesterol in meats. It is well known that cholesterol consumption is related to the development of some human diseases. Thus, the correct determination of this compound provides both the consumer and the manufacturer with important information.

Similarly, the content of several minerals is important to human health. Thus, in Chapter 7, a comprehensive explanation about the sample treatment and mineral composition analysis using IPC-OES is provided.

Chapter 8 focuses on the analytical determination of nitrites and nitrates in meat and meat products, using liquid chromatography. These additives are necessary for several meat products manufacture since they control the growth of some pathogenic microorganisms and stabilize the color. However, they are also related to the development of toxic molecules; thus the residual content of both is regulated.

The release of biogenic amines during processing, mostly in fermented or ripened products, is also frequent. These compounds are related to some allergenic reactions and are harmful for humans. Therefore, Chapter 9 provides a complete description of the materials and methods for their determination using liquid chromatography.

On the other hand, it is well known that oxidative processes are the most important degradative reactions during both the processing and storage of meat and meat products. They produce loss of nutrients, reduce sensory properties, develop dangerous compounds, and decrease their shelf life. With this in mind, the analysis of protein carbonyls using a spectrophotometric technique for the determination of protein oxidation is completely explained in Chapter 10, while comprehensive procedures for the determination of primary and secondary lipid oxidation products are described in Chapter 11.

Volatile compounds are vital for the correct development of the typical aroma of meat and meat products. Additionally, the release of some specific compounds allows the monitoring of degradative processes. Therefore, the analysis of these compounds is important to monitor the sensory quality and the manufacturing process. Chapter 12 provides a complete description of the analysis of volatile compounds using a solvent-free technique, based on solid-phase microextraction with gas chromatography-mass spectrometry determination.

Chapter 13 provides a brief overview of specific issues in the meat industry, which may be resolved using the current proteomics-based methodology. The quantification, analysis, and curation of proteomics data are explained in this chapter.

Finally, the antioxidant activity of meat and meat products could exert a protective effect against oxidative degradation. Therefore, Chapter 14 explains in a comprehensive way the main techniques used for the determination of antioxidant capacity in meat and meat products.

Ourense, Spain

José Manuel Lorenzo Rubén Domínguez Mirian Pateiro Paulo E. S. Munekata

#### **Series Preface**

The *Methods and Protocols in Food Science* series is devoted to the publication of research protocols and methodologies in all fields of food science. The series is unique as it includes protocols developed, validated, and used by food and related scientists as well as theoretical bases are provided for each protocol. Aspects related to improvements, adaptations, and further developments in the protocols may also be approached.

The *Methods and Protocols in Food Science* series aims to bring the most recent developments in research protocols in the field as well as very well-established methods. As such the series targets undergraduate, graduate, and researchers in the field of food science and correlated areas. The protocols documented in the series will be highly useful for scientific inquiries in the field of food sciences, presented in such a way that the reader will be able to reproduce the experiments in a step-by-step style.

Each protocol will be characterized by a brief introductory section, followed by a short aims section, in which the precise purpose of the protocol is clarified. Then, an in-depth list of materials and reagents required for employing the protocol is presented, followed by comprehensive and step-by-step procedures on how to perform that experiment. The next section brings the dos and don'ts when carrying out the protocol, followed by the main pitfalls faced and how to troubleshoot them. Finally, template results will be presented and their meaning/conclusions addressed.

The *Methods and Protocols in Food Science* series will fill an important gap, addressing a common complaint of food scientists, regarding the difficulties in repeating experiments detailed in scientific papers. With this, the series has a potential to become a reference material in food science laboratories of research centers and universities throughout the world.

Campinas, São Paulo, Brazil

Anderson S. Sant'Ana

### **Contents**

Prej Seri	face	v vii
Con	ntributors	xi
1	Chemical Composition Roberto Bermúdez, Noemí Echegaray, Maria João Fraqueza, Mirian Pateiro, Rubén Domínguez, Paulo E. S. Munekata, José Manuel Lorenzo, and Eva María Santos	1
2	pH and Color Alfredo Teixeira, Rubén Domínguez, Javier F. Rey, Gonzalo Aleu, Mirian Pateiro, and José Manuel Lorenzo	17
3	Texture Analysis Noemí Echegaray, Marcelo Rosmini, Mirian Pateiro, Rubén Domínguez, Paulo E. S. Munekata, José Manuel Lorenzo, Eva María Santos, and Roberto Bermúdez	29
4	Fatty Acids Rubén Domínguez, Laura Purriños, Mirian Pateiro, Paulo C. B. Campagnol, Jorge Felipe Reyes, Paulo E. S. Munekata, and José Manuel Lorenzo	41
5	Amino Acids (Free and Hydrolyzed) Olalla López-Fernández, Rubén Domínguez, Mirian Pateiro, Silvina C. Andrés, Paulo E. S. Munekata, Laura Purriños, José Manuel Lorenzo, and Marco Antonio Trindade	53
6	Cholesterol. Olalla López-Fernández, Rubén Domínguez, Claudia Ruiz-Capillas, Mirian Pateiro, María Elena Sosa-Morales, Paulo E. S. Munekata, Anderson S. Sant'Ana, José Manuel Lorenzo, and Ana M. Herrero	65
7	Mineral Profile. Laura Cutillas, Olalla López-Fernández, Rubén Domínguez, Mirian Pateiro, Adriana Pazos, Paulo E. S. Munekata, Laura Purriños, José Manuel Lorenzo, Andrea Carla da Silva Barretto, and José Angel Pérez-Álvarez	73
8	Nitrate and Nitrite Elisa Rafaela Bonadio Bellucci, Camila Vespúcio Bis Souza, José Manuel Lorenzo, Gonzalo Aleu, Alfredo Teixeira, Rubén Domínguez, and Andrea Carla da Silva-Barretto	85
9	Biogenic Amines Claudia Ruiz-Capillas, Mehdi Triki, Rubén Domínguez, José Manuel Lorenzo, and Ana M. Herrero	97
10	Spectrophotometric Analysis of Protein Carbonyls Francesca Soglia, Giulia Baldi, Alberto González-Mohino, Silvia Díaz-Velasco, Massimiliano Petracci, and Mario Estévez	107

x Contents

11	Lipid Oxidation (Primary and Secondary Products) Yasmim S. V. Leães, José Manuel Lorenzo, Alexandre J. Cichoski, Roger Wagner, Eva María Santos, Jorge F. Reyes, and Paulo C. B. Campagnol	115
12	Volatile Organic Compound Profile Olalla López-Fernández, Rubén Domínguez, Laura Cutillas, Paulo E. S. Munekata, Laura Purriños, José Manuel Lorenzo, Nestor Sepúlveda, Alfredo Teixeira, and Mirian Pateiro	133
13	Proteomics María López-Pedrouso, José Manuel Lorenzo, and Daniel Franco	141
14	Antioxidant Capacity Noemí Echegaray, Roberto Bermúdez, Gema Nieto, Rubén Domínguez, Mirian Pateiro, Nestor Sepúlveda, Marco Antonio Trindade, and José Manuel Lorenzo	153
Ind	ex	169

#### Contributors

- GONZALO ALEU Faculty of Agricultural Sciences, Catholic University of Córdoba, Cordoba, Argentina
- SILVINA C. ANDRÉS Centro de Investigación y Desarrollo en Criotecnología de Alimentos (CIDCA, CONICET-CICPBA-UNLP), Facultad de Ciencias Exactas, UNLP, La Plata, Argentina
- GIULIA BALDI Department of Agricultural and Food Sciences, Alma Mater Studiorum-University of Bologna, Cesena, Italy
- ELISA RAFAELA BONADIO BELLUCCI Department of Food Technology and Engineering (DETA), São Paulo State University (UNESP), São Paulo, Brazil
- ROBERTO BERMÚDEZ · Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- PAULO C. B. CAMPAGNOL Department of Food Science and Technology (DTCA), Universidade Federal de Santa Maria, Santa Maria, Rio Grande do Sul, Brazil

ALEXANDRE J. CICHOSKI • Department of Food Science and Technology (DTCA), Universidade Federal de Santa Maria, Santa Maria, Rio Grande do Sul, Brazil

- Laura Cutillas Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- ANDREA CARLA DA SILVA-BARRETTO Institute of Biosciences, Humanities and Exact Sciences – IBILCE, Department of Food Technology and Engineering, São Paulo State University (UNESP), São Paulo, Brazil
- SILVIA DÍAZ-VELASCO IPROCAR Research Institute, Universidad de Extremadura, Cáceres, Spain
- RUBÉN DOMÍNGUEZ Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- NOEMÍ ECHEGARAY . Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- MARIO ESTÉVEZ IPROCAR Research Institute, Universidad de Extremadura, Cáceres, Spain
- DANIEL FRANCO Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- MARIA JOÃO FRAQUEZA · Faculty of Veterinary Medicine, CIISA, Centre for
- Interdisciplinary Research in Animal Health, University of Lisbon, Lisbon, Portugal
- Alberto González-Mohino IPROCAR Research Institute, Universidad de Extremadura, Cáceres, Spain
- ANA M. HERRERO Institute of Food Science, Technology and Nutrition (ICTAN-CSIC), Madrid, Spain
- YASMIM S. V. LEÃES Department of Food Science and Technology (DTCA), Universidade Federal de Santa Maria, Rio Grande do Sul, Brazil
- Olalla López-Fernández Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- MARÍA LÓPEZ-PEDROUSO Departamento de Zooloxía, Xenética e Antropoloxía Física, Universidade de Santiago de Compostela, Santiago de Compostela, Spain
- JOSÉ M. LORENZO Centro Tecnológico de la Carne de Galicia, Ourense, Spain; Universidad de Vigo, Área de Tecnoloxía dos Alimentos, Facultade de Ciencias, Ourense, Spain
- PAULO E. S. MUNEKATA Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- GEMA NIETO Department of Food Technology, Nutrition and Food Science, Veterinary Faculty, University of Murcia, Espinardo, Spain
- MIRIAN PATEIRO Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- Adriana Pazos Instituto Tecnología de Alimentos, INTA, Buenos Aires, Argentina

- JOSÉ ANGEL PÉREZ-ÁLVAREZ IPOA Research Group, Agro-Food Technology Department, Centro de Investigación e Innovación Agroalimentaria y Agroambiental (CIAGRO-UMH), Miguel Hernández University, Alicante, Spain
- MASSIMILIANO PETRACCI Department of Agricultural and Food Sciences, Alma Mater Studiorum-University of Bologna, Cesena, Italy
- LAURA PURRIÑOS Centro Tecnológico de la Carne de Galicia, Ourense, Spain
- JAVIER F. REY Facultad de Ingeniería, Programa de Ingeniería de Alimentos, Universidad de La Salle, Bogotá D.C., Colombia
- JORGE FELIPE REYES Department of Agricultural Sciences and Food, Universidad Técnica Particular de Loja, Loja, Ecuador; Departamento de Química, Universidad Técnica Particular de Loja, Loja, Ecuador
- MARCELO ROSMINI Department of Public Health, Faculty of Veterinary Science, National University of Litoral, Esperanza, Argentina
- CLAUDIA RUIZ-CAPILLAS Institute of Food Science, Technology and Nutrition (ICTAN-CSIC), Madrid, Spain

ANDERSON S. SANT'ANA • Department of Food Science, Faculty of Food Engineering, University of Campinas, São Paulo, Brazil

- Eva María Santos Universidad Autónoma del Estado de Hidalgo, Área Académica de Química, Tulancingo, Hidalgo, Mexico
- NESTOR SEPÚLVEDA Laboratorio de Producción Animal, Facultad de Ciencias Agropecuarias y Forestales, Universidad de La Frontera, Temuco, Chile
- FRANCESCA SOGLIA Department of Agricultural and Food Sciences, Alma Mater Studiorum-University of Bologna, Cesena, Italy
- María Elena Sosa-Morales División de Ciencias de la Vida, Departamento de Alimentos, Campus Irapuato-Salamanca, Universidad de Guanajuato, Guanajuato, Mexico
- CAMILA VESPÚCIO BIS SOUZA Department of Food Technology and Engineering (DETA), São Paulo State University (UNESP), São Paulo, Brazil
- Alfredo Teixeira Escola Superior Agrária, Instituto Politécnico de Bragança Centro de Investigação de Montanha (CIMO), Bragança, Portugal
- MEHDI TRIKI . Ministry of Public Health, Doha, Qatar
- MARCO ANTONIO TRINDADE Departamento de Engenharia de Alimentos, Faculdade de Zootecnia e Engenharia de Alimentos, Universidade de São Paulo, São Paulo, SP, Brazil
- ROGER WAGNER Department of Food Science and Technology (DTCA), Universidade Federal de Santa Maria, Rio Grande do Sul, Brazil



## **Chapter 1**

#### **Chemical Composition**

#### Roberto Bermúdez, Noemí Echegaray, Maria João Fraqueza, Mirian Pateiro, Rubén Domínguez, Paulo E. S. Munekata, José Manuel Lorenzo, and Eva María Santos

#### Abstract

Meat composition influences the nutritional and sensory quality of meat products, therefore elucidating its composition is essential for the production chain. Moisture, fat, protein, and ash are the main components of these products whose determinations have been widely studied for many years. That is why today traditional methods, new techniques, and a host of smart equipment can be used, which could affect the results obtained and their comparison.

This chapter give a complete analysis guideline of the aforementioned determinations. Moisture determination is carried out by thermogravimetry, according to sample weight loss. Fat and protein contents are determined by Ankom technology and Kjeldahl method, respectively. Ashes are calculated by weight loss once the sample has been calcined in a muffle furnace.

Key words Moisture, Protein, Fat, Ash, Proximate composition, Gravimetric method, Ankom extractor, Hydrolysis, Kjeldahl method, Calcination process

#### 1 Introduction

Chemical composition, also known as proximate composition, is very important in the food industry for the development of new products, quality control, consumer information or regulatory purposes. In this way, it is necessary to know the percentage of moisture, protein, fat, and ash of the products to establish rules that regulate the diet of the people to make it healthier. Therefore, these analyses are very important since they are directly related to the health of consumers, as well as for the food production [1, 2]. All of these parameters could be modified both by changing the diet of the animals or by using new ingredients during processing [3]. The best level of each of them depends on the type of consumer.

Taking into account the aforementioned facts, people have the right to know the nutritional composition of the meat and meat

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products, Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_1, © The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

products that they eat. Therefore, Governments ensure this to their citizens through several rules, such as Regulation (EU) 1169/2011 in the European Union (EU) [4] or CFR (21CFR101) of the Food and Drug Administration (FDA) of the United States of America (USA) [5], among others. So, companies have to label their products according to these regulations. They are many similarities between them, only with slight differences. According to Regulation 1169/2011, it is compulsory to label the percentage of fat, protein, carbohydrates, total sugars, sodium chloride, saturated fatty acids and energy (Kcal and KJ), as well as, all allergens presented in products. In addition, other nutrients can also be labeled, such as dietary fiber, mono and polyunsaturated fatty acids, vitamins, etc. In addition, it is necessary to include on the label important information about the origin, expiration date, etc.

All methods, which we are going to describe below, are supported by international organizations such as the International Organization for Standardization (ISO) or the American Oil Chemists Society (AOCS). Firstly, water occurs as the main constituent of meat and meat products. Therefore, the determination of the water content of meat and meat products is very important, since water is an essential constituent and its interaction with other components condition the shelf life and the quality of the product [6]. Moisture analysis is a gravimetric method based on loss of weight by drying. This method has been used in the same way for many years. Similarly, ashes determination is based on a loss of weight due to the calcination of organic matter in a muffle furnace. Secondly, lipids play an important role in sensory quality of meat and meat products, since it contributes to attributes such as texture (juiciness), the formation of the characteristic flavor of the product, and its final appearance. Therefore, fat determination is one of the required analyses in food labeling and quality control [7]. Fat analysis can be performed with or without hydrolysis, depending on the type of the sample. In many samples, fat is not freely accessible to the solvent and cannot be fully extracted. The bonding of fats in starch-lipid complexes or with lipoproteins or phospholipids, prevents direct extraction of the fat. For this reason, hydrolysis with HCl to prepare samples for fat determination is prescribed by law in a wide range of national and international methods. Fat determination with hydrolysis is known as "Total fat determination," while the method without hydrolysis determines crude fat. In addition, the analysis of fat has changed over the years, specially, regarding the equipment used. In this regard, Soxhlet extractor sometimes has limitations, especially when the fat content is very low, which makes it not very reproducible. Below, an alternative to this apparatus is shown.

Proteins are also very important constituents of meat and meat products, since they have a high nutritional value and their structural and physicochemical characteristics are related to their functional properties [7]. As with fat determination, protein analysis has also changed over the years regarding the equipment used, so numerous methods have been developed. Two steps are necessary, an acid digestion and a titration to obtain the total nitrogen content present in the sample. The application of a correction factor, which is specific for each type of sample, allows to calculate the percentage of protein.

The aim of this chapter is to make a guide for the analysis of the chemical composition of meat and meat products, so that they can compare the results minimizing the error due to the method used.

#### 2 Materials

	Prepare all solutions using distilled water and/or analytical grade reagents. Prepare and store all reagents at room temperature (unless indicated otherwise). Diligently follow all waste disposal regulations when disposing of waste materials ( <i>see</i> <b>Note 1</b> ).
2.1 Sample Preparation for Analysis	<ol> <li>Professional mill KN 295 Knifetec (Foss).</li> <li>Homogeniser HM 294 (Foss).</li> <li>Cutting board.</li> <li>Knife.</li> </ol>
2.2 Moisture Determination	<ol> <li>Drying oven with capacity to dry the samples at 105 ± 5 ° C, mod. UFP 600 (Memmert).</li> <li>Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius).</li> <li>Desiccator equipped with an effective dehydrator (Silica Gel with humidity indicator).</li> <li>Stainless steel capsules of 60 mm diameter and 25 mm high.</li> <li>Glass rod with rounded tip, which goes completely into the capsule.</li> <li>Eppendorf Research Micropipette 500–5000 μL.</li> <li>Washed sea sand, fine grain. General purpose grade.</li> <li>Ethanol 96% for analysis, ACS, Reagent grade.</li> </ol>
2.3 Fat Determination	<ol> <li>Filter Bags, mod. XT4 (Ankom).</li> <li>Indelible marker.</li> <li>Filters support (Ankom).</li> <li>Stainless steel capsules of 60 mm diameter and 25 mm high.</li> <li>Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius).</li> </ol>

- 6. Thermosealer Impulse Sealer, mod. AIE-200.
- 7. Drying oven with capacity to dry the samples at  $105 \pm 5$  ° C, mod. UFP 600 (Memmert).
- 8. Desiccator equipped with an effective dehydrator (Silica Gel with humidity indicator).
- 9. Extraction Unit, mod. XT10 (Ankom).
- 10. Ankom hydrolysis unit, mod. HCl (Ankom).
- 11. Weight to dry the samples after hydrolysis (Blotter), mod. H35 (Ankom).
- 12. Rapid Dryer HCl Filter RD2 (Ankom).
- 13. Support for samples.
- 14. Borosilicate glass graduated cylinder of 250 mL, class A.
- 15. Solvent recovery glass bottle 500 mL.
- 16. Petroleum ether with a boiling point range 40–60 °C, reagent grade, ACS, ISO (*see* Note 2).
- 17. Celite Hyflo Super Cel, RE.
- Hydrochloric acid solution (3 N): Mix 248 mL of hydrochloric acid (37%) with 752 mL of distilled water.

#### 2.4 Protein Determination

- 1. Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius).
  - 2. Precision balance with a resolution of 0.01 g, mod. TE 612 (Sartorius).
  - 3. Temperature Controller, mod. TZ (Gerhardt).
  - 4. Nitrogen Digestor Kjeldatherm, mod. KB20 (Gerhardt).
  - 5. Nitrogen Distiller, mod. VAPODEST 50 carrousel (Gerhardt).
  - 6. Kjeldatherm digestion tubes, 250 mL (Gerhardt).
  - 7. Metal tube rack (Gerhardt).
  - 8. Filter paper  $5 \times 8$  cm approximately.
  - 9. 1000 mL Glass Beaker.
- 10. 1000 mL volumetric flask, Class A.
- 11. Liquid dispenser with a resolution of 0.5 mL and capable of dispensing in a range of 2.5–25 mL, mod. Genius 1,605,006 (Walu).
- 12. 500 mL washing bottle.
- 13. Distilled water.
- 14. Sulfuric acid 96% for analysis, ISO (see Note 2).
- 15. Universal indicator of pH, solution for volumetric analysis.
- 16. Kjeldahl catalyst tablets: Kjeltabs 5 g K<sub>2</sub>SO<sub>4</sub> + 0.5 g CuSO<sub>4</sub> ×  $5H_2O$  (Gerhardt).

- 17. Boric acid 2%. Dissolve 20 g of boric acid in 1 L of distilled water.
- 18. Sodium hydroxide solution 32% for Kjeldahl.
- 19. Sodium hydroxide pellets, ACS, reagent grade.
- 20. Sodium hydroxide 15%. Dissolve 180 g of sodium hydroxide pellets in 1.2 L of distilled water. Add several drops of universal pH indicator. It will be valid until the indicator turns from blue to yellow (*see* **Note 3**).
- 21. Hydrochloric acid 0.1 N PA, quality titrated solution.
- 22. Buffer solution pH 7.00  $\pm$  0.02.
- 23. Buffer solution pH 4.00  $\pm$  0.02.
- 1. Muffle furnace, mod. RWF12–13 (Carbolite).
- 2. Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius).
- **3**. Desiccator equipped with an effective dehydrator (Silica Gel with humidity indicator).
- 4. Socorex ACURA 825 Micropipette 100–1000 μL.
- 5. Porcelain crucibles of 25 mL, glazed inside and out, except base. Certificate according to DIN 12904. Withstands maximum temperature of 1000 °C.
- 6. Suitable forceps for handling porcelain crucibles.
- 7. Heat resistant gloves (see Note 4).
- 8. Drying oven with capacity to dry the samples at  $105 \pm 5$  ° C, mod. UFP 600 (Memmert).

#### 3 Methods

Carry out all procedures at room temperature unless otherwise specified.

3.1 Sample	The first step is to chop the sample into small cubes and homoge-
Preparation for	nize it correctly [8]. Both chopping and homogenization are car-
Analysis	ried out with a professional mill KN 295 Knifetec (Foss) or a Homogeniser HM 294 (Foss) depending on the type/size of the sample. This step is very important in order to achieve good and representative results.

3.2 Moisture Moisture content is determined by oven drying (Memmert UFP 600, Schwabach, Germany) at 105 °C until constant weight, and is calculated as sample weight loss, according to ISO 1442 (1997)
 [9]. Figure 1 shows in a simplified and schematic way the steps of the moisture determination.

2.5 Ash Determination





- 1. Firstly, prepare the stainless steel capsules with sea sand and a glass rod (*see* **Note 5**). Then, the set of capsules, sea sand, and glass rod must be dried for at least 1 h in the oven (set at  $105 \text{ }^{\circ}\text{C} \pm 2 \text{ }^{\circ}\text{C}$ ).
- 2. Before weighing the capsules, sea sand and glass rod must be at room temperature. To do this, they need to be placed in a desiccator (containing an efficient desiccant, such as silica gel) for 1 h.
- 3. Next, weigh the set (capsule, sea sand, and glass rod) on an analytical balance, with an accuracy of 0.0001 g.
- 4. Weigh 5 g of minced and homogenized sample into the capsule which has sea sand and a glass rod.
- 5. Mix the contents of the capsule (sample + sea sand) with the help of the glass rod (*see* **Note 6**).
- 6. Place the capsule with its content (sea sand + sample + ethanol) in the oven (set at 105 °C  $\pm$  2 °C) for at least 16 h (until reaching constant weight).
- 7. After the necessary time, remove the capsules with its contents from the oven and place them in the desiccator for 1 h (until reaching room temperature).
- 8. Finally, weigh the capsules and its content on an analytical balance, with an accuracy of 0.0001 g.
- 9. Expression of results
  - (a) Calculate the moisture content, as a percentage by mass, using the following equation:

%Moisture = 
$$\left[\frac{(W_1 - W_2)}{(W_1 - W_0)}\right] \times 100$$

where:

- $W_0$  is the capsule, glass rod, and sea sand weigh (g).
- *W*<sub>1</sub> is the capsule containing the sample, glass rod, and sea sand weigh before drying (g).
- $W_2$  is the capsule containing the sample, glass rod, and sea sand weigh after drying (g).



Fig. 2 Schematic and simplified representation of the steps of the fat determination without hydrolysis

**3.3 Fat Determination**Fat content is obtained by weight difference. It is carried out with or without hydrolysis, depending on the type of the sample (*see*  **Note** 7). We performed the analysis according to AOCS Am5-04 (2005) [10].

*3.3.1 Total Fat Extraction* Figure 2 shows in a simplified and schematic way the steps of the fat determination without hydrolysis.

- 1. Label the filter using a permanent marker. Place the filter holder on the analytical balance and tare. Place the previously labeled filter on the holder so that the filter remains open to allow the sample to enter in it. Weigh the filter on the analytical balance with an accuracy of 0.0001 g, note the value obtained ("Filter weight (g)") on the laboratory sheet and tare again.
- 2. Weigh 1 g of sample into the filter and record on the laboratory sheet ("Sample weight  $W_1$  (g)"). Take the filter and seal the open part with the heat sealer about 2 mm from the end of the filter (*see* Note 8).
- 3. Place the sealed filter in a stainless steel capsule, recording the number and weight of it previously, with an accuracy of 0.0001 g ("No./Weight Ankom capsule (g)"), on the worksheet (*see* Note 9).
- 4. Bring the samples to the oven for drying at  $105 \pm 5$  °C for at least 16 h. Record on the worksheet. After drying, place the samples in a desiccator for 1 h and then, weigh them on the analytical balance with an accuracy of 0.0001 g ("Dry sample weight 1  $W_2$  (g)").
- 5. Then take the samples to the Ankom XT10 extractor and place them in the sample holder (maximum 15 samples). In a fume hood, pour 200 mL of petroleum ether, measured with a measuring cylinder, into the metal beaker of the extractor, place the plastic upper part and the samples inside it and pour 150 mL of solvent over them, and close the equipment and press start. You can see the extraction conditions in Table 1 (*see* **Note 10**).

## Table 1 Ankom fat extractor working conditions

Fat extractor working conditions	
Extraction time (min)	60
Solvent recovery time (min)	10
Extraction temperature (°C)	90



Fig. 3 Schematic and simplified representation of the steps of the fat determination with hydrolysis

6. Finally, remove the samples from the equipment, and place them in the oven under the same conditions as during drying  $(105 \pm 5 \,^{\circ}\text{C})$  for 1 h, to eliminate the residual solvent. After this time, place the samples in a desiccator for 1 h, and weigh them on an analytical balance with an accuracy of 0.0001 g ("Dry sample weight 2  $W_3$  (g)") (see Note 11).

*3.3.2 Total Fat Extraction* Figure 3 shows in a simplified and schematic way the steps of the fat determination with hydrolysis.

- 1. Label the filter using a permanent marker. Place the filter holder on the analytical balance and tare. Place the previously labeled filter on the holder so that the filter remains open to allow the sample to enter in it. Weigh the filter on the analytical balance with an accuracy of 0.0001 g, note the value obtained ("Filter weight (g)") on the laboratory sheet and tare again.
- 2. Weigh 0.5 g of celite onto the filter, tare and weigh 0.5 g of sample onto the filter (inside of the celite) with an accuracy of 0.0001 g, and record both on the procedure test sheet ("Sample weight  $W_1$  (g)") (*see* Note 12). Take the filter and seal the open part with the heat sealer about 2 mm from the end of the filter (*see* Notes 8 and 13).
- 3. Place the samples in the HCl Hydrolysis System equipment to carry out the hydrolysis process. To do this, put the samples in the corresponding support and the set, samples-support, in the equipment vessel. We fill it with 500 mL of 3 N HCl. You can see the hydrolysis conditions in Table 2 (*see* Note 10).

Table 2				
Ankom h	nydrolysis	system	working	conditions

Hydrolysis system working conditions	
Hydrolysis time (min)	60
Hydrolysis temperature (°C)	90
Rinse time (min)	20

- 4. Once the cycle is finished (1.20 h, approx.), we turn off both the water and the equipment. Remove the samples from the equipment, place them between absorbent paper, and place the blotter on top to dry them for 2 min. We repeat this step one more time.
- 5. Then, we transfer the samples to the rapid drying equipment (Rapid Dryer), and we keep them in it for 3 h at  $105 \pm 5$  °C. Once finished, place the samples in a desiccator for 1 h and weighed them on an analytical balance with an accuracy of 0.0001 g, recording this value ("Dry sample weight 1  $W_2$  (g)") on the worksheet.
- 6. Then, take the samples to the Ankom XT10 extractor and place them in the sample holder (maximum 15 samples). In a fume hood, pour 200 mL of petroleum ether, measured with a measuring cylinder, into the metal beaker of the extractor, place the plastic upper part and the samples inside it and pour 150 mL of solvent over them, and close the equipment and press start. You can see the extraction conditions in Table 1 (*see* **Note 10**).
- 7. Finally, remove the samples from the equipment, and place them in the oven under the same conditions as during drying  $(105 \pm 5 \,^{\circ}\text{C})$  for 1 h, to eliminate the residual solvent. After this time, place the samples in a desiccator for 1 h, and weigh them on an analytical balance with an accuracy of 0.0001 g ("Dry sample weight 2  $W_3$  (g)") (see Note 11).
- 8. Expression of results
  - (a) The results for the samples made without hydrolysis are expressed as a percentage of fat following the formula:

%Fat (without hydrolysis) = 
$$\left[\frac{((W_2 - C) - W_3)}{W_1}\right] \times 100$$

where:

- *W*<sub>1</sub> is the sample weight (g).
- W<sub>2</sub> is the sample, capsule, and filter weight before drying (g).

- $W_3$  is the sample and filter weight after extraction (g).
- *C* is the capsule weight (g).
- (b) The results for the samples made with hydrolysis are expressed as a percentage of fat following the formula:

%Fat(with hydrolysis) = 
$$\left[\frac{(W_2 - (W_3 + (B_1 - B_2)))}{W_1}\right] \times 100$$

where:

- $W_1$  is the sample weight (g).
- $W_2$  is the dry sample + filter + celite weight after hydrolysis (g).
- $W_3$  is the dry sample + filter + celite weight after extraction (g).
- $B_1 =$  "Blank" sample weight after hydrolysis (g).
- $B_2 =$  "Blank" sample weight after extraction (g).

3.4 ProteinProtein determination is carried out into two phases according toDeterminationISO 937 (1978) [11]. First a digestion of the sample and second a<br/>distillation, and subsequent potentiometric titration. Figure 4<br/>shows in a simplified and schematic way the steps of the protein<br/>determination.

3.4.1 Digestion1. Weigh the sample on a filter paper on the analytical balance with a precision of 0.1 mg, and take it to a digestion tube that will be supported by the metal rack designed for this purpose (*see* Note 14).

- 2. Depending on the type of samples, proceed as follows:
  - (a) For dehydrated samples:  $0.5 \text{ g} \pm 0.1 \text{ g}$  of sample must be weighed on filter paper and placed in the digestion tube.
  - (b) For fresh samples: 1.0 g  $\pm$  0.2 g of sample should be weighed on filter paper and introduced into the digestion tube.



Homogenize the sample. Weigh 1 g or 0.5 g depending on the sample



Distillation and titration are done automatically with 0.1 N HCl

The determination of the endpoint is done with a pH-meter



Table 3Summary of the digestion programs

Program	Step	Time	Temperature (°C)
1	1	15 min	0–400
	2	1 h 30 min	400
2	1	15 min	0–200
	2	20 min	200
	3	15 min	200–400
	4	2 h	400

## Table 4 Washing program characteristics of distillation equipment

Step	Value
H <sub>2</sub> O addition	120 mL
NaOH addition	0 mL
Distillation time	7 min
Esteam generator	100%
Sample suction	25 s
Titration vessel suction	25 s
H <sub>2</sub> BO <sub>3</sub> addition	80 mL

- 3. Add two catalyst tablets to each tube and 20 mL of 96% sulfuric acid and place the rack with the tubes in the digestor. Then, you must select the digestion program (Table 3; *see* **Note 15**). Once digestion is complete, allow tubes to cool for 30 min.
- 1. While digestion is taking place, we must wash the distillation equipment. For that, we place three tubes with distillated water (up to half of the tube) in the "Vapodest" and start the washing program in the software (Table 4). Previously, we must calibrate the pH electrode with buffers 4 and 7.

3.4.2 Distillation and

Titration

- 2. Once the digestion is finished, we take the samples to the distillation equipment (*see* **Note 16**) and we place the samples on the carousel in the corresponding order. In this moment, we must record the sample weight with its name in the software. Then, we start the sample program (Table 5; *see* **Note 17**).
- 3. At the end of the distillation and titration, the equipment gives us the total nitrogen content of the sample. In addition, the software gives us the total protein content, obtained by multiplying the nitrogen percentage by a conversion factor that depends on the type of sample (Table 6) [12].

Table 5Samples program characteristics of distillation equipment

Step	Value
H <sub>2</sub> O addition	100 mL
NaOH addition	80 mL
Distillation time	4 min
Esteam generator	100%
Sample suction	25 s
Titration vessel suction	25 s
H <sub>2</sub> BO <sub>3</sub> addition	80 mL
Conversion factor	6.25

## Table 6Nitrogen to protein conversion factors

Product	Conversion factor
Rye, barley, oats	5.83
Flour, whole wheat flour	5.83
Flour, medium or low extraction; wheat pasta	5.70
Rice	5.95
Soy *, seeds	5.71 (6.25) *
Chestnut, coconut	5.30
Milk	6.38
Other foods	6.25

4. Expression of results

(a) The results for the samples are expressed as a percentage of protein following the formula:

%Nitrogen = 
$$\frac{(1.4007 \times \text{TF} \times [\text{HCl}] \times (\text{Sample}(\text{mL}) - \text{Blank}(\text{mL})))}{W_1}$$

%Protein = (%Nitrogen × CF)

where:

- *W*<sub>1</sub> is the sample weight (g).
- TF is the Titration Factor = 1.
- [HCl] is the HCl concentration = 0.1 N
- Sample (mL) is the HCl consumption in the sample.



Fig. 5 Schematic and simplified representation of the steps of the ash determination

3.5 Ash

Determination

- Blank (mL) is the HCl consumption in the blank.
- CF is the conversion factor.

Ash determination is carried out by weight loss by calcination of organic matter in a muffle furnace at 550 °C, according to ISO 936 (1998) [13]. Figure 5 shows in a simplified and schematic way the steps of the ash determination.

- 1. Firstly, porcelain crucibles must be dried for at least 1 h in the oven (set at 105  $^\circ\mathrm{C}\pm2$   $^\circ\mathrm{C}).$
- 2. Then, cool the porcelain crucibles in a desiccator at room temperature and weigh  $(W_0)$  on the analytical balance, with an accuracy of 0.0001 g.
- 3. Weigh 5 g of minced and homogenized sample into the porcelain crucibles  $(W_1)$ .
- 4. Place the porcelain crucibles with its contents in the cold muffle furnace and gradually increase the temperature of the muffle furnace over 8 h to  $550 \pm 25$  °C. Continue the incineration at  $560 \pm 25$  °C until the ash has a gray-white appearance (*see* Note 18).
- 5. When the ash has a gray-white appearance, remove the porcelain crucibles with their contents from the muffle furnace and allow to cool in the desiccator to room temperature (1 h).
- 6. Finally, weigh the porcelain crucibles and their content (ashes)  $(W_2)$  on the analytical balance, with an accuracy of 0.0001 g.
- 7. Expression of results.
  - (a) The results for the samples are expressed as a percentage of ash following the formula:

$$\% \text{Ash} = \left[\frac{W_2 - W_0}{W_1 - W_0}\right] \times 100$$

where:

- $W_0$  is the porcelain crucible weigh (g).
- *W*<sub>1</sub> is the sample and crucible weigh before calcination (g).
- $W_2$  is the sample and crucible weigh after calcination (g).

#### 4 Notes

- 1. All laboratory waste must be eliminated according to its nature following current regulations, differentiating at least acids, bases, organic solvents, halogenated solvents, and non-halogenated solvents.
- 2. Due to the toxic and/or corrosive nature of the solvents and reagents used in all described determinations, it is necessary for the operator to take the necessary protective measures (gloves, glasses, etc.), as well as to carry out all the operations in laboratory fume hoods.
- 3. The concentration of this reagent is approximate, since it is only used to neutralize gases in the digestor.
- 4. Temperatures of over 500 °C are reached in the muffle furnace, so it is necessary to handle the crucibles with gloves and special forceps to avoid burns.
- 5. In order to prepare the stainless steel capsules, you need to add an amount of sand (dry sea sand) equal to three to four times the mass of the sample portion that you will add to the stainless steel capsules, and place the glass rod inside the capsule and on the sea sand.
- 6. If you find it difficult to mix the sample with sea sand, you can use as much ethanol as you need (approximately 5 mL). The ethanol will evaporate during drying of the sample in the oven.
- 7. In samples where the fat is not freely accessible to the solvent such as fats in starch-lipid complexes or to lipoproteins or phospholipids, as in mixed products such as baked goods, hydrolysis must be performed.
- 8. For proper sealing, the heat sealer must be kept at a power of 6 and keep the filter in it for approximately 5 s after the red light has turned off.
- 9. At the same time as samples, we will insert an empty filter that will serve as a blank and another one with a reference material in the same way as the rest of the samples.
- 10. We must be sure that the water is open for cooling the equipment before starting it.
- 11. The result is expressed as a percentage, grams of fat per 100 g of product.



Fig. 6 Digestion programs

- 12. In order to place the sample inside the celite, a hole must be made with a glass rod.
- 13. At the same time as samples, we will insert a filter with 0.5 g of celite that will serve as a blank, and another one with a reference material in the same way as the rest of the samples.
- 14. In each batch of samples, a blank, ammonium sulfate, and a control sample (reference material) will be analyzed. With ammonium sulfate we can check the correct functioning of the digestion, and with the reference material we can check both the digestion and the distillation.
- 15. You must select a digestion program depending on the type of sample:
  - (a) Program 1 (Fig. 6): Fresh samples.
    - *Step 1*: 15 min—400 °C. During 15 min there is a constant rise in temperature until reaching 400 °C.
    - *Step 2*: 1 h 30 min—400 °C. Once 400 °C has been reached, it is kept at this temperature for 1 h and a half.
  - (b) Program 2 (Fig. 6): Dehydrated samples.
    - Step 1: 15 min—200 °C. During 15 min there is a constant rise in temperature until reaching 200 °C.
    - *Step 2*: 20 min—200 °C. Once 200 °C is reached, it is kept at this temperature for 20 min.
    - *Step 3*: 15 min—400 °C. After 20 min of the previous step, there is another constant rise in temperature until reaching 400 °C.
    - Step 4: 2 h—400 °C. Once 400 °C has been reached, it is kept at this temperature for 2 h.
- 16. The equipment must be turned on 20 min before starting the analysis. At the same time, we must turn on the "Vapodest Manager" software.
- 17. The sample distillate is collected in a beaker with boric acid (2%) and titrated with hydrochloric acid (0.1 N). The

equipment records the initial pH of the boric acid and ends the analysis when the final pH is equal to the initial one.

 The muffle furnace is programmed to maintain incineration for 8 h. This time should be sufficient for most samples.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568). Noemí Echegaray acknowledges to Consellería de Cultura, Educación e Ordenación Universitaria (Xunta de Galicia) for granting with a predoctoral scholarship (Grant number IN606A-2018/002).

#### References

- 1. Pérez Montes A, Rangel-Vargas E, Lorenzo JM et al (2021) Edible mushrooms as a novel trend in the development of healthier meat products. Curr Opin Food Sci 37:118–124
- 2. López-Pedrouso M, Lorenzo JM, Gullón B et al (2021) Novel strategy for developing healthy meat products replacing saturated fat with oleogels. Curr Opin Food Sci 40:40–45
- 3. Domínguez R, Munekata PE, Pateiro M et al (2021) Immobilization of oils using hydrogels as strategy to replace animal fats and improve the healthiness of meat products. Curr Opin Food Sci 37:135–144
- European Commission (2011) Regulation of the European Parliament and of the Council of 25 October 2011 on the provision of food information to consumers, amending Regulations (EC) No 1924/2006 and (EC) No 1925/2006 of the European Parliament and of the Council, and repealing Commission Directive 87/250/EEC, Council Directive 90/496/EEC, Commission Directive 1999/ 10/EC, Directive 2000/13/EC of the European Parliament and of the Council, Commission Directives 2002/67/EC and 2008/5/EC and Commission Regulation (EC) No 608/2004, 1169/2011/EU. In: Official Journal, L 304/18, 22/11/2011
- FDA (2016) Food labeling: revision of the nutrition and supplement facts labels; final rule. 21 CFR 101. Federal Register 2016. Food Drug Administration 103:33742–33999
- 6. Wijaya CH, Wijaya W, Mehta BM (2015) General properties of major food components. In:

Cheung PCK, Mehta BM (eds) Handbook of food chemistry. Springer-Verlag, Berlin, pp 15–54

- 7. Otles S, Ozyurt VH (2015) Classical wet chemistry methods. In: Cheung PCK, Mehta BM (eds) Handbook of food chemistry. Springer-Verlag, Berlin, pp 133–149
- Presidencia del Gobierno (1979) Métodos oficiales de análisis de aceites y grasas, productos cárnicos, cereales y derivados, fertilizantes, productos fitosanitarios, productos lácteos, piensos, aguas y productos derivados de la uva. B.O.E. núm. 207 de 29 de Agosto de 1979
- 9. ISO 1442 (1997) Meat and meat products. Determination of moisture content (reference method), 2nd edn. International Organization for Standarization, Geneva
- AOCS Oficial Procedure Am5-04 (2005) Rapid determination of oil/fat utilizing high temperature solvent extraction. American Oil Chemists Society, Urbana, IL
- 11. ISO 937 (1978) Meat and meat products. Determination of nitrogen content (reference method), 1st edn. International Organization for Standarization, Geneva
- MAFF (1975) Energy allowances and feeding systems for ruminants. UK Ministry of Agriculture, Fisheries and Food. In: Technical bulletin, vol 33. HMSO, London, pp 1–79
- ISO 936 (1998) Meat and meat products. Determination of total ash, 2nd edn. International Organization for Standarization, Geneva



## **Chapter 2**

#### **pH and Color**

#### Alfredo Teixeira, Rubén Domínguez, Javier F. Rey, Gonzalo Aleu, Mirian Pateiro, and José Manuel Lorenzo

#### Abstract

pH measurement and color evaluation are two important parameters for determining meat quality. There are the characteristics that after slaughter until the installation of rigor mortis should be assessed to define the quality of a meat and its ability to be processed into quality products. The aim of this chapter is to give a clear and concise view of the techniques used to assess pH and color both physical (CIELAB) and chemical (heme pigment content) in meat and meat products.

Key words Meat, Carcass, pH, Color, CIELAB, Heme pigments, Spectrometer, Spectrophotometer

#### 1 Introduction

The pH and color are probably the most important physicalchemical characteristics of meat quality evaluation. Both are important in the detection of meat quality deviations once they are related with the post-mortem chemical and biochemical reaction occurring to conversion of muscle to meat [1] and particularly the color is the only characteristic that consumers could assess at the time of purchase and decide this option to buy or not since consumers relate the color of meat with its sensory properties. On the other hand, the discoloration is one of the first indicator of meat spoilage. The pH value of meat affects the color and its quality in terms of waterholding capacity and the sensory quality namely its tenderness and flavor [2]. At slaughter the meat pH is neutral (around 7.0) and after slaughter the muscle under anaerobic metabolism conditions the pH decreases as result the conversion of glycogen to lactate [1] and the pH value at rigor mortis is named the ultimate pH and depending on animal species and muscle type the ultimate pH of 5.3-5.8 is reached at different times [2]. As a result of the effect of the pH in the physicochemical characteristics of the meat, a wrong

kinetics of pH during the installation of rigor mortis can affect the suitability of meat to technological processing.

Therefore, pH measurement gives important information on the quality of meat and particularly in the pork the pH value 24 h after slaughter is a fundamental data to detect PSE (pale, soft, exudative) and DFD (dark, firm, dry) meat. Also, the pH measurement is essential to detect DFD beef and sheep meats. So, to monitor the ultimate pH kinetics the method and process of measurement should be defined.

In addition, color deviations in the abnormal course of rigor mortis and meat aging conditions could occur [3]. Color deviations assume a color perceived as normal and in meat quality a normal color is not universally consensual. Anyway, globally is expected that beef and mutton are red, pork, veal, and lamb pink and poultry white meats. Besides white, or red, terms such as shine, pale, or dark are also usually used in meat color characterization. Thus, depending on what is intended to characterize or classify it is important to set the color evaluation procedure method. Color measurement in association with pH is useful for detection of quality deviations such as PSE and DFD in pork or DFD in beef and sheep. Beyond the DFD and PSE meats the National Pork Producers Council (NPPC) [4], based in the pH and color assessment, defines three more categories: RFN (red, firm, non-exudative), RSE (red, soft exudative), and PFN (pale, firm, non-exudative). Color is also used by NPPC to analyze and define the marbling scores corresponding to intramuscular lipid content (Fig. 1). Normal color deviations can result in abnormal kinetics of muscle talk in meat, compromising their quality and adequacy for industry processing or influencing consumer purchase decision. Thus, the color measurement during slaughter and rigor process as well as the values of pH measured at 45-60 min after slaughter and the ultimate pH are essential to assess and define the quality of meat.

Until the middle of twentieth century, pH measurement was made using as pH indicators a wide diversity of color substances in solution or paper strips or subsequently using glass electrodes that incorporate reference electrodes. Nowadays pH was measured using pH-meter, with large portability and easy handling, with different devices and producers. In the recent years, non-invasive hyperspectral methodologies [5] for rapid pH monitoring for meat quality assessment have been developed. The Raman spectroscopy, a portable device shows considerable potential for some applications in the meat sector, including color measurements and pH [6]. In any case, in the slaughterhouse in the system most currently used to measure the pH is the pH-meter directly on carcass or meat using a probe or an electrode provided with a penetration blade.

The color of the meat can be analyzed using visual or instrumental methods. The visual evaluation of the color consists basically in the use of color pattern cards or photographic scales,



Fig. 1 Pork Quality Standards defined as Color—Texture—Exudation, Color Standards, and Marbling standards as described [4]

resulting in a subjective evaluation regarding the conditions of assessment, lighting, description of the different color tones, training of assessors, and the difficulty to find corrected matching between the patterns and the samples analyzed. Instrumental evaluation using the CIELAB system [7] specifies the color in a space of three coordinates defined as L\*, a\*, b\*. L\* is the lightness and varies from 0 (black) to 1 (white). Each series of L \*, a\*, and b\* values correspond a point defined as an exact color in the three-dimensional color sphere (Fig. 2). The a\* and b\* are chromatic coordinates and measure de red-greenness and yellow-blueness, respectively and are used to calculate saturation or chroma (C\*) and hue (h°) [1]. Color coordinates are measured using portable color analyzers. These devices also automatically compute the C\* and h°.

Measuring the concentration of heme pigments is another way to analyze the color of meat. Color is determined by the chemical form and the concentration of the heme pigment myoglobin and hemoglobin: oxymyoglobin (MbO<sub>2</sub>), deoxymyoglobin (Mb), and metmyoglobin (MMb). The MbO<sub>2</sub> is responsible for the bright red color of fresh meat and the deoxymyoglobin or reduced myoglobin





(Mb) shows a purple-red meat color while the MMb exhibits a brown color as a result of oxidation of myoglobin pigment [1]. The quantification of total three states of pigment in meat samples could be assessed by reflectance spectrophotometry [9].

Therefore, the present chapter of the book aims to describe the procedures to assess pH and color characteristics as important factors to evaluate meat and meat products quality.

2	Ма	iter	ials
---	----	------	------

	In this section, we will describe the procedures for the determina-
	tion of pH, color by the CIELAB system, and color measurement of total heme pigment concentration.
2.1 pH Determination	The pH determination can be carried out on the carcass or on meat samples or meat products, depending on what is to be analyzed or evaluated ( <i>see</i> <b>Note 1</b> ). Currently, the most common method uses the following material:
	1. A portable pH-meter provided with a penetration electrode ( <i>see</i> <b>Note 2</b> ).
	2. Buffer solutions with pH of 7.0 and 4.0.
2.2 Physical Color (Cielab System)	The model and make of the device used can be varied. The most usual are the chromameters Konica Minolta from the series CR200, CR300, CR400, CR410 ( <i>see</i> <b>Note 2</b> ).
2.3 Chemical Color (Heme Pigments)	The chemical color determination needs a meat sample preparation that requires the use of some solutions.
	1. Distilled water.
	2. Acetone.
	3. HCl 35%.
	4. Filter paper Ø 150 mm and a maximum of 31 $\mu$ m of pore.
	5. A spectrophotometer (see Note 3).
	For the measurement of pigments deoxymyoglobin (Mb), oxymyoglobin (MbO <sub>2</sub> ), and metmyoglobin (MMb):
	1. PVC film with very low oxygen permeability.
	2. 10% of sodium dithionite solution $(Na_2S_2O_4)$
	3. 1% ferricyanide and potassium cyanide solution [K <sub>4</sub> Fe(CN) <sub>6</sub> ].

#### 3 Methods

3.1	pH Measurement	Before taking a measurement the pH-meter should be calibrated
		using buffer solutions with a known pH of 4.0 and 7.0. It is very
		important to stabilize the temperature of the electrode to the
		temperature of the muscle (see Note 4). There are metal and glass
		electrodes, equipped with a piercing tip that allows the measure-
		ment of pH directly in the carcass muscles or in pieces of meat or
		processed meat products (Fig. 3) (see Note 5). Measurements made
		directly on the carcass are usually taken at m. Longissimus thoracis et
		lumborum (also known as longissimus dorsi) and in



Fig. 3 pH measurement in a carcass, a piece of meat, and a processed meat product

*m. Semimembranosus (see* **Note 6**). The equipment is also able to measure in homogenized meat or in a meat sample homogenized in water.

The instruments take measurements at 400-700 nm and give readings at intervals of 10 nm, allowing for thorough color data analysis and come with software to automatically compute the h\* and C\* (Fig. 4). Some practical considerations in relation to its use must be considered:

- 1. A record of the specifications regarding the animal (species, breed, food, age, sex) conditions of transport and slaughter especially the conditions until reaching the rigor with record of pH (*see* Note 7).
- 2. Time of sampling and location of measurement should be previously defined (*see* **Note 8**).
- 3. If the samples were stored, the conditions of storage (temperature, moisture, light, and overwrap) should be specified.
- 4. Fresh muscle samples need at least 15 min of blooming time to the pigments at surface to oxygenate.
- 5. Three repetitions of each measurement are recommended (*see* **Note 9**).
- 6. A calibration based on the black standard as L \* = 0 and white as L \* = 100 provided with each chromometer must be performed.

3.2 Color Measurement (CIELAB System)



Fig. 4 Color coordinates L<sup>\*</sup>, a<sup>\*</sup>, and b<sup>\*</sup> assessment by a Minolta chromameter

Finally, whenever color measurements are made on samples of meat or fat, all information on the conditions under which they were carried out, as described here, from 1 to 6 must be provided, including the make and model of the equipment used.

3.3 Color Measurement (Heme Pigments Content) The chemical analysis consists of measuring the amount of pigments present in the muscle. The Hornsey method [10] is a simple and rapid indirect method that allows the determination of the myoglobin hierarchy and hemoglobin. As the residual hemoglobin content is very low, the estimation of the myoglobin obtained in this way is very close to the real value. The procedure consists in (Fig. 5):

- 1. Take 5 g of meat sample of *m. Longissimus thoracis et lumborum* (*longissimus dorsi*) (see Note 10).
- 2. Mince the sample in meat grinder type Moulinex.
- 3. A duplicate of each sample will be performed with an extraction volume of 25 mL.
- 4. In a pyrex tube with screw cap, place 1 mL of distilled water and 5 g of ground beef, 20 mL of acetone. Shake with a glass rod until the meat is a uniform whitish color. Then add 0.5 mL of HCl (35%) (*see* **Note 11**).
- 5. The solution should stand for 24 h in a refrigerator in the dark.
- 6. The solution is filtered with double filter paper ( $\emptyset$  150 mm and a maximum of 31  $\mu$ m of pore).
- 7. After pigment extraction, measure optical density (OD) with a spectrophotometer at 512 or 640 nm (*see* Note 12).
- 8. The results are expressed in percent (see Note 13):
  - At 640 nm:  $OD \times 17.75 = mg myoglobin/g fresh muscle$ .
  - At 512 nm:  $OD \times 8.82 = mg myoglobin/g fresh muscle$ .

The pigment percentages of the meat, deoxymyoglobin (Mb), oxymyoglobin (MbO $_2$ ), and metmyoglobin (MMb) could be



Fig. 5 Determination of meat pigment content [10]

obtained by means of the reflectance on the surface of the meat by the method developed by Hunt et al. [11] (*see* Note 14).

- 1. The reflectance spectrum obtained for a sample of meat  $(3 \times 3 \times 2 \text{ cm})$  without visible fat or connective tissue after cutting is 100% of Mb.
- 2. After the time of blooming, the meat sample should remain in contact with the air in a tray cover with a film permeable to the oxygen. The reflectance spectrum is 100% MbO<sub>2</sub>.
- 3. Finally, the meat piece becomes 100% (MMb) when the meat is introduced in a 0.5% ferricyanide and potassium cyanide solution.
- 4. To quantify the amounts of myoglobin redox forms we must have the reflectance values of each one of the pigments (Mb, MMb, MbO<sub>2</sub>). The process involves the use of surface reflectance to calculate the ratio between K (absorbance coefficient) and S (scattering coefficient) at isobestic wavelengths of each myoglobin redox form, 474, 525, 572, and 610 nm.
- 5. Once the myoglobin is converted to 100%, the procedure is as follows:
- Deoxymyoglobin: Place the samples in a 10% sodium dithionite (Na<sub>2</sub>S2O<sub>4</sub>) for 1 min. Remove and clean with absorbent paper. Pack the samples in a vacuum and leave them for 2 h at room temperature. At the end, take the measurements immediately after opening the package.
- Metmyoglobin: Place the samples in a 1% ferricyanide and potassium cyanide solution  $[K_4Fe(CN)_6]$  for 1 min. Remove and clean with absorbent paper. Pack the samples in a vacuum and leave them for 12 h at 2 °C refrigeration. At the end, take the measurements immediately after opening the package.
- Oxymyoglobin: Place the samples in an atmosphere with a high proportion of oxygen with a temperature between 0 and 2 °C. Oxygenation can be carried out by means of a flow of 100% oxygen for 10 min. After performing the measurements, place the samples in an.

Having obtained the measurements of the K/S ratios, calculate the percentages of the pigments using the formulas [11] (*see* Note 15):

$$\% \text{Deoxymyoglobin} = \frac{\frac{\text{K/S}474}{\text{K/S}525} \text{ for } 100\% \text{ of } \text{MMb} - \frac{\text{K/S}474}{\text{K/S}525} \text{ for sample}}{\frac{\frac{\text{K/S}474}{\text{K/S}525} \text{ for } 100\% \text{ of } \text{MMb} - \frac{\text{K/S}474}{\text{K/S}525} \text{ for sample}}{\times 100}$$

$$\% Metmyoglobin = \frac{\frac{K/S}{K/S} \frac{572}{525} \text{ for } 100\% \text{ of } MMb - \frac{K/S}{K/S} \frac{572}{525} \text{ for sample}}{\frac{K/S}{K/S} \frac{572}{525} \text{ for } 100\% \text{ of } MMb - \frac{K/S}{K/S} \frac{572}{525} \text{ for sample}}{\times 100}$$

$$\% Oxymyoglobin = \frac{\frac{K/S\,610}{K/S\,525} \text{ for } 100\% \text{ of } \text{MMb} - \frac{K/S\,610}{K/S\,525} \text{ for sample}}{\frac{K/S\,610}{K/S\,525} \text{ for } 100\% \text{ of } \text{MMb} - \frac{K/S\,610}{K/S\,525} \text{ for } 100\% \text{ of } \text{MMb}}{\times 100}$$

#### 4 Notes

1. Although the procedure is the same, the determination of the pH in the carcasses fundamentally depends on the animal species, the time that elapses after slaughter until installation of rigor mortis, the anatomical point, and muscle of the carcass in which we intend to measure.

- 2. There are different devices of different equipment producers. Before the pH measurement, the pH-meter should be calibrated using buffer solutions.
- 3. There are several devices from different brands. In meat science, the spectrophotometer commonly used are the Hunter-Lab series Miniscan and Ultra scan, some of them are portable; Konica Minolta series CM-3600A, CM-3700A, CM-36dg.
- 4. The pH-meter is provided by an automatic system to adjust the temperature to the temperature of the carcass or muscle to the temperature of the carcass and muscle.
- 5. A scalpel blade or a knife can be used to make a little incision into the muscle to enable the contact of the electrode with the muscle tissue. After each measurement, the electrode must be washed with distilled water. Over time, they may suffer deterioration, delay in response time or with instability in reading. For better conservation, the electrodes should be stored in a 3 M KCl solution.
- 6. As there are differences in the pH measured in different muscles, the definition of the anatomical location is of great importance, to provide a series of works with great repeatability and that provide comparisons between different studies and laboratories.
- 7. Due to the close relationship between color and pH, it is recommendable in carcasses to record the pH at the beginning of chilling and the ultimate pH.
- 8. At least 24 h post-mortem, the measurement must be made. The anatomical point on the carcass must be defined. Usually, color is assessed in a cross section of *m. Longissimus thoracis et lumborum* (Longissimus dorsi) and a minimum of 1–3 cm of thickness is recommended to prevent light passing though it.
- 9. Different colors in the same muscle can occur and muscle infiltrations of marbling and connective tissue can generate some color variability.
- 10. The sample must be homogeneous, without fascia and fat, blood vessels, nerves, and tendons. If the meat sample is frozen, the liquid exudated during defrosting should be recovered.
- 11. A meatless solution (white reading) must be prepared, which will be used for the calibration of the spectrophotometer between sample readings.
- The points where the absorbances or reflectances are equal for the three forms of myoglobin pigment is an isobestic points (512 or 640 nm).

13. The factors 8.82 and 17.75 correspond to the calculation of the final concentration of pigments calculated by the formula [10]:

Factor

× OD  

$$= \frac{\text{volume of extract} \times 652 \times 1.000 \text{g/kg}}{\varepsilon_{\lambda}^{\text{mM}} \times 10^{3} \text{ g of sample}} \text{ and } \varepsilon_{\lambda}^{\text{mM}}$$

$$= 4.8 \text{ at } 640 \text{ nm or} = 9.52 \text{ at } 512 \text{ nm}.$$

- 14. The K/S coefficients used are:
  - (a) K/S474/K/S525 to estimation of Mb.
  - (b) K/S572/K/S525 to estimation of MMb.
  - (c) K/S610/K/S525 to estimation of MbO<sub>2</sub>.
- 15. Once the myoglobin is converted to 100% and the calculations of % Metmyoglobin and % Deoxymyoglobin were processed, the % Oxymyoglobin could easily be calculated as:

%Oxymyoglobin = 100

- (%Deoxymyoglobin + %Metmyoglobin).

#### Acknowledgments

The authors are grateful to the Foundation for Science and Technology (FCT, Portugal) for financial support by national funds FCT/MCTES to CIMO (UIDB/00690/2020) and to Laboratory of Carcass and Meat Quality of Agriculture School of Polytechnic Institute of Bragança "Cantinho do Alfredo." The authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568).

#### References

- 1. Warriss PD (2010) Measuring the composition and physical characteristics of meat. In: Meat science: an introductory text, 2nd edn. CBI Publishing, Wallingford, pp 229–251
- 2. Honikel KO (2004) pH measurement. In: Encyclopedia of meat sciences, 1st edn. Elsevier Academic Press, London, pp 238–242
- 3. Monin G (2004) Colour and texture deviations. In: Encyclopedia of meat sciences, 1st edn. Elsevier Academic Press, London, pp 323–330
- National Pork Producers Council (2000) Pork composition and quality assessment procedures. NPPC, Des Moines, IA

- 5. Yao X, Cai F, Zhu P et al (2019) Non-invasive and rapid pH monitoring for meat quality assessment using alow-cost portable hyperspectral scanner. Meat Sci 152:73–80
- 6. Beganović A, Hawthorme LM et al (2019) Critical review on the utilization of handheld and portable Raman spectrometry in meat Science. Foods 8:49
- CIE (1976) Recommendations on uniform color spaces – color difference equations. Psychometric color terms. Supplement No. 2 to CIE Publication No. 15 (E-1.3.1) 19078, 1971/(TC-1-3). Commission International de l'Eclairage, Paris, France

- 8. Konica Minolta (2021) Konica Minolta colorimetric technology. Giving shape to ideas. https://www5.konicaminolta.eu/en/measur ing-instruments/media-centre/poster-down load.html
- Franke WC, Solberg M (1971) Quantitative determination of metmyoglobin and total pigment in an intact meat sample using reflectance spectrophotometry. J Food Sci 36:515–519
- Hornsey KO (1956) The colour of cooked curded pork estimation of the nitric-oxid heam pigments. J Sci Food Agric 7:534–540
- Hunt MC, King DA et al (2012) Meat color measurement guidelines. Champaign, IL, American Meat Science Association, p 124



# **Chapter 3**

# **Texture Analysis**

### Noemí Echegaray, Marcelo Rosmini, Mirian Pateiro, Rubén Domínguez, Paulo E. S. Munekata, José Manuel Lorenzo, Eva María Santos, and Roberto Bermúdez

#### Abstract

One of the most important quality parameters of the meat and meat products are their texture, which may be influenced by breed, sex, production system, *post-mortem* factors and/or production process. Several years ago, meat industries tested their food products with a panel of expert tasters (sensory analysis). This method alone can have problems because evaluations of the tasters could vary depending on their mood, the ambient temperature, their traditions and culture, among others. In short, this method is good but very subjective. Currently, companies are using both expert tasters and texture analyzers because they are complementary, and instrumental measurement provides more objective values. In this way, it is possible to know if a product satisfies (to a greater or lesser degree) the preferences of most consumers.

This chapter provides a comprehensive guide to instrumental texture analysis in meat and meat products. Warner-Bratzler test (WB) and Texture Profile Analysis (TPA) methods will be detailed step by step carefully.

Key words Warner-Bratzler, TPA, Shear Force, Hardness, Adhesiveness, Springiness, Cohesiveness, Gumminess, Chewiness

#### 1 Introduction

Texture is a very important characteristic of meat and meat products since they give a lot of information about the product that consumers are going to eat or that they are eating. This fact is important, because the texture perception begins in our mind when we see the food. First of all, texture is defined as "all the mechanical, geometrical and surface attributes of a product perceptible by means of mechanical, tactile and, where appropriate, visual and auditory receptors" [1]. Texture greatly influences the perception that the consumer has and feels when eating a product. Consumers can identify several information through it, such as the type of product, its freshness, whether the product is raw or cooked,

José Manuel Lorenzo et al. (eds.), *Methods to Assess the Quality of Meat Products*, Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_3,

© The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

overcooked or undercooked, and also, it influences taste perception. Therefore, it is a key parameter that should take into account when formulating or improving a meat product [2].

There are a lot of factors (*ante*- and *post-mortem*) that can affect the texture of meat and meat products, such as animal handling [3], the type of muscle, the type of product (fresh and aged meat [4, 5], dry-cured meat products [6] and meat paste as pâté [7]), and how people usually eat it (according to their culture and tradition). In addition, many strategies to make meat products healthier could affect the texture of the products, since reducing the amount of salt, fat, or sugar, it will not only affect the taste but also the texture that consumer expects to find in this product. Therefore, it is important to note that we must analyze the sample in the same way that this product should be consumed, so that the results are a reflection of what the consumer will experience.

Among the textural characteristics, the most commonly used are hardness, cohesiveness, and juiciness. Three types of methods are applied to evaluate texture: sensory, instrumental (known as objective, physical, or mechanical), and indirect methods (collagen content, dry matter, among others) [8]. Instrumental methods are generally based on mechanical tests, which evaluate the resistance of the meat product to forces greater than gravity acting on it. Warner-Bratzler test (WB) and texture profile analysis (TPA) are the most common ways to evaluate meat tenderness. These methods simulate the conditions that the meat or meat product is exposed to in the mouth [8]. Texture analyzers are used for this purpose, usually equipped with interchangeable load cells, WB blade, and cylindrical probes. WB test consists of cutting the sample with WB blade to simulate chewing with the incisor teeth, and maximum shear force, shear firmness, and total necessary work are obtained after selecting parameters such as test speed, probe height, or contact force. The first parameter shown by the peak higher of the force-time curve, represents the maximum resistance of the sample to the cut. Shear firmness is represented by the slope from the beginning of the cut up to the highest point of the forcetime curve and total work by the area under the curve [9]. TPA consists of two-cycle compression and simulates chewing with the molars. Hardness, cohesiveness, springiness, gumminess, and chewiness were obtained. Hardness represents the maximal force of the first compression of the product. Cohesiveness is represented by the ratio of work done between the second and the first deformation, whereas springiness is measured at the down stroke of the second compression. Finally, gumminess and chewiness are calculated as Hardness  $\times$  Cohesiveness and Gumminess  $\times$  Springiness, respectively [10].

The aim of this chapter is to be a guide for instrumental texture analysis of meat and meat products, from giving the parameters to set up the texture analyzer to the methodology to perform both test, WB and TPA. So, this chapter will help to standardize the methodologies used, and in this way, it will facilitate the comparison of results.

### 2 Materials

	All tests will be performed with a TA-XT Plus (Stable Micro Systems, Godalming, UK) texture analyzer ( <i>see</i> <b>Note 1</b> ) and its corresponding accessories ( <i>see</i> <b>Note 2</b> ). In addition, texture analyzer is connected to a computer equipped with a texture analysis software package "Exponent" ( <i>see</i> <b>Note 3</b> ).
2.1 Sample	1. Vernier caliper (0–150 mm).
Preparation for	2. Cutting board.
Analysis	3. Knife.
2.2 Warner-Bratzler	The Warner-Bratzler set (Fig. 1) consists of:
Test	1. Warner-Bratzler Blade Set with "V" slot blade for USDA Stan- dard (HDP/WBV).
	2. Slotted blade insert.
	3. Blade holder.
2.3 Texture Profile	The texture profile analysis set (Fig. 2) consists of:
Analysis (Compression	1. Cylinder probe of 50 mm of diameter $(P/50)$ .
Testy	2. Flat platform.
	3. Probe insert.
2.4 Texture Profile	The texture profile analysis set (Fig. 3) consists of:
Analysis (Penetration	1. Cylinder probe of 6 mm of diameter $(P/6)$ .
resij	2. Flat platform.
	3. Probe insert.

### 3 Methods

	All procedures will be carried out at room temperature unless otherwise specified.
3.1 Sample Preparation for Analysis	First of all, we need to prepare the sample. This step is very impor- tant because the results depend on it. How to prepare samples for one test or another is different and depends on the type of sample. As a general rule, the sample is analyzed in the same way as it is



#### Fig. 1 Warner-Bratzler set



#### Fig. 2 TPA set (compression test)



Fig. 3 TPA set (penetration test)

usually eaten, so raw meat should be cooked before. Therefore, in some samples the water holding capacity is also determined at the same time.

3.1.1 Water Holding Capacity (WHC) The water holding capacity (WHC) of meat is determined by cooking losses according to the method proposed by Honikel (1997) [11].

- 1. Use steaks approximately 2.5 cm thick.
- 2. Once weighed, vacuum packed (80%) each fillet in a plastic bag and cooked in a water bath at 80 °C until it reaches an internal temperature of 70 °C (controlled with thermocouples) (*see* Note 4).
- 3. Then, the bags are removed from the bath and placed on a tray until they reach room temperature (approximately 30 min).
- 4. Finally, cooked steaks are removed from the bags and carefully dried to remove any remaining liquid on the surface. Cooking losses are expressed as the percentage of weight lost with respect to the initial one.
- 5. Expression of results

Calculate the cooking loss as a percentage, using the following equation:

%Cooking loss = 
$$\left[\frac{(W_0 - W_1)}{W_0}\right] \times 100$$

where:

- $W_0$  is the steak weigh before cooking (g).
- $W_1$  is the steak weigh after cooking (g).

Once the steak is cooked, we must cut it for carry out the WB test.

- 1. Firstly, remove the edges of the steak (*see* **Note 5**).
- 2. Next, we cut six pieces of the steak (*see* **Note 6**). Using vernier caliper, we must cut pieces with a size of  $1 \text{ cm} \times 1 \text{ cm} \times 2.5 \text{ cm}$  (high  $\times$  width  $\times$  length).
- 3. Finally, take the samples to the texture analyzer.

In this test we must divide the samples in fresh or dry-cured sausages, and dry-cured meat products such as ham (*see* Note 7).

- 1. Fresh and dry-cured sausages.
  - (a) Firstly, fresh sausages must be cooked before in the same way that fresh meat and next, remove the skin from the sausage where it was stuffed.
  - (b) Then, we cut four slices of 2 cm wide using a vernier caliper in both fresh and dry-cured sausages. The skin must be also removed in dry-cured sausages.
  - (c) Finally, take the samples to the texture analyzer.

3.1.2 Sample Preparation for WB Test

3.1.3 Sample Preparation for TPA (Compression Test) 2. Other dry-cured meat products

In the case of dry-cured meat products such as dry-cured ham, we must cut a representative slice from the muscle that interests us (usually in dry-cured ham *semimembranosus* or *biceps femoris* muscle).

- (a) Firstly, remove the edges of the slice.
- (b) Next, we cut six pieces of the steak. Using vernier caliper, we must cut pieces with a size of  $1 \text{ cm} \times 1 \text{ cm} \times 1 \text{ cm}$  (high  $\times$  width  $\times$  length).
- (c) Finally, take the samples to the texture analyzer.

3.1.4 Sample Preparation for TPA (Penetration Test)

3.2 Warner-Bratzler Test Usually, we carry out this analysis in spreadable paste of meat like to pâté. These products are ready for analysis, you just need open the can or glass bottle and do the analysis.

Usually, this test is used in fresh and aged meat. Warner-Bratzler test is carried out according to Honikel (1997) [11] with slight modifications. Figure 4 shows in a simplified and schematic way the steps of the WB test. Once the samples are ready for analysis:



Fig. 4 Schematic and simplified representation of the steps of the WB test

#### Table 1 Analysis parameters to perform WB test

WB parameters	
Load cell (Kg)	30
Calibration weight (kg)	5
Return distance (mm)	50
Return speed (mm/s)	10
Contact force (g)	2
Pretest speed (mm/s)	3.33
Test speed (mm/s)	3.33
Posttest speed (mm/s)	15
Distance (mm)	30



Fig. 5 Typical force deformation curve of the WB test

- 1. We must set up the parameters of analysis in the "Exponent" software (*see* **Note 8**). You can see the WB parameters for fresh and aged meat in Table 1.
- 2. Then, samples which have been prepared for the WB test can be cut with the WB blade. In order to achieve a representative result, six pieces must be tested.
- 3. In Fig. 5, you can see a typical force deformation curve of the WB test.
- 4. We analyze the curve with "Exponent" software for obtaining the following parameters: Firmness (N/s), Work (N\*mm), and Shear Force (N/cm<sup>2</sup>).
- 5. Finally, the result is calculated as the average of the six pieces.

#### 3.3 Texture Profile Analysis (Compression Test)

As we explained above, this test could be carried out on fresh and dry-cured sausages, and other dry-cured meat products. Texture profile analysis is performed according to Honikel (1997) [11] and Bourne et al. (1978) [12] with slight modifications. Figure 6 shows in a simplified and schematic way the steps of the TPA (Compression test). Once the samples are ready for analysis:

- 1. We must set up the parameters of analysis in the "Exponent" software (*see* **Note 8**). You can see the TPA parameters for fresh and dry-cured sausages in Table 2. In Table 3 you can see the parameters for other dry-cured meat products (*see* **Note 9**).
- 2. Then, samples which have been prepared for the TPA can be compressed with the cylinder probe of 50 mm. In order to achieve a representative result, six pieces must be tested.
- 3. In Fig. 7, you can see a typical force deformation curve of the TPA test.

- 4. We analyze the curve with "Exponent" software to obtain the following parameters: Hardness (N), Adhesiveness (N\*s), Springiness (mm), Cohesiveness, Gumminess (N), and Chewiness (N\*mm).
- 5. Finally, the result is calculated as the average of the six pieces.

· Dry-cured sausage example



Fig. 6 Schematic and simplified representation of the steps of the TPA (compression test)

#### Table 2

#### Analysis parameters to perform TPA (Compression test) on fresh and dry-cured sausages

TPA parameters (Compression test)	
Load cell (Kg)	30–50
Calibration weight (kg)	5
Return distance (mm)	30
Return speed (mm/s)	20
Contact force (g)	2
Pretest speed (mm/s)	10
Test speed (mm/s)	1
Posttest speed (mm/s)	10
Deformation (%)	50

Table 3									
Analysis	parameters	to perform	TPA (Con	pression	test)	on other	dry-cured	meat	products

TPA parameters (Compression test)	
Load cell (Kg)	30
Calibration weight (kg)	5
Return distance (mm)	20
Return speed (mm/s)	20
Contact force (g)	2
Pretest speed (mm/s)	3.33
Test speed (mm/s)	3.33
Posttest speed (mm/s)	3.33
Deformation (%)	60



Fig. 7 Typical force deformation curve of the TPA (compression test)

3.4 Texture Profile Analysis (Penetration Test) Usually, we carry out this analysis in products that are made from a meat paste, as in the case of pâtés. Texture profile analysis is performed according to Vargas-Ramella et al. (2020) [13]. We test this meat product directly in the commercial packaging. Figure 8 shows in a simplified and schematic way the steps of the TPA (Penetration test).



Samples such as pâté are ready for analysis Samples are penetrated with the cylinder probe of 6 mm

Fig. 8 Schematic and simplified representation of the steps of the TPA (penetration test)

Table 4						
Analysis	s parameters	to	perform	TPA	(Penetration	test)

TPA parameters (Penetration test)	
Load cell (Kg)	5
Calibration weight (kg)	2
Return distance (mm)	50
Return speed (mm/s)	20
Contact force (g)	2
Pretest speed (mm/s)	3.33
Test speed (mm/s)	0.80
Posttest speed (mm/s)	3.33
Distance (mm)	8

- 1. We must set up the parameters of analysis in the "Exponent" software (*see* **Note 8**). You can see the TPA parameters for fresh and dry-cured sausages in Table 4.
- 2. Then, once the container has been opened, the sample can be penetrated with the cylinder probe of 6 mm. In order to achieve a representative result, the test is performed six times in different areas.
- 3. In Fig. 9, you can see a typical force deformation curve from the TPA penetration test.
- We analyze the curve with "Exponent" software to obtain the following parameters: Hardness (N), Adhesiveness (N\*s), Springiness (mm), Cohesiveness and Gumminess (N).
- 5. Finally, the result is calculated as the average of the six pieces.



Fig. 9 Typical force deformation curve of the TPA (penetration test)

#### 4 Notes

- 1. TA-XT Plus has 500 N of force capacity, 0.1 g of force resolution, a speed range between 0.01 and 40 mm/s, a maximum aperture of 370 mm, a distance resolution of 0.001 mm, and a data acquisition rate of 2000 pps.
- 2. The accessories are:
  - (a) On the one hand, two interchangeable load cells of 30 kg and 50 kg. One or the other will be used depending on the expected hardness of the sample. Usually, a 30 kg load cell is used with fresh meat and a 50 kg load cell with drier samples such as dry-cured meat products.
  - (b) On the other hand, three different probes. A Warner-Bratzler probe and two cylindrical TPA probes of 50 mm and 6 mm of diameter.
- 3. "Exponent" software allows us to take control of the texturometer, testing programming, automated curve analysis, statistical analysis, presentation of results, etc.
- 4. The temperature probe must be placed carefully on the fillet so as not to damage the sample to be used for texture analysis and to correctly control its internal temperature.
- 5. The edges of the steak are usually very irregular, have some fat and are dried. These facts could affect the results.

- 6. The pieces of steak should be cut parallel to the fibers.
- 7. These kinds of sample (dry-cured meat products) should not be cooked because, as a general rule, they are eaten raw. But in the case of fresh sausages, you have to cook them.
- 8. Texture analyzer should be turned on at least 30 min before analysis. In addition, the equipment must be calibrated. The calibration weight should be 5 kg for 30 kg and 50 kg load cells.
- 9. Note that the parameters of sausages are different from other dry-cured meat products.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568). Noemí Echegaray acknowledges to Consellería de Cultura, Educación e Ordenación Universitaria (Xunta de Galicia) for granting with a predoctoral scholarship (Grant number IN606A-2018/002).

#### References

- ISO 11036 (1994) Sensory analysis Methodology – texture profile. International Organization for Standarization, Geneva
- 2. Barros JC, Munekata PES, de Carvalho FAL et al (2021) Healthy beef burgers: effect of animal fat replacement by algal and wheat germ oil emulsions. Meat Sci 173:108396
- 3. Kiran M, Naveena BM, Smrutirekha M et al (2019) Traditional halal slaughter without stunning versus slaughter with electrical stunning of sheep (*Ovis aries*). Meat Sci 148: 127–136
- 4. Lorenzo JM, Pateiro M, Franco D (2013) Influence of muscle type on physicochemical and sensory properties of foal meat. Meat Sci 94:77–83
- 5. Ruiz M, Beriain MJ, Insausti K et al (2018) Efecto de la maduración en la conservación de la carne de potro. ITEA 114:45–60
- 6. Contreras M, Benedito J, Quiles A et al (2020) Assessing the textural defect of pastiness in dry-cured pork ham using chemical, microstructural, textural and ultrasonic analyses. J Food Eng 265:106690

- 7. Martins AJ, Lorenzo JM, Franco D et al (2020) Characterization of enriched meatbased Pâté manufactured with oleogels as fat substitutes. Gels 6:17
- Saláková A (2012) Instrumental measurement of texture and color of meat and meat products, vol 2. Maso International, Brno, pp 107–114
- Pateiro M, Lorenzo JM, Díaz S et al (2013) Meat quality of veal: discriminatory ability of weaning status. Span J Agric Res 11: 1044–1056
- Bourne MC (1978) Texture profile analysis. Food Technol 32:62–66
- 11. Honikel KO (1997) Reference methods supported by OECD and their use in Mediterranean meat products. Food Chem 5:573–582
- Bourne MC, Kenny JF, Barnard J (1978) Computer-assisted readout of data from texture profile analysis curves. J Texture Stud 9: 481–494
- Vargas-Ramella M, Pateiro M, Barba FJ et al (2020) Microencapsulation of healthier oils to enhance the physicochemical and nutritional properties of deer pâté. LWT 125:109223

# **Chapter 4**



# **Fatty Acids**

### Rubén Domínguez, Laura Purriños, Mirian Pateiro, Paulo C. B. Campagnol, Jorge Felipe Reyes, Paulo E. S. Munekata, and José Manuel Lorenzo

#### Abstract

The development of a common technique for the determination of fatty acids is very important due to the role they play in the nutritional, sensorial, and technological quality of meat and meat products. However, and despite being one of the most studied determinations, there are several techniques for fat extraction, for the methylation, and for identification/quantification of fatty acids. With this in mind, it should be mentioned that each of the procedures has certain limitations.

Therefore, and taking into account the above, this chapter gives a clear, comprehensive and complete vision of a procedure for the fatty acid analysis of meat and meat products, including the fat extraction, fatty acids methylation, and subsequent analysis by gas chromatography with FID detection. All stages are described in detail so that the conditions proposed by us can be reproduced by other researchers. Finally, it should be mentioned that the proposed procedure includes a fast methylation phase at room temperature, combining acidic and basic methylation to ensure complete methylation of fatty acids in a simple way and minimizing the formation of artifacts.

Key words Lipid fraction, Omega-3, Omega-6, Saturated fatty acids, Monounsaturated fatty acids, Polyunsaturated fatty acids, Gas chromatography, Basic-acid methylation, Transesterification

#### 1 Introduction

The fat content as well as the fatty acid content of meat and meat products is of vital importance due to its implications on human health and on the sensory quality of meat products [1]. Due to this, during the last decades, multiple strategies have been proposed by both researchers and the meat industry with the aim of producing healthier products [2]. These strategies range from modifying the animals' diet, which affects the content and profile of fatty acids in fresh meat, to the reformulation of multiple meat products, substituting animal fat (with high values of saturated fatty acids) by vegetable or marine origin oils [3]. However, there are numerous technological limitations, because the consistency of animal fat

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_4,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

(saturated) is solid, while oils have a liquid consistency. To overcome this drawback, the immobilization of these oils in gels has been proposed, both using water-based (hydrogels) and lipid-based gels (oleogels) [2]. Furthermore, it must be taken into account that unsaturated fatty acids are also very susceptible to lipid oxidation processes [4], so it is important to control all these aspects when designing strategies for the reformulation of meat and meat products.

With this in mind, it seems clear that a correct determination of the content of fatty acids in meat and meat products is very important. The lipid fraction of fresh meat is mostly made up of neutral lipids or triglycerides (located in adipocytes) and polar lipids or phospholipids (which are constituents of cell membranes). However, in many meat products, mainly those that have a dryingripening process in their manufacture, there is a significant release of fatty acids, so the fraction of free fatty acids in these products can also be important [5]. In these fractions, about 20 fatty acids, with a chain length ranging between C12 and C22, account for more than 85% of fatty acids, but the number of minor fatty acids is much larger [6]. Currently, there is a great diversity of methodologies for the extraction of total lipids in terms of the combination and proportions of solvents. Some of them use a single solvent (such official method proposed by AOAC, which use petroleum ether), but a combination of various solvents ensure adequate polarity to extract both polar and nonpolar lipids [7]. The two most used methods for fat extraction in meat and meat products are Folch et al. [8] and Bligh and Dyer [9]. In both cases, a mixture of methanol and chloroform are used, but differ in the proportion of chloroform:methanol and solvent:sample ratios [7]. Similarly to fat extraction, several procedures were proposed for the fatty acid esterification, a recommended step for gas chromatography analysis, as ester derivatives are more volatile than their corresponding underivatized fatty acids. In this sense, both, acid and base methylation (or transesterification) were used in different meat and meat products. It is well known that the acid methylation is suitable for a complete transesterification of free fatty acids and O-acyl lipids, while base methylation is preferable for the formation of FAME from neutral and polar lipids [6]. Moreover, the use of some acids could form allylic methoxy ether artifacts. Thus, the use of sodium methoxide to preserve the steric configuration of double bounds and to reduce artifact formation was proposed, but it did not methylate free fatty acids. In addition, several methylation procedures use high temperatures, which promote oxidative reactions [4] and could change the fatty acids composition. Therefore, the methodology proposed in the present chapter used combined methylation, acid and base, at room temperature, in order to ensure the complete transesterification of fatty acids from all fractions and overcome the problems that other procedures have. Finally, various

chromatographic techniques, including gas (GC) and liquid chromatography (HPLC) were used in the fatty acids separation and quantification. Among them, GC is the most used by far. In this regard, capillary columns with polar stationary phases had higher resolution capacity than apolar stationary phase columns [6].

Within the several methods used for fatty acids quantification, the most common methods imply three main steps: (a) a solvent extraction, followed by (b) methylation and (c) gas chromatography (GC) identification. With all the aforementioned, the present book chapter aims to fully and comprehensively describe a procedure for determining fatty acids in meat and meat products. In this case, these three stages are specified, being the extraction of the fat with a mixture of solvents, followed by combined methylation (base and acid transesterification) at room temperature and finally the chromatographic analysis by gas chromatography.

#### 2 Materials

	Prepare all solutions using distilled water and/or analytical grade reagents. Use GC-grade hexane. Prepare and store all reagents at room temperature (unless indicated otherwise). Diligently follow all waste disposal regulations when disposing of waste materials ( <i>see</i> <b>Note 1</b> ).
2.1 Fat Extraction Step	1. NaCl (1%): Dissolve 10 g of sodium chloride in 1 L of distilled water.
	<ol> <li>Chloroform with 0.1% BHT: Dissolve 1 g of Butylated hydro- xytoluene (BHT) in 1 L of chloroform.</li> </ol>
2.2 Methylation Step	<ol> <li>Sodium methoxide (0.5 N): Dissolve 11.5 g of extra pure sodium metal (≥99%) in 1 L of methanol (see Note 2).</li> </ol>
	2. Sulfuric acid-methanol solution (10%): Mix 100 mL of sulfuric acid (≥98%) with 900 mL of methanol ( <i>see</i> <b>Note 3</b> ).
	<ol> <li>Sodium bicarbonate (saturated solution): Dissolve 100 g of sodium bicarbonate (85%) in 1 L of distilled water.</li> </ol>
	<ol> <li>Nonadecanoic acid (C19:0; 5 mg/mL) (Internal Standard for sample methylation step): Dissolve 0.25 g of nonadecanoic acid (≥98%) in 50 mL of hexane (see Note 4).</li> </ol>
2.3 Calibration	All fatty acid methyl ester solutions must contain 10 mg/mL. In this case, the authors indicate the references to the specific stan-
Identification	dards used for fatty acid determination; however, others with the same characteristics may be used. All standards should be analytical grade and preferably GC grade analysis. All fatty acid methyl esters standards should be stored at $-20$ °C.

#### Table 1

Volume ( $\mu$ L) of the fatty acid methyl ester standards and hexane for the elaboration of the different points of the calibration curve (final volume 500  $\mu$ L)

Fatty acid standard	STD1	STD2	STD3	STD4	STD5
FAME mix 37 comp. (10 mg/mL)	25	50	100	200	400
11 t-C18:1 (TVA; 10 mg/mL)	1.25	2.5	5	10	20
C18:1n-7 (CVA; 10 mg/mL)	1.25	2.5	5	10	20
C19:0 ME (IS; 10 mg/mL)	15	15	15	15	15
C22:5n-3 (DPA; 10 mg/mL)	1.25	2.5	5	10	20
9c, 11t-C18:2 (CLA; 10 mg/mL)	1.25	2.5	5	10	20
Hexane (GC grade)	455	425	365	245	5

- 1. Fatty acid mixture: 37 component FAME Mix (Supelco; ref.: CRM47885). Ampule of 1 mL with 10 mg/mL of total fatty acid methyl esters (in dichloromethane). The amount of each individual fatty acid ranged between 200 and 600  $\mu$ g/mL.
- 2. Trans-11-vaccenic methyl ester (11t-C18:1; TVA; 10 mg/mL): (Supelco; ref.: CRM46905). Ampule of 1 mL with 10 mg/mL TVA in heptane.
- 3. Cis-11-vaccenic methyl ester (C18:1n-7; CVA; 10 mg/mL): (Supelco; ref.: CRM46904). Ampule of 1 mL with 10 mg/mL CVA in heptane.
- Cis-9, trans-11-octadecadienoic methyl ester (9c,11t-C18:2; CLA; 10 mg/mL): (Matreya LLC; ref.: 1255). Vial with 25 mg CLA (≥98%). In order to obtain a 10 mg/mL solution, dissolve 25 mg in 2.5 mL of hexane.
- Cis-7, 10, 13, 16, 19-docosapentaenoic methyl ester (C22:5n-3; DPA; 10 mg/mL): (Supelco; ref.: 17269). Ampule of 1 mL with 10 mg/mL DPA in heptane.
- 6. Nonadecanoic acid methyl ester (10 mg/mL) (*Internal Standard for calibration curve*): Dissolve 0.5 g of nonadecanoic acid methyl ester (C19:0 ME; ≥98%) in 50 mL of hexane.
- 7. Fatty acid methyl esters standards for calibration curve: In order to obtain a five-point calibration curve, different volumes ( $\mu$ L) of the previously described standards (37 FAME Mix, TVA, CVA, CLA, DPA, and C19:0 ME) were mixed with hexane (Table 1). The final concentrations (ppm) and elution order of each fatty acid methyl ester are those listed in Table 2 (*see* **Note 5**).

Separation and quantification of FAMEs are carried out using a gas chromatograph Agilent mod. 7890B (Agilent Technologies)

2.4 Gas Chromatograph

#### Table 2

Fatty acid methyl esters (in order of elution) and final concentration (ppm) of each point of the calibration curve

N° #	Fatty acid	STD1	STD2	STD3	STD4	STD5
1	C4:0	20	40	80	160	320
2	C6:0	20	40	80	160	320
3	C8:0	20	40	80	160	320
4	C10:0	20	40	80	160	320
5	C11:0	10	20	40	80	160
6	C12:0	20	40	80	160	320
7	C13:0	10	20	40	80	160
8	C14:0	20	40	80	160	320
9	C14:1n-5	10	20	40	80	160
10	C15:0	10	20	40	80	160
11	C15:1n-5	10	20	40	80	160
12	C16:0	30	60	120	240	480
13	C16:1n-7	10	20	40	80	160
14	C17:0	10	20	40	80	160
15	C17:1n-7	10	20	40	80	160
16	C18:0	20	40	80	160	320
17	9t-C18:1	10	20	40	80	160
18	11t-C18:1 (TVA)	25	50	100	200	400
19	C18:1n-9	20	40	80	160	320
20	C18:1n-7 (CVA)	25	50	100	200	400
21	9t,11t-C18:2	10	20	40	80	160
22	C18:2n-6	10	20	40	80	160
23	C19:0 ( <b>IS</b> )	300	300	300	300	300
24	C18:3n-6	10	20	40	80	160
25	C18:3n-3	10	20	40	80	160
26	9c, 11t-C18:2(CLA)	25	50	100	200	400
27	C20:0	20	40	80	160	320
28	C20:1n-9	10	20	40	80	160
29	C20:2n-6	10	20	40	80	160
30	C21:0	10	20	40	80	160
31	C20:3n-6	10	20	40	80	160
32	C20:4n-6	10	20	40	80	160
33	C20:3n-3	10	20	40	80	160

(continued)

Table 2
(continued)

N° #	Fatty acid	STD1	STD2	STD3	STD4	STD5
34	C22:0	20	40	80	160	320
35	C20:5n-3 (EPA)	10	20	40	80	160
36	C22:1n-9	10	20	40	80	160
37	C22:2n-6	10	20	40	80	160
38	C23:0	10	20	40	80	160
39	C24:0	20	40	80	160	320
40	C22:5n-3 (DPA)	25	50	100	200	400
41	C24:1n-9	10	20	40	80	160
42	C22:6n-3 (DHA)	10	20	40	80	160

equipped with a flame ionization detector (FID) and PAL RTC-120 autosampler with liquid injection tool (Pal System). For the separation of FAMEs, a DB-23 fused silica capillary column (60 m, 0.25 mm i.d., 0.25  $\mu$ m film thickness; Agilent Technologies) is used.

#### 3 Methods

Carry out all procedures at room temperature unless otherwise specified.

# **3.1 Fat Extraction** For fatty acid analysis, about 20 mg of fat should be extracted from 10 g of sample following the procedure described by Bligh and Dyer [9], with modifications.

- 1. Weigh 10 g of the sample in a centrifuge tube of 80 mL and add *Vol-1* of NaCl (1%) to achieve moisture of 80% (Table 3) (*see* **Note 6**).
- 2. Add 2 mL of chloroform with 0.1% of BHT, 20 mL of methanol, and 8 mL of chloroform. Homogenize in UltraTurrax during 30 s (at 12,000 rpm). Add 10 mL of chloroform and homogenize (10s). Add 10 mL of NaCl (1%) and homogenize (10s). Centrifuge samples at 3100 × g for 10 min. Remove the upper phase (aqueous phase: water and methanol) using a plastic Pasteur pipette. Remove the "meat phase" using a spatula and/or laboratory tweezers (*see* Note 7).
- 3. Transfer *Vol-2* (Table 3; *see* **Note 6**) of lower phase (organic phase; chloroform and fat) towards a test tube (previously weighted). Evaporate to dryness under  $N_2$  on a TurboVap

#### Table 3

Sample moisture (%)	Vol-1 NaCl (1%) (mL)	Sample fat (%)	Vol-2 chloroform (mL)
$\geq 80$	0.00	$\leq 2$	13.0
70	1.00	5	5.40
60	2.00	10	2.70
50	3.00	15	1.80
40	4.00	20	1.35
30	5.00	30	0.90
20	6.00	40	0.675
10	7.00	50	0.540

Volume of NaCl (1%) that must be added to the sample depending on sample moisture and volume of chloroform that must be taken to evaporate depending on sample fat content

evaporator (*see* **Note 8**). After cooling the tube to room temperature, the test tube is weighed again and the amount of fat is calculated by difference (*see* **Note 9**).

3.2 Fatty Acids Methylation	The fatty acids were transesterified according to the procedure previously described by Barros et al. [10], with some modifications. Figure 1 shows in a simplified and schematic way the steps of the fat extraction and methylation phases.
	1. For the fatty acids transesterification, $20 \pm 5$ mg of extracted fat is mixed with 120 µL of C19:0 (5 mg/mL; nonadecanoic acid; internal standard) and 880 µL of toluene. The mixture must be vortexed to dissolve the fat in the toluene. Then, add 2 mL of a sodium methoxide (0.5 N) solution, vortex during 10 s and allow standing for 15 min at room temperature.
	2. Add 4 mL of a sulfuric acid-methanol solution (10% of $H_2SO_4$ in methanol) and vortex for a few seconds. Add 2 mL of saturated sodium bicarbonate solution and vortex again ( <i>see</i> <b>Note 10</b> ). For the extraction of fatty acid methyl esters, add 1 mL of hexane, vortex for 10 s and allow standing for phase separation. Transfer the organic phase (upper phase) to an appropriate vial.
3.3 Fatty Acid Identification	Separation and quantification of FAMEs are carried out using a gas chromatograph, following the next conditions (Table 4):
and Quantification (GC)	<ol> <li>One microliter of the sample (or the standard) is injected in split mode (50:1) (see Note 11). The injector is maintained at 250 °C and 64.2 mL/min of total flow. A DB-23 fused silica capillary column (60 m, 0.25 mm i.d., 0.25 μm film thickness; Agilent Technologies) is used for FAMEs separation. This</li> </ol>



Fig. 1 Schematic and simplified representation of the steps of the fat extraction and methylation phases

column has excellent resolution and a wide operating temperature range, and is well suited for procedures using cis- and trans- FAMEs isomers.

- 2. The oven conditions are as follows: initial oven temperature of 50 °C (hold for 1 min), first ramp at 25 °C/min to 175 °C, second ramp at 3 °C/min to 230 °C (hold for 3 min), and third ramp at 2 °C/min to a final temperature of 235 °C (hold for 3 min). Helium is used as a carrier gas at a constant flow rate of 1.2 mL/min, with the column head pressure set at 22.9 psi.
- 3. The FID detector is maintained at 280 °C, while the operational flows are set as 40 mL/min of H<sub>2</sub>, 450 mL/min of air, and 30 mL/min of makeup flow (He). The detector signal is recorded at 10 Hz data rate. The total time for chromatographic analysis is 32.83 min.

Inlet	Temperature (° C)	Pressure (psi)	Total flow (mL/min)	Split ratio
Split mode	250	22.916	64.2	50:1
Oven	Rate (° C/min)	Temperature (° C)	Hold time (min)	Total time (min)
Initial	-	50	1	1
Ramp 1	25	175	0	6
Ramp 2	3	230	3	24.33
Ramp 3	2	235	3	32.83
Column	Characteristics	Temperature (° C)	Pressure (psi)	Flow (mL/min)
DB-23	$60 \text{ m} \times 250  \mu\text{m} \times 0.25  \mu\text{m}$	50 (initial)	22.916	1.2
Detector	Temperature (° C)	$H_2$ flow (mL/min)	Air flow (mL/min)	He flow (mL/min)
FID	280	40	450	30

# Table 4 Summary of the chromatographic conditions used for the FAMEs analysis

Data acquisition, equipment control, and data analysis are carried out using the GC software (in our case MassHunter GC/MS; Agilent Technologies). Individual FAMEs are identified by comparing their retention times with those of authenticated standards (37 FAME Mix, TVA, CVA, CLA, DPA, and C19:0 ME) (*see* **Note 12**). Figure 2 shows an example of FAME chromatograms of standard (a) and sample (b). The results could be expressed as mg/100 g of meat, g/100 g of fat, and g/100 g of total fatty acids.

#### 4 Notes

- 1. Due to the toxic and/or corrosive nature of the solvents and reagents used in the extraction and methylation of fatty acids, it is necessary for the operator to take the necessary protective measures (gloves, glasses, etc.) as well as to carry out all the operations in laboratory fume hoods.
- Exothermic reaction. Make cubes of sodium (metal) of approximately 2 cm and add them to 500 mL of methanol in a 1 L bottle (constant stirring). Finally, wait for the mixture to cool down and add the remaining 500 mL.
- 3. Extremely exothermic reaction. This solution must be carried out in an ice bath, with constant stirring (magnetic). In a 2 L bottle, add 900 mL of methanol and slowly add concentrated sulfuric acid, letting it run off the wall of the bottle. If bubbles appear (dangerous point of the reaction), wait a few minutes for the temperature to drop and then continue. In the end, add a spatula tip of methyl red so that the solution has a reddish



Fig. 2 Fatty acid chromatograms of a standard (a) and a fresh meat sample (b)

color, and favor the visualization of the two phases (aqueous = red/organic = colorless) after the methylation process.

- 4. The internal standard (C19:0) should be stored at -20 °C. At this temperature C19:0 precipitates. Therefore, the standard internal needs to be warmed prior to use.
- 5. Due to the extremely small quantities (few  $\mu$ L), the pipetting process must be scrupulously careful. The tip of the micropipettes should be moistened with the solvent (or standard) to ensure that the correct amount of each standard is taken.
- 6. The volume of NaCl (1%) [Vol-1] to be added to the samples, as well as the volume of chloroform (which contains the fat) [Vol-2] that must be evaporated to obtain a final content of approximately 20 mg of fat, depending on the initial moisture and fat content of the samples, respectively. A summary of the

volumes that must be added to samples with different moisture and fat contents is shown in Table 3.

- 7. In order to ensure the complete dehydration of the organic phase, after removing the meat phase, a spatula tip of anhydrous sodium sulfate can be added. After this step, the organic phase is filtered through a filter paper, eliminating the anhydrous sodium sulfate as well as the meat that may remain in the organic phase.
- 8. The evaporator should be turned on a few minutes before evaporating the solvent from the samples to allow the water bath to have the correct temperature. The conditions of Tubo-Vap used for evaporation are: 1.2 bar nitrogen pressure at 50 °C (water bath temperature) during minimum 30 min. The evaporation time can be extended if the solvent is not evaporated.
- 9. An analytical balance with 0.1 mg resolution or higher should be used to weigh the tubes. The amount of fat obtained is calculated by difference (tube with fat after evaporating empty tube). In case the amount of fat obtained is greater than 25 mg, the excess can be removed with a spatula.
- 10. After vortexing the samples when the sulfuric acid-methanol solution was added, the pressure in the tube should be relieved (slowly opening the cap). The bicarbonate must be added slowly as it causes a reaction that can cause the sample and reagents to jump out of the tube. Once again, after vortexing the sample with the bicarbonate, the tube pressure should be relieved by slowly uncapping the tube (release pressure,  $CO_2$  is generated).
- 11. Standards should be injected first, for identification and to create calibration curves for each fatty acid. Identification/ calibration standards must be prepared each time a calibration is performed. During the analysis of the samples, a blank (hexane) should be injected every 15 samples, and a standard (FAME) should be injected in order to verify the retention times every 40 samples. Calibrations should be performed at least once a month, or when any chromatograph maintenance is performed that may affect the resolution of the equipment or the detector signal.
- 12. Standards prepared for calibrations can be stored frozen  $(-20 \ ^{\circ}\text{C})$  and used to check retention times (injection every 40 samples). However, these standards cannot be used for calibration, only to verify retention times. Calibration curve standards must be prepared each time a new calibration is performed.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568).

#### References

- Vargas-Ramella M, Munekata PES, Gagaoua M et al (2020) Inclusion of healthy oils for improving the nutritional characteristics of dry-fermented deer sausage. Foods 9:1487
- Domínguez R, Bohrer B, Munekata PES et al (2021) Recent discoveries in the field of lipid bio-based ingredients for meat processing. Molecules 26:190
- 3. Domínguez R, Munekata PE, Pateiro M et al (2021) Immobilization of oils using hydrogels as strategy to replace animal fats and improve the healthiness of meat products. Curr Opin Food Sci 37:135–144
- 4. Domínguez R, Pateiro M, Gagaoua M et al (2019) A comprehensive review on lipid oxidation in meat and meat products. Antioxidants 8:429
- Lorenzo JM, Cittadini A, Bermúdez R et al (2015) Influence of partial replacement of NaCl with KCl, CaCl2 and MgCl2 on proteolysis, lipolysis and sensory properties during the manufacture of dry-cured lacón. Food Control 55:90–96

- Raes K, De SS (2009) Fatty acids. In: Nollet LM, Toldra F (eds) Handbook of muscle foods analysis. CRC Press, Boca Raton, FL, pp 161–174
- Pérez-Palacios T, Estévez M (2020) Analysis of lipids and lipid oxidation products. In: Kumar AB, Kumar PM (eds) Meat quality analysis: advanced evaluation methods, techniques, and technologies. Elsevier, pp 217–239
- Folch J, Lees M, Stanley GH (1957) A simple method for the isolation and purification of total lipides from animal tissues. J Biol Chem 226:497–509
- Bligh EG, Dyer WJ (1959) A rapid method of total lipid extraction and purification. Can J Biochem Physiol 37:911–917
- Barros JC, Munekata PES, De Carvalho FAL et al (2020) Use of tiger nut (Cyperus esculentus L.) oil emulsion as animal fat replacement in beef burgers. Foods 9:44



# **Chapter 5**

## **Amino Acids (Free and Hydrolyzed)**

### Olalla López-Fernández, Rubén Domínguez, Mirian Pateiro, Silvina C. Andrés, Paulo E. S. Munekata, Laura Purriños, José Manuel Lorenzo, and Marco Antonio Trindade

#### Abstract

Amino acids are an important compound since they form part of the proteins and are intermediates in the metabolism. Some amino acids are synthesized by the organism, however, in other cases, they should be ingested in the diet (essential amino acids). Because of the high content of amino acids in meat and meat products and their influence on the nutritional and sensorial quality, it is necessary the validation of new techniques to be more sensitive, faster, and versatile for the determination of both free and hydrolyzed amino acids. Despite that there are published several techniques for amino acids extraction, derivatization and for identification/quantification, it should be mentioned that some of them present certain limitations.

The present chapter gives a clear and complete vision of a procedure for the free and hydrolyzed amino acids analysis in meat and meat products. Free amino acid extraction includes the following steps: homogenization and extraction from the meat matrix, deproteinization, derivatization with 6-aminoquinolyl-N-hydroxysuccinimidyl carbamate, and subsequent analysis by liquid chromatography with fluorescence detection (HPLC-FL). Hydrolyzed amino acids are determinate in the following manner: first, the hydrolysis step during 24 h at 110 °C and then the derivatization step with 6-aminoquinolyl-N-hydro-xysuccinimidyl carbamate prior to chromatographic analysis by HPLC-FL. All steps are described in detail so that the conditions proposed by us can be reproduced by other researchers.

Key words Food analysis, Free amino acids, Hydrolyzed amino acids, Derivatization, Liquid chromatography

#### 1 Introduction

Amino acids are considered important compounds not only because they are part of structural units of proteins but also are intermediates in metabolism [1-3]. Some of them serve as precursors for the biosynthesis of neurotransmitters, porphyrins, polyamines, and nitric oxide [4]. They are widely existing in all kinds of food and beverages [3, 5]. In addition to the nutritional aspects, their presence affects the quality of food products with respect to taste, aroma, and color [2, 5].

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_5,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

Amino acids are organic compounds formed by both an amino group and a carboxyl group bound to a specific side chain [1, 6]. In function of the number of carboxylic and amino groups, they are divided into three groups: neutral (e.g., serine), acidic (e.g., glutamic acid), and basic (e.g., arginine) [6]. A total of twenty amino acids form part of proteins, of these there are nine amino acids that are not synthesized by the body and therefore must be introduced into the body through diet. These amino acids are called essential amino acids [1].

In meat and meat products, its quantification is important because these compounds play an important role in the nutritional quality, sensorial characteristics, and acceptability of meat products. Thus, several factors such as the food handling processes, the storage conditions, and the maturation grade affect the proteolytic phenomena, which can change its structure and content producing changes in the sensory of the food [3, 7]. Some amino acids affect the flavor [8] and influence its palatability [9, 10], and also contribute to the formation of amines and volatile compounds [9, 11]. For this, the development of reliable, rapid, and accurate methods of extraction and determination of amino acids is interesting to evaluating the nutritional quality of foods [4, 5]. The extraction of amino acids from the food matrix is the first step to their determination and quantification. Free amino acid extraction includes an extraction step and deproteinization previously to analysis. The extraction consists in the homogenization of the sample in an appropriate solvent, normally hot water, hydrochloric acid 0.01–0.1 N or diluted phosphate buffers [1, 9]. Once homogenized, then the sample is centrifuged at refrigeration and the supernatant is filtered through glass wool [12, 13] or nylon membrane [14] to retain any fat material. Then, the deproteinization process takes place where a portion of the sample is mixed with an organic solvent (methanol, ethanol, or acetonitrile) to precipitate the proteins by denaturation [1]. On the other hand, the extraction of hydrolyzed amino acids consists of a protein hydrolysis step prior to analysis. The hydrolysis of samples can be acid, alkaline, or enzymatic although the more common is the acid hydrolysis normally with hydrochloric acid 6 N at 110 °C during between 20 and 96 h [1].

The first analysis of amino acids had been carried out by ion-exchange chromatography followed by post-column derivatization with ninhydrin and ultraviolet detection. However, this technique is not very selective since amino acids absorb at wavelengths between 190 and 210 nm the same as the majority of solvents and other components of the samples [5]. Due to these, the technique has gradually been supplanted by faster, more sensitive, and versatile methodologies [9], like gas chromatography (GC) [7, 9] and high-performance liquid chromatography (HPLC) [3, 15–17]. Nowadays, in food, the most used method to determine amino acids is reverse phase HPLC [9]. However, to use these new techniques it is necessary a previously derivatization prior to analysis to convert amino acids into more detectable forms. The principal derivation agents are ninhydrin, dansyl chloride, 1-fluoro-2,4-dinitrobenzene, phenylisothiocyanate, orthophthalaldehyde, 9H-fluoren-9-ylmethyl chloroformate, diethyl 2 (ethoxymethylidene)propanedioate, and 6-aminoquinolyl-Nhydroxysuccinimidyl carbamate [5].

With this in mind, the purpose of this book chapter is to thoroughly describe an analytical procedure for determining free and hydrolyzed amino acids in meat and meat products. Four stages are detailed, one for the extraction of free amino acids, another for hydrolyzed amino acids, a stage common to both extractions in which the derivation process is detailed and finally the chromatographic analysis by liquid chromatography.

#### 2 Materials

2.1 Hydrolyzed

Extraction Step

Amino Acids

Prepare all solutions using milli-Q water and/or analytical grade reagents. Prepare and store all reagents at room temperature (unless indicated otherwise). Diligently follow all waste disposal regulations when disposing of waste materials (*see* **Note 1**).

- 1. Hydrochloric acid solution (6 N): To prepare 1 L, measure 497 mL of HCL (37%) and bring it to 1 L with milli-Q water (*see* **Note 2**).
- 2.2 Free Amino Acids
   1. Hydrochloric acid solution (0.1 N): To prepare 1 L, measure 8.28 mL of HCL (37%) and bring it to 1 L with milli-Q water (see Note 2).

All amino acids standards should be analytical grade (preferably HPLC grade analysis) and should be stored at -20 °C.

- 1. Amino acid standard mixture: Amino acid Standard H (Thermo Scientific, ref.: 20088). Ampule of 1 mL that contain eighteen amino acids at a concentration of  $2.5 \pm 0.1 \,\mu mol/mL$  in hydrochloric acid 0.1 N each one, except cystine with a concentration of  $1.25 \pm 0.1 \,\mu mol/mL$ .
- 2. Taurine (2-Aminoethanesulfonic acid)  $\geq$ 99.0% from (Sigma-Aldrich; ref.: T0625). In order to obtain a 2.5 µmol/mL standard solution, dissolve 16.4 mg in 50 mL of 0.1 N HCl.
- 3. L-4-Hydroxyproline  $\geq$ 99.0% from (Fluka; ref.: 56250). In order to obtain a 2.5 µmol/mL standard solution, dissolve 15.6 mg in 50 mL of 0.1 N HCl.
- 4. In order to obtain a six-point calibration curve, different volumes  $(\mu L)$  of the previously described standards [Amino

2.3 Calibration, Amino Acids Derivatization and Identification

#### Table 1

Volume ( $\mu$ L) of the amino acid standards and milli-Q water for the elaboration of the different points of the calibration curve (final volume 1 mL)

Amino acids standard	STD0	STD1	STD2	STD3	STD4	STD5
Amino acids standard (2.5 µmol/mL)	1.25	2.5	5	10	20	40
Hydroxyproline (2.5 $\mu$ mol/mL)	1.25	2.5	5	10	20	40
Taurine (2.5 $\mu$ mol/mL)	1.25	2.5	5	10	20	40
Milli-Q water	996.25	992.5	985	970	940	880

#### Table 2

Amino acids (in order of elution) and final concentration (ng/mL) of each point of the calibration curve

N° #	Amino acid	STD0	STD1	STD2	STD3	STD4	STD5
1	Hydroxyproline	40.98	81.96	163.91	327.83	655.65	1311.30
2	Aspartic acid	41.59	83.19	166.38	332.75	665.50	1331.00
3	Serine	32.84	65.68	131.36	262.73	525.45	1050.90
4	Glutamic acid	45.98	91.96	183.91	367.83	735.65	1471.30
5	Glycine	23.46	46.92	93.84	187.68	375.35	750.70
6	Histidine	48.48	96.97	193.94	387.88	775.75	1551.50
7	Taurine	39.11	78.22	156.44	312.88	625.75	1251.50
8	Arginine	54.44	108.88	217.75	435.50	871.00	1742.00
9	Threonine	37.23	74.45	148.90	297.80	595.60	1191.20
10	Alanine	27.84	55.68	111.36	222.73	445.45	890.90
11	Proline	35.98	71.96	143.91	287.83	575.65	1151.30
12	Cysteine	37.86	75.73	151.45	302.90	605.80	1211.60
13	Tyrosine	56.62	113.24	226.49	452.98	905.95	1811.90
14	Valine	36.61	73.22	146.44	292.88	585.75	1171.50
15	Methionine	46.63	93.26	186.51	373.03	746.05	1492.10
16	Lysine	45.68	91.37	182.74	365.48	730.95	1461.90
17	Isoleucine	40.99	81.98	163.96	327.93	655.85	1311.70
18	Leucine	40.99	81.98	163.96	327.93	655.85	1311.70
19	Phenylalanine	51.62	103.24	206.49	412.98	825.95	1651.90

acid Standard H, Taurine (2.5  $\mu$ mol/mL) and L-4-Hydroxyproline (2.5  $\mu$ mol/mL)] were mixed with milli-Q water (Table 1). The final concentrations (ng/mL) and elution order of each amino acid are those listed in Table 2 (*see* **Note 3**).

	<ol> <li>For derivatization of amino acids, the commercial derivatization kit (based on derivatization with 6-aminoquinolyl-N-hydroxysuccinimidyl carbamate—AQC) supplied by Waters is used (AccQ-Tag Ultra Derivatization Kit; ref. 186003836). This kit includes AccQ-Fluor borate buffer, Reagent powder, and reagent diluent.</li> <li>Mobile phase was prepared from commercial Waters AccQ-Tag Eluent A concentrate (ref. WAT052890). According to the instructions, in order to obtain the correct concentration of mobile phase A, add 200 mL of commercial Eluent A concentrate to 2 L to milli-Q water, and filter the mixture through 0.45 µm membrane prior the HPLC analysis.</li> </ol>
2.4 Liquid Chromatograph	Separation and quantification of amino acids are carried out using high-performance liquid chromatography (Alliance 2695 model, Waters, Milford, MA, USA) equipped with a scanning fluorescence detector (model 2475, Waters, Milford, MA, USA). For the separation of amino acids, a Waters AccQ-Tag column ( $3.9 \times 150$ mm, with a particle size of 3 µm) is used.
3 Methods	
	Carry out all procedures at room temperature unless otherwise specified.
3.1 Hydrolyzed Amino Acids	Hydrolyzed amino acids extraction is done according to the proto- col described by Domínguez et al. [15].
Extraction	<ol> <li>Weigh 100 mg of samples in glass ampoules and add 5 mL of 6 N hydrochloric acid solution [HCl (6 N)] (see Note 4).</li> </ol>
	2. Seal the ampoule with fire and kept at 110 °C for 24 h until hydrolysis of proteins has been completed ( <i>see</i> <b>Note 5</b> ).
	<ul><li>3. After the time, let the vials cool to room temperature (<i>see</i> Note 6).</li></ul>
	4. Dilute $625 \ \mu L$ of the hydrolysate with 25 mL of milli-Q water, mix and then filter through a 0.45 $\mu m$ filter to an Eppendorf (PP syringe filter).
3.2 Free Amino Acids Extraction	Free amino acids analysis is determinate according to the procedure described by Lorenzo et al. [16].
	1. Weigh 5 g of sample in a beaker.
	2. Add 25 mL of hydrochloric acid 0.1 N [HCl (0.1 N)].
	3. Homogenize the samples with Ika Ultra-Turrax for 8 min while cooled by submerging the extract in ice.
	4. Centrifuge the samples at 10,000 $\times g$ for 20 min at 4 °C.

- 5. Filter the supernatant material through glass wool (see Note 7).
- 6. Take 200  $\mu$ L of this extract to an Eppendorf and deproteinize adding 800  $\mu$ L of acetonitrile (ACN).
- 7. Vortex and put in fridge during 30 min.
- 8. Centrifuge the Eppendorfs at  $10,000 \times \text{g}$  for 5 min at 4 °C.
- 9. Filter with 0.45 µm filter (PP syringe filter).

**3.3** Amino Acid The derivatization process is necessary for standards, free and hydrolyzed amino acids. Figure 1 shows in a simplified and schematic way the steps of the free and hydrolyzed amino acid extraction and derivatization. The derivatization is performed as follows:

- 1. Reconstitute the commercial reagent powder for derivatization (6-aminoquinolyl-N-hydroxysuccinimidyl carbamate—AQC) with the reagent diluent (Acetonitrile). For this, take 1 mL of reagent diluent and transfer to reagent powder vial and then shake vigorously with a vortex to the complete dissolution. Finally, incubate exactly 10 min at 55 °C in a heater (*see* Notes 8 and 9).
- 2. Add 10  $\mu$ L of sample or standard to a total recovery vial of 2 mL and 70  $\mu$ L of AccQ-Fluor borate buffer to obtain a pH of 8.8 and vortex.
- 3. Add 20  $\mu$ L of Reagent solution in each sample or standard vial. Then cap the vial and vortex to ensure complete derivatization (*see* **Note 10**).
- 4. Let stand the vials for 1 min at room temperature (RT).
- 5. Heat the samples or standards vials in an oven for exactly 10 min at 55 °C to complete the derivatization (*see* **Note 8**).
- 6. Vortex the vials before the injection (see Note 11).

3.4 Amino Acid Identification and Quantification (HPLC-FL) Separation and quantification of amino acids are carried out using a liquid chromatograph coupled with a fluorescence detector following the next conditions (Table 3):

- 1. Ten microliters of the sample (or the standard) is injected (*see* **Note 12**). A Waters AccQ-Tag column  $(3.9 \times 150 \text{ mm}, \text{with a particle size of 3 } \mu\text{m})$  is used for amino acid separation. The temperature of the column oven is adjusted at 37 °C.
- 2. The mobile phase composition and the gradient were defined according to the validated and patented method of Waters Corporation (AccQ-Tag Amino acid analysis protocol), with minor changes. Three solvents make up the mobile phase: (A) AccQ Tag Eluent A solution for amino acids analysis (Waters, Milford, MA, USA), (B) acetonitrile (HPLC grade), and (C) ultra-pure water (Milli-Q). The flow rate is 1.0 mL/min and the solvent gradient was set as follows: 0.0–0.5 min



Fig. 1 Schematic and simplified representation of the steps of the amino acids (free and hydrolyzed) extraction and derivatization phases

Parameters	Values					
Injection volume	10 µL					
Flow rate	l mL/min					
Column	AccQ-tag colum	in $(3.9 \times 150$ mm, with a	particle size of 3 µm)	, waters		
Oven temperature	37 °C					
Mobile phase	AccQ tag eluent Acetonitrile (HI	A solution for amino acids PLC grade) and ultra-pure	analysis (waters, Milfo water.	ord, MA, USA),		
Gradient elution	Time (min) 0.0 0.5 17 22 24 31.5 36 42 44	AccQ tag eluent (%) 99 99 96 95 91 83 83 0 99	Acetonitrile (%) 1 4 5 9 17 17 60 1	Water (%) 0 0 0 0 0 0 0 40 0		
Detector	Fluorescence de	tector				
Excitation wavelength	250 nm					
Emission wavelength	395 nm					
Retention time	Compound Hydroxyproline Aspartic acid Serine Glutamic acid Glycine Histidine Taurine Arginine Threonine Alanine Proline Cysteine Tyrosine Valine Methionine Lysine Isoleucine Leucine Phenylalanine		Retention time (min 10.04 12.05 13.71 14.84 16.17 17.33 21.45 22.55 23.08 25.07 27.13 29.94 30.01 30.97 31.37 33.31 34.05 34.55 35.65	.)		
Run time	45 min					

# Table 3 Summary of the chromatographic conditions used for the amino acids analysis


Fig. 2 Amino acids chromatograms of a standard (a) and a fresh meat sample (b)

99% A and 1% B; 0.5–17.0 min 99% A and 1% B; 17.0–22.0 min 96% A and 4% B; 22.0–24.0 min 95% A and 5% B; 24.0–31.5 min 91% A and 9% B; 31.5–36.0 min 83% A and 17% B; 36.0–42.0 min 83% A and 17% B; 42.0–44.0 min 60% B and 40% C; 42.0–45.0 min 99% A and 1% B.

3. The detection of amino acids is carried out using a fluorescence detector (FL). The wavelengths for excitation and emission are set at 250 and 395 nm, respectively. The total time for chromatographic analysis is 45 min.

Data acquisition, equipment control, and data analysis are carried out using the HPLC software (in our case Empower 3TM; Waters, Milford, MA, USA). The amino acids in meat products are identified by comparing their retention time with the authenticated standards. Figure 2 shows an example of chromatograms of standard (a) and meat sample (b). The results are expressed as mg/100 g of sample.

#### 4 Notes

- 1. Due to the toxic and/or corrosive nature of the solvents and reagents used in the extraction and methylation of fatty acids, it is necessary for the operator to take the necessary protective measures (gloves, glasses, etc.) as well as to carry out all the operations in laboratory fume hoods.
- 2. Add the hydrochloric acid slowly to the water, shake, and finally make the volumetric flask up to the mark with water (use a hood to make these solutions).
- 3. The pipetting process must be careful due to the small quantities to take (few  $\mu$ L). The tip of the micropipettes should be moistened with the solvent (or standard) to ensure that the correct amount of each standard is taken.
- 4. An analytical balance with 0.1 mg resolution or higher should be used to weigh the tubes.
- 5. Preheat an oven at 110 °C.
- 6. Wait a reasonable time to avoid burning yourself when handling the vials. Use gloves or suitable utensils to open the vials safely, and avoid possible cuts.
- 7. The filtered extract can be saved at -20 °C until use.
- 8. Preheat an oven at 55 °C.
- 9. It is important to dilute completely the reagent powder for a correct derivatization.
- 10. Make sure that there are no bubbles in the vial insert to ensure the correct derivatization process.
- 11. Shake vigorously to avoid bubbles in the vial insert that may interfere with the injection.
- 12. Standards should be injected first, for identification and to create calibration curves for each amino acid. During sample analysis, a blank (acetonitrile) should be injected every 10 samples. Calibrations should be performed each time the reagent powder for derivatization is prepared or when performing any chromatograph maintenance that may affect the resolution of the equipment or the detector signal.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568).

#### References

- 1. Nollet LM, Toldrá F (2012) Handbook of analysis of active compounds in functional foods. CRC Press, Boca Raton, FL
- Janiszewska K, Aniołowska M, Nowakowski P (2012) Free amino acids content of honeys from Poland. Pol J Food Nutr Sci 62:85–89
- Xie Y, Luo T, Yang J et al (2018) Rapid determination of amino acids in beer, red wine, and donkey-hide gelatin by gradient elution of HPLC: from micellar liquid chromatography to high submicellar liquid chromatography. J AOAC Int 101:249–255
- 4. Guo S, Duan JA, Qian D et al (2013) Rapid determination of amino acids in fruits of Ziziphus jujuba by hydrophilic interaction ultrahigh-performance liquid chromatography coupled with triple-quadrupole mass spectrometry. J Agric Food Chem 61:2709–2719
- 5. Callejón RM, Troncoso AM, Morales ML (2010) Determination of amino acids in grape-derived products: a review. Talanta 81: 1143–1152
- 6. Dołowy M, Pyka A (2013) Application of TLC, HPLC and GC methods to the study of amino acid and peptide enantiomers: a review. Biomed Chromatogr 28:84–101
- Leggio A, Belsito EL, De Marco R et al (2012) Simultaneous extraction and derivatization of amino acids and free fatty acids in meat products. J Chromatogr A 1241:96–102
- Chiang PD, Yen CT, Mau JL (2007) Non-volatile taste components of various broth cubes. Food Chem 101:932–937
- 9. Pérez-Palacios T, Melo A, Cunha S et al (2013) Determination of free amino acids in coated foods by GC–MS: optimization of the

extraction procedure by using statistical design. Food Anal Method 7:172–180

- Baryłko-Pikielna N, Kostyra E (2007) Sensory interaction of umami substances with model food matrices and its hedonic effect. Food Qual Prefer 18:751–758
- Cornet M, Bousset J (1999) Free amino acids and dipeptides in porcine muscles: differences between red and white muscles. Meat Sci 51: 215–219
- 12. Pérez-Palacios T, Ruiz J, Barat JM et al (2010) Influence of pre-cure freezing of Iberian ham on proteolytic changes throughout the ripening process. Meat Sci 85:121–126
- 13. Jiménez-Martín E, Ruiz J, Pérez-Palacios T et al (2012) Gas chromatography–mass spectrometry method for the determination of free amino acids as their dimethyl-tert-butylsilyl (TBDMS) derivatives in animal source food. J Agric Food Chem 60:2456–2463
- 14. Nimbalkar MS, Pai SR, Pawar NV et al (2012) Free amino acid profiling in grain Amaranth using LC-MS/MS. Food Chem 134: 2565–2569
- Domínguez R, Borrajo P, Lorenzo JM (2015) The effect of cooking methods on nutritional value of foal meat. J Food Compos Anal 43: 61–67
- 16. Lorenzo JM, Cittadini A, Bermúdez R et al (2015) Influence of partial replacement of NaCl with KCl, CaCl2 and MgCl2 on proteolysis, lipolysis and sensory properties during the manufacture of dry-cured lacón. Food Control 55:90–96
- Munekata PE, Pateiro M, Domínguez R et al (2020) Nutritional characterization of sea bass processing by-products. Biomol Ther 10:232



# **Chapter 6**

# Cholesterol

## Olalla López-Fernández, Rubén Domínguez, Claudia Ruiz-Capillas, Mirian Pateiro, María Elena Sosa-Morales, Paulo E. S. Munekata, Anderson S. Sant'Ana, José Manuel Lorenzo, and Ana M. Herrero

#### Abstract

The cholesterol determination in meat and meat products is very important because these products are highly demanded. The intake of high levels of cholesterol is associated with several diseases. Therefore, the development of analytical techniques is necessary to inform the consumer about the meat and meat products nutritional quality.

Cholesterol determination in meat and meat product was studied for decades. The first published studies were based on a previous lipid extraction followed by saponification before its quantification. Nowadays, new methods have been published that simplify the process by making direct saponification with KOH and ethanol. These methods allow a faster determination of cholesterol with less consumption of solvents. The most common chromatographic techniques for determining and quantified cholesterol in meat and meat products are gas chromatography and high-performance liquid chromatography.

This chapter gives a clear and complete vision of a rapid and precise methodology for the cholesterol analysis of meat and meat products, including all the extraction steps and the subsequent analysis by high-performance liquid chromatography with Photodiode Array detection (HPLC-DAD) using a normal phase. The conditions of extraction and analysis proposed by us are described afterwards with exactitude with the aim reproduced by other researchers.

Key words Food analysis, Cholesterol, Normal phase, HPLC-DAD

#### 1 Introduction

Cholesterol is an essential lipid molecule that is commonly found as a component in the cell membrane [1]. One of the major functions is to participate in the biosynthesis of bile acids in the liver [2, 3]. It is an important precursor for bile acids, provitamin D, and several steroidal hormones as testosterone and estrogen [2–5]. Moreover, they also take part in the production and absorption of vitamin D [3]. Due to all the aforementioned, the intake of low cholesterol levels could have benefits for human health [3]. However, high cholesterol levels are associated with a higher risk of

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_6,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2021

hypercholesterolemia, obesity, diabetes, brain diseases, and cardiovascular diseases [3, 6, 7].

Cholesterol is mostly present in meat, meat products and derived products from livestock as eggs, cheese, milk, butter, etc. [4]. The consumption of meat and meat products provide one third to one half of daily-recommended cholesterol intake [6]. To prevent the appearance of diseases associated with cholesterol, it is necessary accurate determination of cholesterol. The development of analytical techniques for the determination and quantification of total cholesterol in meat and meat products is important mainly to inform of their quality and safety [8].

The AOAC International adopted the first validated cholesterol determination method for foods (AOAC International 976.26, [9]). This method was based on a prior lipid extraction followed by saponification, which makes it a long and tedious method. Moreover, this extraction method increases the analysis time and the consumption of dissolvents. In order to overcome these drawbacks, the method was modified, by employing direct saponification (AOAC International 994.10, [10-12]). Nowadays, new extraction methods based on direct saponification followed by extraction with organic solvents were validated with good results [4, 8, 11, 13]. The saponification step is indispensable to separate cholesterol and other unsaponifiable materials from fatty acids and to remove triglyceride interferences [12, 14]. The most suitable direct saponification conditions are the use of an alcoholic KOH solution (between 0.33 and 0.5 M) and temperatures ranging from 55 to 75 °C during 15 to 60 min [11, 12], also known as hot saponification. Despite, cold saponification (at room temperatures) procedures during several hours (overnight) could also be used [15], to limit the cholesterol degradation during heating. It is important to highlight that in hot saponification, the addition of antioxidants (i.e., Vitamin C) and the use of anaerobic conditions (air removed with nitrogen or argon) during saponification and processes contribute to preventing cholesterol extraction oxidation [6].

For cholesterol analysis several, analytical methods have been developed, including spectroscopic and gravimetric procedures, enzymatic assays, and chromatographic methods based on gas chromatography, liquid chromatography, and mass spectrometry [1, 3]. In food, the chromatographic methods were more reliable and selective because of avoidance of the interference from other sterols [1, 4, 12, 14]. One of the most used methods in meat cholesterol analysis is based on liquid chromatography coupled to photodiode array detector [4, 8, 15].

For cholesterol analysis by HPLC, both normal phase (NP) and reversed phase (RP) can be used [13, 15]. However, due to its low polarity, the use of normal phase is preferable. In this regard, some authors proposed the cholesterol analysis by normal phase

(NP) using a polar stationary phase (silica) and an apolar mobile phase (hexane) [13]. The normal phase analysis provides better chromatograms than reversed phase methods due to the low viscosity of commonly used eluents.

Thus, the aim of this book chapter is to explain in detail a procedure for determining cholesterol in meat and meat products. The extraction consists of direct saponification (hot saponification) of the meat sample with ethanolic KOH solution, a posterior extraction with hexane, and finally the chromatographic analysis by liquid chromatography.

#### 2 Materials

Prepare all solutions using distilled water and/or analytical grade reagents. Use HPLC-grade hexane. Prepare and store all reagents at room temperature (unless indicated otherwise). Diligently follow all waste disposal regulations when disposing of waste materials (*see* **Note 1**).

- 2.1 Extraction Step
  1. Saponification solution: 11% w/v potassium hydroxide (KOH) in a mixture of 55% v/v absolute ethanol (EtOH) and 45% v/v distilled water (H<sub>2</sub>O). For 1 L of solution dissolve 129.4 g of KOH (85% purity) in 450 mL of distiller water and then add 550 mL of absolute ethanol. The saponification solution must be prepared/renewed every week.
  2. BHT (25 ppm) solution in n-hexane: Dissolve 0.0125 g of butylated hydroxytoluene (BHT) in 500 mL of n-hexane.
  2.2 Calibration and Cholesterol
  1. Cholesterol (≥ 99%): For a concentration of 1 mg/mL, dissolve 0.05 g of cholesterol in 50 mL of n-hexane.
  2. Standards for calibration curve: In order to obtain a seven-
  - Standards for calibration curve: In order to obtain a sevenpoint calibration curve, different volumes (μL) of the previously described cholesterol standard were mixed with n-hexane (Table 1). The final concentrations (ppm) of calibration curve standards are those listed in Table 2.

#### Table 1

Volume ( $\mu$ L) of the cholesterol standard solution (1 mg/mL) and hexane for the elaboration of the different points of the calibration curve (final volume 1500  $\mu$ L)

Compound	STD0	STD1	STD2	STD3	STD4	STD5	STD6
Cholesterol (1 mg/mL)	23.44	46.88	93.75	187.50	375	750	1500
Hexane (HPLC grade)	1476.56	1453.13	1406.25	1312.50	1125	750	0

Table 2					
Final concentration (ppm) of	cholesterol in	each poin	t of the	calibration	curve

Concentration (ppm)	STD0	STD1	STD2	STD3	STD4	STD5	STD6
Cholesterol	0.15625	0.3125	0.625	1.25	2.5	5	10

2.3 HPLC System Separation and quantification of cholesterol are carried out using a liquid chromatograph model Alliance 2695 equipped with a 996 Photodiode Array Detector (Waters, Milford, MA, USA). The cholesterol separation was performed using a normal phase silica column (SunFireTM Prep Silica, 4.6 mm ID × 250 mm, 5 μm particle size, Waters, Milford, MA, USA). Empower 3TM advanced software (Waters, Milford, MA, USA) was used to control system operation and analyze results.

#### 3 Methods

Carry out all procedures at room temperature unless otherwise specified.

# **3.1** CholesterolThe extraction was carried out according to the protocol described<br/>by Domínguez et al. [13] with minimal changes. Figure 1 shows a<br/>simplified and schematic way of the extraction phases.

- 1. Weigh 0.50  $\pm$  0.02 g of homogenized meat sample in a screw Teflon-lined cap tube of 15 mL.
- 2. Add 0.25 g of L-ascorbic acid (to avoid the oxidative process) and 5 mL of saponification solution.
- 3. Apply nitrogen gas to eliminate the air from the reaction by displacement and then close the tube (*see* **Note 2**).
- 4. Shake until the ascorbic acid is completely dissolved with a vortex (approximately 30 s), then the samples are left to rest for 5 min and finally they are stirred again for another 30 s.
- 5. The saponification is carried out in a shaking water bath (THER-SPIN, Orto Alresa, Madrid, Spain) (200 rpm) at 85 °C for 45 min. After 20 min of the saponification process, the samples were vortexed. At the end of the saponification process, the samples were cooled at room temperature (about 10 min) (*see* Note 3).
- 6. Follow cooling, 1.5 mL of distilled water and 3 mL of 25 ppm of BHT solution in n-hexane are added and they are vortexed vigorously.
- 7. The samples are transferred to conical centrifuge tubes of 15 mL capacity and are centrifuged at  $1500 \times g$  for 3 min (*see* **Note 4**).
- 8. An aliquot of n-hexane (upper layer) is transferred into another tube. In order to ensure the complete dehydration of the



Fig. 1 Schematic and simplified representation of the steps of the cholesterol extraction phases

organic phase, a spatula tip of anhydrous sodium sulfate are added.

 Finally, the tubes are briefly shaken (vortexed), and an aliquot of the n-hexane was filtered through a 0.45-µm hydrophobic membrane into an amber screw-cap vial with Teflon septum.

Separation and quantification of cholesterol are carried out using a liquid chromatograph coupled with a diode array detector following the next conditions (Table 3):

- 1. Ten microliters of the sample (or the standard) are injected. A SunFireTM Prep Silica, 4.6 mm ID  $\times$  250 mm, 5 µm particle size, (Waters) is used for cholesterol separation. The temperature of the column oven is adjusted at 30 °C.
- 2. The mobile phase is adjusted at 2% 2-propanol and 98% n-hexane. The isocratic mode is used during chromatographic analysis at a flow rate of 1 mL/min.
- 3. The detection of cholesterol are carried out using Photodiode Array detector (DAD) at 208 nm. The total time for chromatographic analysis is 15 min.

Data acquisition, equipment control, and data analysis are carried out using the HPLC software (in our case Empower 3TM; Waters). The cholesterol in meat products is identified by comparing its retention time with the authenticated standards (Cholesterol  $\geq$  99%) (*see* Notes 5 and 6). Figure 2 shows examples of

3.2 Cholesterol Identification and Quantification (HPLC-DAD)

#### Table 3

#### Summary of the chromatographic conditions used for the cholesterol analysis

Parameters	Values
Injection volume	10 µL
Flow rate	l mL/min
Column	SunFireTM prep silica, 4.6 mm ID $\times$ 250 mm, 5 $\mu m$ particle size, waters
Oven temperature	30 °C
Mobile phase	2-propanol/hexane (2:98, v/v)
Elution type	Isocratic
Detector	Photodiode Array detector
Wavelength	208 nm
Retention time	10 min
Run time	15 min



Fig. 2 Cholesterol chromatogram of a standard (a) and a fresh meat sample (b)

chromatograms of standard (a) and sample (b). The results are expressed as mg/100 g of meat.

#### 4 Notes

- 1. Due to the toxic and/or corrosive nature of the solvents and reagents used in the extraction, it is necessary for the operator to take the necessary protective measures (gloves, glasses, etc.) as well as to carry out all the operations in a laboratory fume hood.
- 2. Apply a low pressure of nitrogen to avoid losses due to splashes.
- 3. The water batch should be turned on before (about 15 min) to allow the water bath to have the correct saponification temperature.
- 4. The emulsification of some samples may occur during the extraction process. If this happens, 0.2 mL of absolute ethanol can be added to facilitate the separation process. The ethanol is gently mixed in swirling motion, and the emulsifying mixture allowed to stand to enable the separation.
- 5. Standards should be injected first, for identification and to create the calibration curve. Identification/calibration standards must be prepared each time a calibration is performed. During the analysis of the samples, a blank (hexane) should be injected every 10 samples. It is recommended to inject a standard to verify retention times and the concentration each time a new sequence begins. Calibrations should be performed at least once a month, or when any maintenance is performed on the chromatograph that may affect the resolution of the equipment or the detector signal.
- 6. Standards prepared for calibrations can be stored frozen  $(-20 \ ^\circ\text{C})$  and used to check retention times. However, these standards cannot be used for calibration, only to verify retention times. Calibration curve standards must be prepared each time a new calibration is performed.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568).

#### References

- Osman H, Yap KC (2006) Comparative sensitivities of cholesterol analysis using GC, HPLC and spectrophotometric methods. Mal J Anal Sci 10:205–210
- 2. Boyer JL (2013) Bile formation and secretion. Compr Physiol 3:1035e78
- Li LH, Dutkiewicz EP, Huang YC et al (2019) Analytical methods for cholesterol quantification. J Food Drug Anal 27:375–386
- 4. Albuquerque TG, Oliveira MBP, Sanches-Silva A et al (2016) Cholesterol determination in foods: comparison between high performance and ultra-high performance liquid chromatography. Food Chem 193:18–25
- Connor WE, Connor SL (2002) Dietary cholesterol and coronary heart disease. Curr Atheroscler Rep 4:425–432
- Czauderna M, Kowalczyk J, Niedzwiedzka KM (2009) Simple HPLC analysis of tocopherols and cholesterol from specimens of animal origin. Chem Anal 54:203–214
- 7. Stroher GL, Rodrigues AC, Dias LF et al (2012) Comparative analysis and validation methodologies of GC and HPLC for analysis of cholesterol in meat products. Am J Anal Chem 3:306–311
- Prates JAM, Quaresma MAG, Bessa RJB et al (2006) Simultaneous HPLC quantification of total cholesterol, tocopherols and β-carotene in Barrosã-PDO veal. Food Chem 94:469–477
- 9. AOAC International (1996) AOAC official method 976.26: cholesterol in multi-

component foods-gas chromatographic. Association of Official Analytical Chemists, Arlington

- AOAC International (1996) AOAC Official-Method 994.10: cholesterol in foods. Direct saponification-gas chromatographic method. Association of Official Analytical Chemists, Arlington
- 11. Dinh TTN, Blanton JR Jr, Brooks JC et al (2008) A simplified method for cholesterol determination in meat and meat products. J Food Compos Anal 21:306–314
- 12. Dinh TT, Thompson LD, Galyean ML et al (2011) Cholesterol content and methods for cholesterol determination in meat and poultry. Compr Rev Food Sci 10:269–289
- 13. Domínguez R, Barba FJ, Centeno JA et al (2018) Simple and rapid method for the simultaneous determination of cholesterol and retinol in meat using normal-phase HPLC technique. Food Anal Method 11:319–326
- Bragagnolo N (2009) Cholesterol and cholesterol oxides in meat and meat products. In: Nollet LM, Toldra F (eds) Handbook of muscle foods analysis. CRC Press, Boca Raton, FL, pp 187–219
- 15. Domínguez R, Martínez S, Gómez M et al (2015) Fatty acids, retinol and cholesterol composition in various fatty tissues of Celta pig breed: effect of the use of chestnuts in the finishing diet. J Food Compos Anal 37: 104–111



# **Chapter 7**

# **Mineral Profile**

### Laura Cutillas, Olalla López-Fernández, Rubén Domínguez, Mirian Pateiro, Adriana Pazos, Paulo E. S. Munekata, Laura Purriños, José Manuel Lorenzo, Andrea Carla da Silva Barretto, and José Angel Pérez-Álvarez

#### Abstract

In recent years, there has been an increase in consumer interest in nutritional and health aspects, which has affected the consumption of animal products. Meat is a source of essential minerals for humans, among which iron stands out, as well as others, such as zinc, phosphorus, magnesium, and potassium. Moreover, meat products can contain high levels of NaCl, and this can be harmful to health. Therefore, it is very important to have an appropriate method for a correct determination of the mineral content in meat and meat products.

Although a wide variety of analytical methods are available for the analysis of minerals and trace elements in food products, Inductively Coupled Plasma Optical Emission Spectroscopy (ICP-OES) is one of the most used techniques, as it allows the simultaneous analysis of a large number of elements.

Taking this into account, this chapter is intended to describe in detail a procedure for mineral analysis of meat and meat products, including determination of total ash and analysis of mineral content by ICP-OES, so that it can be reproduced by other researchers.

Key words Mineral content, Meat, Calcium, Sodium, Iron, Magnesium, Phosphorous, Food analysis, Inductively coupled plasma optical emission spectroscopy (ICP-OES)

#### 1 Introduction

In the last few decades, nutritional habits have notably changed. Specifically, meat consumption has been on the rise since the 1960s in most of countries, but especially from 1980 to the present [1–4]. Moreover, a trend has been observed in recent years: consumers concern for their health and their interest in the nutritional aspects of food has increased, which is also reflected in the consumption of meats and meat products [4, 5].

Meat provides important nutrients for the human diet, mainly proteins, vitamins, and minerals. Minerals have very varied

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_7,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

functions in our body, such as electrolytes, as enzymatic constituents and as building materials (in bones and teeth). They are divided into macro minerals, trace elements (microminerals), and ultra-trace elements. The main elements (Na, K, Ca, Mg, Cl, P, S) are essential for humans in amounts >50 mg/day and trace elements (Fe, I, F, Zn, Se, Cu, Mn, Cr, Mo, Co, Ni) are essential in concentrations of <50 mg/day. Health authorities in most countries have established recommendations for daily intake levels of these essential minerals [6–8].

The mineral content in meat depends on factors like the species, breed, rearing practices, feeds composition, slaughter age, and the cut [8, 9]. According to Hermida et al. [10], the macro minerals and trace elements concentrations in tissues depend on the type of cut, the age of the animals, and other factors. Greenfield and Southgate [11] concluded that the lean/fat tissue ratio affects the levels of most nutrients, which are distributed differently in the two fractions. In addition, thermal processes can change the mineral content of meat [12].

One of the most relevant minerals in meat is iron as it is highly bioavailable. The form in which this iron is found (mainly in heme form, which is bound with myoglobin and hemoglobin) allows 20–30% of it to be absorbed. In addition, the presence of meat favors the absorption of iron forms from vegetables [13]. Another important contribution of meat is zinc as well as magnesium, potassium, copper, and other minerals and microelements [14].

On the other hand, meat products (especially dry-cured products like ham, sausages, and bacon) may contain high concentrations of sodium chloride (NaCl), since salting with NaCl is widely employed to preserve these products, contributes to achieve a characteristic flavor, gives microbial stability and improves proteins solubility [15–17]. However, an excessive consumption of salt is not advisable. Several researches have revealed that high levels of NaCl intake can raise blood pressure and promote certain diseases, which is why it is currently intended to offer consumers healthier food products with low salt content and without losing quality [15, 16]. Potassium chloride is the most common salt substitute in meat products, although calcium and magnesium chlorides also can be another alternative [17].

Considering this, it seems clear that it is important to develop a suitable method for determining the mineral content in meat and meat products.

Nowadays, there are a great variety of analytical methods for the analysis of minerals and trace elements in food. The most commonly used methods include: spectrophotometry, fluorometry, atomic absorption spectrometry (AAS), flame atomic absorption spectrometry (FAAS), graphite furnace atomic absorption spectrometry (GFAAS), hydride generation atomic absorption spectrometry (HGAAS), inductively coupled plasma optical emission spectroscopy (ICP-OES), and inductively coupled plasma mass spectrometry (ICP-MS). The choice of analytical method generally depends on available instrumentation, laboratory experience, and analyte concentration levels [18].

Among all techniques, the use of inductively coupled plasma optical emission spectroscopy (ICP-OES) to identify and quantify minerals in meat and meat products is frequent. It is based on the measurement of the radiation of the spectral line emitted by excited atoms in an Ar plasma generated by inductive heating with a highfrequency electromagnetic field.

ICP-OES is one of the most widely used techniques around the world to determine elements in a wide variety of samples (which must be previously digested) as it allows the simultaneous analysis of a large number of elements [18, 19].

The method we have developed includes two main steps: (1) total ash determination, followed by (2) mineral content determination by ICP-OES.

With all the above, the present book chapter aims to clearly and completely describe a procedure for determining minerals elements in meat and meat products.

#### 2 Materials

Prepare all solutions using mili-Q water and analytical grade reagents, and store stock and standard solutions at refrigerate temperature  $(4 \ ^{\circ}C)$  (see Note 1). Follow all waste disposal regulations when disposing of waste materials. Total Ash and Analytical balance (see Note 2), porcelain crucibles, desiccator for 2.1 storing them, and a muffle furnace for incinerate samples are Mineral Content needed for total ash determination. Determination For mineral content determination, these are required: 110 mm quantitative paper filters (FILTER-LAB, Barcelona, Spain), plastic funnels, racks and plastic tubes for ICP-OES autosampler. 1 M HNO<sub>3</sub> solution is required. For this, mix 69.23 mL of nitric acid (65%) (Suprapur or Ultratrace grade) with 900 mL of miliQ water. 2.1.1 Inductively Coupled The quantification of mineral elements is performed using a Perki-Plasma Optical Emission nElmer Avio 200 optical emission spectrometer (PerkinElmer, Spectroscopy (ICP-OES) Massachusetts, USA) with an autosampler (S23, PerkinElmer). The main components of an ICP-OES instrument are the

The main components of an ICP-OES instrument are the plasma torch, the nebulizer, and the polychromator. The torch consists of three concentric quartz tubes surrounded by a watercooled induction coil connected to a high-frequency generator. Plasma is created by making Ar conductive by exposing it to an electrical discharge that creates electrons and ions. Under the influence of the high-frequency electromagnetic field, the charged particles heat the argon until the plasma reaches temperatures above 9700 °C. This leads to almost complete vaporization of the analyte and high atomization efficiency. The sample is introduced through the nebulizer into the torch using a transporter flow of Ar [18].

ICP-OES is also equipped with a radio frequency source set of 27.12 MHz, a peristaltic pump, a spraying chamber, and a concentric spray nebulizer (MEINHARD glass nebulizer, type K1), using 99.996% liquid argon plasma gas (Praxair, Madrid, Spain).

A 2% nitric acid solution is needed for equipment cleaning between sample measurements.

2.1.2 Calibration Stock solutions at 1000 mg/L for Ca, K, Mg, Na, P, Fe, Mn, Zn, and Cu (SCP-SCIENCE, Courtaboeuf, France) were used for preparing the standard solutions in 1 N HNO<sub>3</sub>, v/v.

The concentration ranges of the standard solutions were: 5–40 mg/L for Ca and Zn, 0.5–2.5 mg/L for Cu and Mn, 2–15 mg/L for Fe, and 50–200 mg/L for K, Mg, Na and P (*see* **Note 3**). Both, final concentrations (mg/L) of each mineral and the volume ( $\mu$ L) of each stock solution (1000 mg/L) used for the preparation of 50 mL of each standard are those listed in Tables 1 and 2 (*see* **Note 4**).

Table 1					
<b>Concentration ranges</b>	of the st	andard solution	s (prepared	in 1	n hno <sub>3</sub> )

	Mineral standard (mg/L) <sup>a</sup>				Adjust to 50 mL with $HNO_3$ (1 N) <sup>b</sup>			
	Std1	Std2	Std3	Std4	μ <b>L (Std1)</b>	μ <b>L (Std2)</b>	μ <b>L (Std3)</b>	μ <b>L (Std4)</b>
Ca	5	10	25	40	250	500	1250	2000
Cu	0.5	1	1.5	2.5	25	50	75	125
Fe	2	5	10	15	100	250	500	750
Κ	50	100	150	200	2500	5000	7500	10,000
Mg	50	100	150	200	2500	5000	7500	10,000
Mn	0.5	1	1.5	2.5	25	50	75	125
Na	50	100	150	200	2500	5000	7500	10,000
Р	50	100	150	200	2500	5000	7500	10,000
Zn	5	10	30	40	250	500	1500	2000

Std. standard

<sup>a</sup>Final concentrations (mg/L) of each element in the standards used for calibration

<sup>b</sup>Volume ( $\mu$ L) of the stock solutions (1000 mg/L) used for the preparation of 50 mL of each standard

(Std5)

25,000

(Std6)

50,000

(Std4)

10,000

Sodium concentration ranges of the standard solutions (prepared in 1 N $\rm HNO_3$ )								
Na standard (mg/L) <sup>a</sup> Adjust to 50 mL with HNO <sub>3</sub> (1 N) <sup>b</sup>								
				ы	1			

(Std2)

5000

Table 2 Sodium concentration ranges of the standard solutions (prepared in 1 N  $HNO_3$ )

Na 50 *Std* standard

100

<sup>a</sup>Final concentrations (mg/L) of sodium in the standards used for calibration

500

Std1 Std2 Std3 Std4 Std5 Std6

200

150

<sup>b</sup>Volume ( $\mu$ L) of the Na stock solution (1000 mg/L) used for the preparation of 50 mL of each standard level

(Std1)

1000 2500

#### 3 Methods

Three-gram  $(3.00 \pm 0.05 \text{ g})$  samples are weighed into porcelain crucibles for mineral analysis. Afterward, the samples are incinerated in a furnace at 450 °C for 12 h (using the protocol ISO 936: 1998 [20]). Then, the crucibles are stored at room temperature in the desiccator until constant weight. The amount of ash obtained is calculated by the difference between the crucible with the sample after incinerating—empty crucible. The ash obtained is dissolved in 10 mL of 1 M HNO<sub>3</sub> and filtered to plastic tubes (Fig. 1).

(Std3)

7500

The next step is the mineral content determination using inductively coupled plasma optical emission spectroscopy (ICP-OES) in order to determine the concentration of Ca, K, Mg, Na, P, Fe, Mn, Zn, and Cu following the protocol defined by Lorenzo et al. [21] with modifications (*see* Note 5). Operating conditions of the ICP-OES equipment are: vertical plasma, argon pressure: 6.5 bar, reflected power: 1400 W, sample flow rate: 1 mL/min, nebulizer gas flow: 0.7 L/min, auxiliary argon flow: 0.2 L/min, main argon flow: 10 L/min, purge gas flow: 1.5 L/min, reading time: 2 seconds, height of vertical observation: 15 mm and radial torch configuration.

The operating wavelengths are: Ca, 317.933 nm; K, 404.721 and 766.490 nm; Mg, 279.077 and 285.213 nm; Na, 589.592 nm; P, 213.617 nm; Fe, 238.204 and 239.562 nm; Mn, 257.610 nm; Zn, 206.200 and 213.857 nm; and finally Cu, 324.752 and 327.393 nm (*see* **Note 6**).

Figure 2a and b shows the calibration curves obtained for Ca, Cu, Fe, Mg Na, Zn, Mn, and P at different operating wavelengths (*see* **Note** 7).

*Syngistix* ICP software is used to control the system and acquire the data. The external standard procedure is used to determine the concentration of each mineral (*see* **Note 8**).

# Total ash and mineral content determination



Fig. 1 Schematic and simplified representation of the steps of the total ash and mineral content determination

An example of the spectra obtained for Ca, Mg, and Na at the corresponding wavelength in a meat product sample is shown in Fig. 3. The final value for each element is obtained by calculating the average of two determinations using the Eq. 1, and the results are expressed as mg/100 g of meat.

Sample mineral content 
$$(mg_{100 g meat}) = \frac{\text{ICP mineral content}(\frac{mg}{L})}{\text{Fresh sample weight }(g)}$$
 (1)



**Fig. 2** (a) Calibration curves for Ca, Cu, Fe, and Mg at different operating wavelengths. (b) Calibration curves for Na, Zn, Mn, and P at different operating wavelengths





#### 4 Notes

- 1. Due to the corrosive nature of the reagent used, it is necessary for the operator to take the necessary protective measures (plastic gloves) as well as to carry out all the operations in laboratory fume hoods.
- 2. An analytical balance with 0.1 mg resolution or higher should be used to weigh the porcelain crucibles.



Fig. 3 Spectra for Ca, Mg, and Na at corresponding wavelength in a meat product sample

- 3. Due to the extremely small quantities (few  $\mu$ L) needed to prepare standard solutions for elements like Cu and Mn, the pipetting process must be scrupulously careful (Tables 1 and 2).
- 4. Stock solutions and standards prepared for calibrations can be stored refrigerated (4 °C).
- 5. Standard solutions and samples must be vortexed before measurement.
- 6. Standards should be injected first, for identification and to create calibration curves for each element at the corresponding wavelength (Fig. 2a, b).
- 7. It must be verified that there are no interferences that compromise the accuracy of the analytical result. Interferences arise due to differences in composition of the analyzed sample and the external standards and blanks used for calibration.
- 8. Calibration should be performed every day and/or when any change is made in the conditions of the equipment.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568).

#### References

- 1. Basu S (2015) The transitional dynamics of caloric ecosystems: changes in the food supply around the world. Crit Public Health 25: 248–264
- 2. Katare B, Wang HH, Lawing J et al (2020) Toward optimal meat consumption. Am J Agric Econ 102:662–680
- 3. FAO (2018) FAOSTAT. www.fao.org/ faostat/en/?#data
- 4. González N, Marquès M, Nadal M et al (2020) Meat consumption: which are the current global risks? A review of recent (2010-2020) evidences. Food Res Int 137:109341
- 5. Fonseca MCP, Salay E (2008) Beef, chicken and pork consumption and consumer safety and nutritional concerns in the City of Campinas, Brazil. Food Control 19:1051–1058
- 6. Belitz HD, Grosch W, Schieberle P (2009) Food chemistry, 4th revised and extended edn. Springer-Verlag, Berlin

- Reilly C (2002) Minerals. In: Henry CJK, Chapman C (eds) The nutrition handbook for food processors. CRC Press, Cambridge, pp 97–116
- Tomovic V, Jokanovic M, Sojic B et al (2015) Minerals in pork meat and edible offal. Procedia Food Sci 5:293–295
- 9. Cabrera MC, Ramos A, Saadoun A et al (2010) Selenium, copper, zinc, iron and manganese content of seven meat cuts from Hereford and Braford steers fed pasture in Uruguay. Meat Sci 84:518–528
- Hermida M, González M, Miranda M et al (2006) Mineral analysis in rabbit meat from Galicia (NW Spain). Meat Sci 73:635–639
- 11. Greenfield H, Southgate DAT (2003) Food composition data: production, management and use, 2nd edn. Food Agriculture Organisation of the United Nations, Rome

- Tomovic VM, Vujadinovic DD, Grujic RP et al (2014) Effect of endpoint internal temperature on mineral contents of boiled pork loin. J Food Process Preserv 39:1854–1858
- Valenzuela C, López de Romaña D, Olivares M et al (2009) Total iron and heme iron content and their distribution in beef meat and viscera. Biol Trace Elem Res 132:103–111
- Valenzuela C, Letelier MA, Olivares M et al (2008) Iron, zinc and copper content of cow meat. Rev Chil Nutr 35:139–146
- 15. Ripollés S, Campagnol PCB, Armenteros M et al (2011) Influence of partial replacement of NaCl with KCl, CaCl2 and MgCl2 on lipolysis and lipid oxidation in dry-cured ham. Meat Sci 89:58–64
- 16. Wu H, Yan W, Zhuang H et al (2016) Oxidative stability and antioxidant enzyme activities of drycured bacons as affected by the partial substitution of NaCl with KCl. Food Chem 201:237–242
- 17. Cittadini A, Domínguez R, Gómez B et al (2020) Effect of NaCl replacement by other

chloride salts on physicochemical parameters, proteolysis and lipolysis of dry-cured foal "cecina". J Food Sci Technol 57:1628–1635

- FAO (1997) Production and data management of chemical composition of foods in nutrition. Chapter 20: analysis of minerals and trace elements in foods. FAO
- 19. FAO (1985) CODEX STAN 150-1985: standard for food grade salt. According to EuSalt/ AS 015-2007 determination of elementsemission spectrometric method (ICP-OES). FAO
- 20. ISO 936 (1998) Meat and meat products determination of total ash, 2nd edn. International Organization for Standarization, Geneva
- 21. Lorenzo JM, Bermúdez R, Domínguez R et al (2015) Physicochemical and microbial changes during the manufacturing process of dry-cured lacón salted with potassium, calcium and magnesium chloride as a partial replacement for sodium chloride. Food Control 50:763–769



# **Chapter 8**

# **Nitrate and Nitrite**

# Elisa Rafaela Bonadio Bellucci, Camila Vespúcio Bis Souza, José Manuel Lorenzo, Gonzalo Aleu, Alfredo Teixeira, Rubén Domínguez, and Andrea Carla da Silva-Barretto

#### Abstract

The determination of the residual content of sodium nitrite and sodium nitrate in meat products is related to food safety, since the excessive consumption of these additives has been associated with the increase in the risk of certain types of cancer. However, the use of these additives in the meat industry is vital, since they play an important role in the controlling of pathogenic microorganisms, reduce oxidative reactions and improve color characteristics of meat products. Moreover, due to the health implications, specific limits (residual amounts) of nitrite and nitrate were stablished for each type of meat products. Thus, in view of the importance of this determination, the use of an appropriate analytic technique in meat and meat product is very important.

Consequently, this chapter gives a comprehensive vision of the procedure for the determination of residual nitrate and nitrite in cured meat products. All stages are described in detail so that the conditions proposed by us can be reproduced by other researchers.

Key words Residual nitrite, Residual nitrate, Colorimetric analyses, Cured meat products

#### 1 Introduction

Sodium nitrite and nitrate are ordinarily added in cured meat products because of their antimicrobial activity [1] due to inhibition of the germination of the spore of *Clostridium botulinum* and resisting the growth of *Listeria monocytogenes* [2], antioxidant activity over reduction in the rancidity of the meat product [3], and the development of the reddish-pink color and the flavor characteristics of cured meat products [4]. Despite their technological benefits, sodium nitrite and nitrate present a high risk to the health of the consumers, especially when consumed in excess. Some authors have already connected these components with the increase in the risk of certain types of cancer [5–7] since sodium nitrite may be the precursor of carcinogenic components such as nitrosamines

© The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

José Manuel Lorenzo et al. (eds.), *Methods to Assess the Quality of Meat Products*, Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_8,

[8]. Its application has been concerning the meat industry and consumers for years [9]. With all this in mind, there is a threshold limit set by the governments and health institutions for nitrite intake. Therefore, the level of nitrite present in the meat products needs to be tested to ensure food safety and protect the consumers from potential health problems [10].

The intake of nitrates and nitrites is predominantly due to the consumption of vegetables and meat products. In vegetables, the amount of nitrate present depends on the type of vegetable, whether it is the root, leaves or fruits, agricultural practices, genetic factors and, mainly, the way it is consumed (raw, cooked, or fried). In a study with the Italian population, it was concluded that vegetables contribute more to the intake of nitrate than meat products through the diet [7]. However, the use of both, nitrate and nitrite are also widely used in the meat industry. Nitrate (NO<sub>3</sub><sup>-</sup>) is used in meat products that will be stored for a long period of time as a source of nitrite for curing reactions. Although nitrate has the same healing function as nitrite, it reacts more slowly and is, therefore, less used. In addition, the conversion of nitrate to nitrite requires an additional step that depends on the action of microorganisms such as starter bacteria with nitrate-reducing properties which use nitrate as a substrate for anaerobic respiration [11].

On the other hand, the formation of the characteristic color of cured meat products is obtained through specific reactions that occur in the curing process. In this regard, the nitrosylation reaction (Fig. 1) occurs when the nitric oxide formed from the nitrite reacts with the myoglobin, forming an unstable compound called nitrosomyoglobin (MbNO), bright red in color, which becomes a stable and colored compound pink (nitrous-hemochrome) after



Fig. 1 Sequence of the nitrosylation reaction in cured products. Adapted from Ramos and Gomide [12]

heating [13]. However, it is difficult to control the amount of nitrite during the meat product processing due to the nature of the conditions of the manufacturing such as the heating that might lead to the conversion of nitrate into nitrite or nitrosamines. Thus, the control process is most appropriate for measuring the residual values in meat products [14].

With all the aforementioned, the present book chapter aims to fully and comprehensively describe a procedure for determining residual nitrite and nitrate in meat and meat products.

#### 2 Materials

	Prepare all solutions using distilled water and/or analytical grade reagents. Prepare and store all reagents at room temperature (unless indicated otherwise).
2.1 Residual Nitrite Determination	1. Sodium tetraborate solution decahydrate 5%: Dissolve 5 g of sodium tetraborate in 100 mL of distilled water.
	2. Potassium ferrocyanide solution 15%: Dissolve 150 g of Potas- sium ferrocyanide in 1 L of distilled water.
	3. Zinc acetate solution heptahydrate 30%: Dissolve 300 g of Zinc acetate heptahydrate in 1 L of distilled water.
	4. Sulfanilamide $(0.5\%)$ : Dissolve 1.25 g of Sulfanilamide $(C_6H_8N_2O_2S)$ in 250 mL of hydrochloric acid solution ( <i>see</i> <b>Notes 1</b> and <b>2</b> ).
	5. N-(1-naphtil)-ethylenediamine dihydrochloride (NED): Dissolve 0.5 g of alpha-naphthyl-ethylenediamine chloride in 100 mL of distilled water ( <i>see</i> <b>Note 3</b> ).
	<ol> <li>Standard solution of sodium nitrite (NaNO<sub>2</sub>) (0.2 g/L): Weigh analytically 0.2 g of sodium nitrite, previously dried for 1 hour at 105 °C. Dissolve in 1 L of distilled water (<i>see</i> Note 4).</li> </ol>
	7. Standard work solution of sodium nitrite $(8 \ \mu g/mL)$ : Pipette 10 mL of the standard solution of sodium nitrite $(0.2 \ g/L)$ in 250 mL of distilled water. Vortex the mixture.
2.2 Residual Nitrate Determination	1. Sodium tetraborate solution decahydrate 5%: Dissolve 5 g of sodium tetraborate in 100 mL of distilled water.
	2. Potassium ferrocyanide solution 15%: Dissolve 150 g of Potas- sium ferrocyanide in 1 L of distilled water.
	3. Zinc acetate solution heptahydrate 30%: Dissolve 300 g of Zinc acetate heptahydrate in 1 L of distilled water.
	4. Sulfanilamide (0.5%): Dissolve 1.25 g of Sulfanilamide $(C_6H_8N_2O_2S)$ in 250 mL of hydrochloric acid solution ( <i>see</i> <b>Notes 1</b> and <b>2</b> ).

- 5. N-(1-naphtil)-ethylenediamine dihydrochloride (NED): Dissolve 0.5 g of alpha-naphthyl-ethylenediamine chloride in 100 mL of distilled water (*see* **Note 3**).
- 6. Disodium ethylenediaminetetraacetate dehydrate (EDTA) 5% w/v: Dissolve 5 g of EDTA in water and complete the volume to 100 mL (*see* **Note 5**).
- Buffer solution (pH: 9.6–9.7)—Dilute 20 mL of hydrochloric acid in 700 mL of water. Add 50 mL of ammonium hydroxide. Adjust the pH to 9.6–9.7 if necessary using HCl or ammonium hydroxide and complete the volume to 1 L.
- 8. Standard solution of sodium nitrate (1 g/L): Weigh analytically 0.1 g of sodium nitrate, previously dried for 1 h at 105 °C. Dissolve in distilled water and add 50 mL of buffer solution (pH: 9.6–9.7). Complete the volumetric flash to 100 mL with distilled water.
- 9. Standard work solution of sodium nitrate  $(10 \ \mu g/mL)$ : Dilute 1 mL of standard solution of sodium nitrate in 100 mL volumetric flash. This work solution must be prepared in the analysis time.

#### 3 Methods

3.1 Residual Nitrite Determination: Samples Reading	The nitrite determination through the colorimetric methodology involves the diazotization of nitrite with sulfanilic acid and coupling with N-(1-naphtil)-ethylenediamine dihydrochloride forming the pink colored alpha-naphthylamino-p-azobenzene-p-sulfonic [15]. Figure 2 shows in a simplified and schematic steps of the nitrite extraction.
	1. Weigh 10 g of a mixed and homogenized sample in a 200 mL beaker and add 5 mL of sodium tetraborate solution decahy- drate 5%. Mix with a glass stick.
	2. Add 50 mL of hot distilled water (80 $^{\circ}$ C).
	<ol> <li>Place the samples in a hot water bath (60–70 °C) for 15 min. The samples must be mixed constantly (see Note 6).</li> </ol>
	4. Transfer the contents to a 200 mL volumetric flask with the aid of a funnel and a glass stick.
	5. Wash the beaker with 50 mL of hot distilled water (80 °C) and add this content in the volumetric flask.
	6. Let the samples cool to room temperature.
	7. Add 5 mL of potassium ferrocyanide solution 15% and 5 mL of zinc acetate solution heptahydrate 30% ( <i>see</i> <b>Note</b> 7).
	8. Complete the volume of the volumetric flask with distilled water.

89



Fig. 2 Schematic steps of the nitrite extraction

- 9. Shake vigorously and let the samples rest for 15 min.
- 10. Filter on qualitative filter paper in a 250 mL Erlenmeyer flask (*see* **Note 8**).



Fig. 3 Schematic steps for a reaction of sample and aliquots for standard curve with reagents to form a colored azo dye for residual nitrite determination

- 11. Add 10 mL of the reagent blank and each sample in a 50 mL volumetric flask. Figure 3 shows in a schematic way the steps for a reaction of sample with reagents to form a colored azo dye.
- 12. Add 5 mL of sulfanilamide solution. Shake the volumetric flask and let it react for 5 min.
- 13. Add 3 mL of NED. Shake each volumetric flask. Complete the volume (50 mL) with distilled water.
- 14. Leave the samples to stand for 15 min.
- 15. Read on a spectrophotometer at 540 nm of absorbance (*see* Note 9).

3.1.1 Residual Nitrite Determination: Analytical Curve

- 1. Add different aliquots (1, 2, 3, 4, 5, 6, 7 mL) of the standard work solution of sodium nitrite  $(8 \ \mu\text{g/mL})$  in a 50 mL volumetric flask. Figure 4 shows an example of analytical curve for nitrite determination.
- 2. Add 5 mL of sulfanilamide solution. Shake the volumetric flask and let it react for 5 min.
- 3. Add 3 mL of NED. Shake each volumetric flask. Complete the volume (50 mL) with distilled water.
- 4. Leave the samples to stand for 15 min.
- 5. Read on a spectrophotometer at 540 nm of absorbance (*see* Note 9).
- 6. Construct the curve with the absorbance values on the *y*-axis and sodium nitrite concentration on the *x*-axis (*see* **Note 10**) (Fig. 4).
- Calculate the linear and angular coefficient of the line (Fig. 4) (see Note 11).
- 8. Use Eq. 1 to quantify the sample's residual nitrite value.

Residual sodium nitrite 
$$\left(\frac{\text{mg}}{\text{kg}}\right) = \frac{(\text{ABS} - b) \times 1000}{p \times a}$$
 (1)

Being that:ABS: Sample absorbance, *b*: linear coefficient of the line obtained in the standard curve, 1000: dilution factor, *p*: sample mass in grams, *a*: absorptivity (Angular coefficient of the line obtained in the standard curve).

**3.2** *Nitrate* For nitrate analysis, nitrate is reduced to nitrite through a cadmium column in an alkaline medium and nitrite amounts is determinate as described in Subheadings 3.1 and 3.2. Another sample (aliquot) is used to determine residual nitrite without the reducing step. The difference between the amount of residual nitrite of sample reduced of nitrate (through the passage in cadmium column) and the residual nitrite is the commonly method to quantify nitrate in cured meat products samples [14].

- 1. Weigh 10 g of a mixed and homogenized sample in a 200 mL beaker and add 5 mL of sodium tetraborate solution decahydrate 5%. Mix with a glass stick.
- 2. Add 50 mL of hot distilled water (80 °C).
- 3. Place the samples in a hot water bath (60–70 °C) for 15 min. The samples must be mixed constantly (*see* Note 6).
- 4. Transfer the contents to a 200 mL volumetric flask with the aid of a funnel and a glass stick.
- 5. Wash the beaker with 50 mL of hot distilled water (80  $^{\circ}$ C) and add this content in the volumetric flask.



Fig. 4 Example of a calibration curve (absorbance "y-axis" vs. sodium nitrite concentration "x-axis")

- 6. Let the samples cool to room temperature.
- 7. Add 5 mL of potassium ferrocyanide solution 15% and 5 mL of zinc acetate solution heptahydrate 30% (*see* **Note** 7).
- 8. Complete the volume of the volumetric flask with distilled water.
- 9. Shake vigorously and let the samples rest for 15 min.
- 10. Filter on qualitative filter paper in a 250 mL Erlenmeyer flask (*see* **Note 8**).
- 11. Transfer 20 mL of the filtrate to a 100 mL beaker (first blank, then sample aliquots).
- 12. Add 5 mL of the buffer solution (see Note 12).
- 13. Pass through the cadmium column and collect in a 100 mL volumetric flask (*see* Notes 13 and 14).
- 14. Pass water through the column until completing 100 mL in the volumetric flask.
- 15. Add 10 mL of the reagent blank and each sample in a 50 mL volumetric flask. Figure 4 shows in a schematic way the steps for a reaction of sample with reagents to form a colored azo dye.
- 16. Add 5 mL of sulfanilamide solution. Shake the volumetric flask and let it react for 5 min.
- 17. Add 3 mL of NED. Shake each volumetric flask. Complete the volume (50 mL) with distilled water.

- 18. Leave the samples to stand for 15 min.
- 19. Read on a spectrophotometer at 540 nm of absorbance (*see* Note 9).

3.2.1 Nitrate Determination: Analytical Curve

Cadmium Column Efficiency

- At first, the efficiency of cadmium column must be tested by passing the standard solution of sodium nitrate in the cadmium column to determinate the amount of nitrite formed.
  - 1. In a 100 mL beaker, add 20 mL of standard work solution of sodium nitrate (10  $\mu$ g/mL) and 5 mL of buffer solution (pH: 9.6–9.7).
  - 2. Pass through the cadmium column (*see* Notes 13 and 14) and collect in a volumetric flash of 100 mL.
  - 3. Pass water through the column until completing 100 mL in the volumetric flask.
  - 4. Add 10 mL of the reagent blank and each sample in a 50 mL volumetric flask.
  - 5. Add 5 mL of sulfanilamide solution. Shake the volumetric flask and let it react for 5 min.
  - 6. Add 3 mL of NED. Shake each volumetric flask. Complete the volume (50 mL) with distilled water.
  - 7. Leave the samples to stand for 15 min.
  - 8. Read on a spectrophotometer at 540 nm of absorbance. Reset the spectrophotometer with the blank of reagents (passed through the column) (*see* Note 15).
  - 9. Use Eq. 2 to calculate the efficiency (%):

Efficiency (%) = 
$$\frac{(ABS - b) \times 30.8}{p \times a}$$
 (2)

Being that: ABS: Standard sodium nitrate absorbance, b: linear coefficient of the sodium nitrite standard curve, p: sample mass in grams, a: absorptivity (Angular coefficient of the sodium nitrite standard curve), 30.8: factor (dilution and conversion of sodium nitrate/sodium nitrite).

3.2.2 Calculation of Amount of Sodium Nitrate Construct the analytical curve using sodium nitrite as standard construct the graphic with the absorbance values on the *y*-axis and sodium nitrite concentration on the *x*-axis and calculate the linear and angular coefficient of the line for residual nitrite as described in Subheading 3.1.1.

Use Eqs. 1, 3, and 4 to quantify the amount of nitrate in the aliquots (*see* Note 10).

Total sodium nitrite 
$$\left(\frac{\text{mg}}{\text{kg}}\right) = \frac{(\text{ABS} - b) \times 5000}{p \times a}$$
 (3)

Sodium nitrate 
$$\left(\frac{\text{mg}}{\text{kg}}\right) = (\text{Eq.3} - \text{Eq.1}) \times 1.231$$
 (4)

Being that: ABS: Sample absorbance, *b*: linear coefficient of the line obtained in the standard curve, *p*: sample mass in grams, *a*: absorptivity (Angular coefficient of the line obtained in the standard curve), 1000: dilution factor, 5000: dilution factor of sample to calculation of total sodium nitrite, 1.231: correction factor from nitrite to nitrate.

#### 4 Notes

- 1. Dilute 125 mL of hydrochloric acid PA in 125 mL of distilled water.
- 2. This solution can be made weekly and stored under refrigeration in amber glass.
- 3. Store in amber glass under refrigeration.
- 4. This standard stock solution is stable for 2 weeks in amber glass at 4  $^{\circ}\mathrm{C}.$
- 5. EDTA solution must be stored in polyethylene bottles.
- 6. Make a sample blank, using the same reagents and conditions but without adding the meat sample or meat product. This will be the reagent blank.
- 7. Shake by rotation after adding each reagent.
- 8. The mixture to be filtered can clog the filters and become slow, so the filter can be changed periodically until the end of the filtration or use a vacuum pump.
- 9. Use the reagent blank to reset the equipment.
- 10. Sodium nitrite concentration: 0.16; 0.32; 0.48; 0.64; 0.80; 0.92; and 1.12 μg/mL of sodium nitrite.
- 11. The quantification limit of residual sodium nitrite for this analysis is  $0.032 \ \mu g/mL$  in the analysis pliers, with deviations of 3.7%.
- 12. If after adding the buffer solution (pH: 9.6–9.7) the solution becomes cloudy, 2 mL of the 5% EDTA solution should be added.
- 13. The flow in the cadmium column should not exceed 6 mL/ min.
- 14. Wash the walls of the cadmium column at least 5 times with approximately 15 mL of distilled water.
- 15. If the recovery is less than 90%, regenerate the column.

#### Acknowledgments

This work was supported by Department of Food Technology and Engineering. Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568).

#### References

- Mohamed AA, Mubarak AT, Fawy KF et al (2008) Modification of AOAC Method 973.31 for determination of nitrite in cured meats. Food Addit Contam 91:820–827
- Feng X, Li C, Jia X et al (2016) Influence of sodium nitrite on protein oxidation and nitrosation of sausages subjected to processing and storage. Meat Sci 116:260–267
- Sucu C, Turp GY (2018) The investigation of the use of beetroot powder in Turkish fermented beef sausage (sucuk) as nitrite alternative. Meat Sci 140:158–166
- 4. Bellucci ERB, Barretto TL, Rodriguez JML et al (2021) Natural colorants improved the physicochemical and sensorial properties of frozen brazilian sausage (Linguiça) with reduced nitrite. Sci Agric 78:e20190211
- Jin SK, Kim YJ, Park JH et al (2012) Effects of purple-fleshed sweet potato (*Ipomoera batatas* Cultivar Ayamurasaki) powder addition on color and texture properties and sensory characteristics of cooked pork sausages during storage. Asian-Australas J Anim Sci 25:1329–1337
- 6. Zhang H, Sun C, Han W et al (2018) Analysis of the monitoring status of residual nitrite in meat products in China from 2000 to 2011. Meat Sci 136:30–34
- Flores M, Toldrá F (2021) Chemistry, safety, and regulatory considerations in the use of nitrite and nitrate from natural origin in meat prdocuts. Meat Sci 171:108272

- Silva DRG, Haddad GBS, Moura AP et al (2021) Safe cured meat using gamma radiation: effects on spores of Clostridium sporogenesand technological and sensorial characteristics of low nitrite cooked ham. LWT- Food Sci Technol 137:110392
- Pinelli JJ, Martins HHA, Guimarães AS et al (2021) Essential oil nanoemulsions for the control of Clostridium sporogenes in cooked meat produtc: an alternative? LWT- Food Sci Technol 143:111123
- Trofimchuk E, Hu Y, Nilghaz A et al (2020) Development of paper-based nicrofluidic divice for the determination of nitrite in meat. Food Chem 316:126396
- 11. Majou D, Christieans S (2018) Mechanisms of the bactericidal effects of nitrate and nitrite in cured meats. Meat Sci 145:273–284
- Ramos EM, Gomide LAM (2017) Avaliação da Qualidade de Carnes: Fundamentos e Metodologias, 2a ed. Editora UFV, Viçosa
- Parthasarathy DK, Bryan NS (2012) Sodium nitrite: the "cure" for nitric oxide insufficiency. Meat Sci 92:274–279
- 14. Azeem SMA, Madbouly MD, El-Shahat MF (2019) Determination of nitrite in processed meat using digital image method and powdered reagent. J Food Compos Anal 82:28–36
- 15. Pereira EA, Petruci JFS, Cardoso AA (2012) Determination of nitrite and nitrate in Brazilian meats using high shear homogenization. Food Anal Methods 5:637–642



# **Chapter 9**

### **Biogenic Amines**

### Claudia Ruiz-Capillas, Mehdi Triki, Rubén Domínguez, José Manuel Lorenzo, and Ana M. Herrero

#### Abstract

The control and determination of biogenic amines in different kinds of foods in particular in meat products is important to assure their safety and high level of quality. Having precise analysis techniques, with few limitations and easy to perform and reproduce is a challenge for analysts, researchers, control laboratories, and companies.

This chapter includes a detailed protocol for a chromatographic simultaneous determination of the most common biogenic amines (BA) (tyramine,  $\beta$ -phenylethylamine, histamine, putrescine, cadaverine, tryptamine, agmatine, spermidine, and spermine) that can be found in meat and meat products. This protocol description includes both the acid samples extraction step of the BA and their determination by liquid chromatography which employs a cation-exchange column for BA separation and with a post-column system for the derivatization process using o-phthalaldehyde as a derivatizing reagent.

This method showed an appropriate, precise, fast, and versatile procedure to determine nine BA simultaneously in different matrices of meat products.

Key words Biogenic amines, HPLC post-column, Ion chromatography, OPA, Meat and meat products

#### 1 Introduction

Biogenic amines (BA) are biologically active low-molecular-weight basic nitrogenous compounds which are present in great majority of foods (fish, meat, cheese, vegetables, soy bean, etc.) [1–3]. Their control in foods is important for safety and quality reasons. The consumption of food containing high concentrations of biogenic amines has been associated with toxic effects and constitutes a potential risk for the human health ("scombroid poisoning" or "histamine poisoning," "tyramine poisoning" or "cheese poisoning," etc.) [3–6]. Besides, BA have been employed as quality indices in food, mainly in meat and fish. They allowed to assess the success of the applied technological process and shelf life of the products during conservation [3, 7–10].

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_9,

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

Biogenic amines are produced by decarboxylation of free amino acids from the action of microbial amino acid decarboxylase enzymes [3]. Its greater or lesser presence in food depends on various factors associated with food composition, processing, and storage conditions that can affect the formation of these BA (free amino acids, microorganisms, medium, processing and preservation conditions, etc.) [3, 8, 11].

Therefore, controlling, detecting, and quantifying these biogenic amines in the different kinds of foods is important to assure a high level of quality and food safety [3, 7, 8]. In this regard, it is worth mentioning that there are specific legislations designed in order to control these compounds [12-14].

Numerous procedures have been developed for the determination of BA in various types of foods. Methods range from the determination of individual amines to the joint simultaneous determination of various amines present in food [1, 3, 7, 15, 16]. Many methods can be found in the literature, from traditional ones, such as colorimetric and fluorometric methods focused mainly on determining histamine individually, to modern ones, such as gas chromatography (GC), gas chromatography-mass spectrometry (GC-MS), high-performance liquid chromatography (HPLC), HPLC-tandem mass spectrometry, flow injection analysis (FIA), capillary electrophoresis, etc., that allow the simultaneous determination of several biogenic amines in the same food [3, 16]. Of all the techniques/methods used for the determination of BA, nowadays HPLC is the most common [3]. The chromatographic determination by HPLC generally uses pre- and post-columns with reverse phase or ion exchange columns and with different derivatizing reagents (ninhydrin, dansyl chloride, ortho-phthalaldehyde (OPA), etc.) to increase sensitivity of the HPLC method, since BA are characterized by their low volatility and lack of chromophores. Depending on the derivatizing reagent, different detection systems are used: UV/Vis, diode array, this latter being the most used fluorescence detector [1, 3, 17, 18].

The determination of biogenic amines also requires a more or less complex extraction process which depends on the type of matrix, the technique, and the separation and identification of the BA. Indeed, many different solvents, such as hydrochloric acid, trichloroacetic acid, perchloric acid, methanol, and other organic solvents, have been used to extract BA from meat products.

Therefore, the methodology proposed in the present chapter for the determination of BA includes the extraction process, the chromatographic separation, the identification, and the quantification of nine BA simultaneously following the methodology performed by Triki et al. [16].

2 Materials	
	All solutions must be prepared with ultra-pure water and analytical grade reagents. Preparation of all reagents is performed at room temperature and storage is carried out at 4 °C.
2.1 Biogenic Amines Extraction	The determination of biogenic amines in solid foods frequently requires extraction or concentration procedures prior to separation and analysis. This is a critical stage for adequate recovery of all BA due to the complexity of the sample's matrices and the fact that they do not contain only one BA [3, 15].
2.2 Biogenic Amines Standard Solutions for Identification and Quantification	<ol> <li>The trichloroacetic acid (TCA) was supplied by Panreac (Barcelona, Spain) (see Note 1). TCA 7.5%: Dissolve 75 g of TCA in 1 l of ultra-pure water (see Note 2). The biogenic amines included in these standard solutions are: tyramine hydrochloride (Tyr), histamine dihydrochloride (His), 2-phenylethylamine hydrochloride (Pea), putrescine dihydrochloride (Put), cadaverine dihydrochloride (Cad), tryptamine-crystalline (Try), agmatine sulfate salt (Agm), spermidine trihydrochloride (Spd), and spermine tetrahydrochloride (Spm). They were purchased from Sigma-Aldrich (Spain).</li> <li>A BA stock solution of 1000 mg/l was prepared, as a free base with TCA 7.5% in a volumetric flask:         <ul> <li>(a) A stock solution of 1000 mg/l of each separate BA was prepared in a volumetric flask (see Note 3).</li> <li>(b) Afterwards, a 1000 mg/l stock solution of all biogenic amines was prepared from the previous separate stock solutions in a volumetric flask and transferred to 5 ml glass tubes. Then, from the glass tubes, the solution was transferred again to vials (see Note 3).</li> </ul> </li> <li>BA intermediate solution of 100 mg/l was prepared with TCA 7.5% from the stock solution (1000 mg/l):         <ul> <li>(a) 2.5 ml of the BA stock solution were poured in a 25 ml volumetric flask and the rest was filled with 7.5% TCA until reaching the mark of the flask (see Note 3).</li> </ul> </li> <li>BA working solutions (Standard solutions):         <ul> <li>(a) From the BA intermediate solution, appropriate working solutions from 0.05–12 mg/l of mixed BAs with TCA 7.5% were prepared to be used for the standards calibration. For example, the 4 mg/l working solution was prepared in a 10 ml volumetric flask where 0.4 ml of the</li> </ul> </li> </ol>
intermediate solution were poured and the rest was filled with TCA 7.5% (*see* **Note 3**).

- (b) The working solutions were also filtered through a  $0.22 \ \mu m$  Nylon Syringe filter (Teknokroma, Barcelona, Spain) into 2 ml amber vials with screw caps (PTFE/silicone) (Perkin Elmer Life and Analytical Sciences, USA) and kept in refrigeration until their use. They will be placed in the auto-sampler which will take a certain volume that will be injected into a high-performance liquid chromatograph (HPLC).
- 2.3 Mobile Phases1. The trichloroacetic acid (TCA) was supplied by Panreac (Barcelona, Spain) and the ultra-pure water was obtained from Milli-Q system (Millipore, France).
  - Methanol and 2-propanol for high-performance liquid chromatography, potassium phosphate dibasic, potassium hydroxide, potassium chloride, acetic acid, and Brij 35 solution (30% w/v) were obtained from Sigma-Aldrich (Spain).
  - O-phthalaldehyde (OPA, ref. O120), Thiofluor Chromatographic Grade (N,N-Dimethyl-2mercaptoethylamine-hydrochloride), OPA diluent (ref OD104: 3% potassium hydroxide, 3% boric acid, 94% water, pH = 10.40), and for the mobile phases, potassium phosphate phase A buffer (ref K600: 11% 2-propanol, 0.9% potassium phosphate dibasic, 0.3% acetic acid, 87.8% water, pH = 6.00), phase B (ref K563: 5% potassium chloride, 4% 2-propanol, 0.9% potassium phosphate dibasic, 0.3% acetic acid, 89.8% water, pH = 5.63), and the potassium regenerating column, phase C (ref: K130: 0.7% potassium chloride, 4% 2-propanol, 0.5% potassium hydroxide, 94.8% water, pH = 13.00) were all purchased from Pickering laboratories (CA, USA) (*see* Note 4).
  - 1. The post-column derivatizing reagent OPA was prepared with 975 ml of OPA solution (ref OD 104) and 0.100 mg of OPA (OPA, ref. O120) dissolved in 10 ml methanol, 2 g Thiofluor, and 3 ml of Brij 35 solution, which were purchased from Sigma-Aldrich (Spain) (*see* Note 5).
    - (a) A part from OPA solution (ref OD 104) (approx. 800 ml) was poured in an opaque bottle (OPA bottle) which was degassed with Helium for 5 min (*see* **Note 6**).
    - (b) 0.100 mg of OPA (OPA, ref. O120) were weighted in a beaker covered with aluminum paper (to avoid contact with light).
    - (c) 10 ml of methanol were added to the beaker and shaken until the OPA (OPA, ref. O120) dissolved completely.

2.4 Derivatizing Reagent OPA Preparation

	(d)	The obtained solution was then added to the degassed OPA bottle.
	(e)	The beaker was cleaned with the rest of the OPA solution (ref OD $104$ ) (that was not poured in the OPA bottle).
	(f)	2 g of Thiofluor were weighted and dissolved in the rest of the OPA solution (ref OD 104) (which was used to clean the beaker).
	(g)	The obtained solution was added to the degassed OPA bottle ( <i>see</i> <b>Note</b> 7).
	(h)	3 ml of Brij 35 solution were added to the degassed OPA bottle ( <i>see</i> <b>Note</b> 7).
	(i)	The whole content in the degassed OPA bottle was thor- oughly mixed ( <i>see</i> <b>Note 8</b> ).
	(j)	The whole solution was then degassed again (see Note $6$ ).
2.5 High- Performance Liquid Chromatography Equipment (HPLC)	The chromatographic determination of BAs was performed us liquid chromatography consisting of: a quaternary pump (see 200, Perkin Elmer, SL Spain), an auto-sampler (series 200, Per Elmer Life and Analytical Sciences, USA), a Pickering PCX 33 post-column system (Pickering Laboratories, CA, USA) contain a cation-exchange column (K <sup>+</sup> , 4 mm × 150 mm) with a 10 particle diameter, and a pre-column (K <sup>+</sup> , 3 mm × 20 mm) also v a 10 $\mu$ m diameter particle (Pickering Laboratories, CA, USA). Whole system was degassed and pressurized with helium ( <i>see</i> N 9).	

## 3 Methods

3.1 Biogenic Amines Extraction	The extraction of BA from meat and meat products samples was performed following the steps mentioned hereafter:
	1. 15 g of the sample were weighted in an omnimixer glass (Ovni International, Waterbury, CT, USA) and 30 ml of TCA (7.5%) were then added.
	2. The mixture was blended in the omnimixer at 20,000 rpm during 3 min.
	3. Afterwards, the obtained blend was centrifuged at $5000 \times g$ for 15 min at 4 °C in a desktop centrifuge (Sorvall RTB6000B, DuPont, USA).
	4. The obtained supernatant was then filtered through a Whatman $n^{\circ}$ 1 filter.
	5. The filtrate was poured in a 50 ml volumetric flask which was filled with TCA (7.5%).

3.2 Biogenic Amines

Identification and Quantification (HPLC) 6. The filtrate was also passed through a  $0.22 \mu$  Nylon filter (Millipore, Ireland) and then placed in opaque vials (2 ml) in the auto-sampler which were stored at 2 °C, until their use (within the next 24 h).

BA Separation and quantification are carried out using a HPLC, according to the following conditions:

- BA were separated using an elution gradient programmed with potassium reagents K600, K563, and K130 following the method used by Ruiz-Capillas and Moral [7] and Triki et al. [16]. With the exception of step 2, where an isocratic gradient curve was applied (curve N° 1), the rest of the program was based on a linear gradient (without curve) (Table 1) (*see* Note 10).
- 2. The column and pre-column temperatures were programmed at 40 °C. In the reaction chamber, the post-column reagent (OPA) flow rate was 0.3 ml/min.
- 3. The applied temperature of the reaction chamber was 45  $^{\circ}$ C.
- 4. Detection was performed using an LC 240 fluorescence detector (Perkin Elmer Life and Analytical Sciences, USA) at 330 nm excitation and 465 nm emission lengths.
- 5. All the chromatographic systems were controlled using a PE Nelson data integrator (Perkin Elmer Life and Analytical Sciences, USA). Data acquisition was carried out using Total-Chrom software (Perkin Elmer Life and Analytical Sciences, USA).
- 6. BA standards working solution were placed in the injector together with the samples for the BA identification and to create calibration curves (Figs. 1 and 2). BA identification and quantification was done by comparing and extrapolating retention times with a calibration curve performed with different standard solutions (Fig. 3). The results were expressed as mg/kg of sample (*see* Note 11).

#### 4 Notes

- 1. Since the trichloroacetic acid (TCA) is very hygroscopic, once it is opened, it must be very well covered and stored in a dry place.
- 2. TCA must be handled carefully; it is extremely corrosive and harmful to the environment. During the handling, the operator must wear goggles and protective equipment. The preparation of the TCA 7.5% must be performed in the laboratory hood.

#### Table 1

Elution gradient program for the chromatographic separation of biogenic amines in meat and meat products

			Mobile phases			
Step	Time (min)	Flow (ml/min)	K 600 (%)	K 563 (%)	K 130 (%)	Gradient curve
0	0	0.8	100	0	0	0
1	6	0.8	100	0	0	0
2	9	0.8	0	100	0	1
3	6	0.8	0	100	0	0
4	3	0.8	0	0	100	0
5	7	0.8	100	0	0	0



Fig. 1 Chromatograms of biogenic amines (tyramine,  $\beta$ -phenylethylamine, histamine, putrescine, cadaverine, tryptamine, agmatine, spermidine, and spermine) of different standard solutions (0.05–12 mg/l) for the preparation of the calibration line

- 3. Solutions must be well covered and stored in refrigeration until their later use.
- 4. Since in chromatographic separation with ion exchange columns, the pH of the mobile phases is a key factor in resolving the various peaks, the effects produced by pH variations related to the mobile phases were taken into account.





#### b) Frankfurter



**Fig. 2** Chromatograms of biogenic amines (tyramine,  $\beta$ -phenylethylamine, histamine, putrescine, cadaverine, tryptamine, agmatine, spermidine, and spermine) in "chorizo" (**a**) and Frankfurter (**b**)

- 5. This solution is prepared again weekly at the start of the working week if, in the meantime, it was not already finished.
- 6. The bottle must be opaque or covered with aluminum foil, avoid contact with light.



**Fig. 3** Chromatograms of biogenic amines (tyramine,  $\beta$ -phenylethylamine, histamine, putrescine, cadaverine, tryptamine, agmatine, spermidine, and spermine) in meat products "Chorizo" (green), "Frankfurter" (orange), and standard solution 4 mg/l (black)

- 7. Prior to the addition of solutions, the gas of the system should be off.
- 8. Attention! This solution makes a lot of foam. It must be left for a while to stabilize.
- 9. Before starting the determination and once the phases have been degassed, the pump and the post-column system must be purged efficiently. The equipment must be left to stabilize for at least 10 min before starting with the injections.
- 10. It is very important to regenerate the column (K130) in order to obtain a stable balance for subsequent determinations.
- 11. The standard working solution should be injected every time before a series of samples injections to be more accurate when quantifying the BA in the samples since the column is very sensitive to slight changes in the conditions of the experiment. For example, the standard working solution can be injected again following a 5-h pause of injections or after performing injections non-stop for 10 h.

#### Acknowledgments

This research was funded by the Spanish Ministry of Science and Innovation (PID2019-107542RB-C21), by the CSIC Intramural projects (grant number 201470E073 and 202070E242), CYTED (grant number reference 119RT0568; Healthy Meat network).

#### References

- 1. Ruiz-Capillas C, Jiménez-Colmenero F (2009) Biogenic amines in seafood products, chapter 42. In: Nollet LML, Toldrá F (eds) Handbook of seafood and seafood products analysis. CRC Press, Boca Raton, pp 833–846
- Linares DM, Martin MC, Ladero V et al (2011) Biogenic amines in dairy products. Crit Rev Food Sci Nutr 51:691–703
- 3. Ruiz-Capillas C, Herrero AM (2019) Impact of biogenic amines on food quality and safety. Foods 8:62
- 4. Ruiz-Capillas C, Jiménez-Colmenero F (2004) Biogenic amines in meat and meat products. Crit Rev Food Sci Nutr 44:489–599
- EFSA European Food Safety Authority (2011) Scientific opinion on risk-based control of biogenic amine formation in fermented foods. EFSA J 9:2393
- Latorre-Moratalla ML, Comas-Basté O, Bover-Cid S et al (2017) Tyramine and histamine risk assessment related to consumption of dry fermented sausages by the Spanish population. Food Chem Toxicol 99:78–85
- Ruiz-Capillas C, Moral A (2001) Production of biogenic amines and their potential use as quality control indices for hake (*Merluccius merluccius* L.) stored in ice. J Food Sci 66: 1030–1032
- 8. Triki M, Herrero AM, Jiménez-Colmenero F et al (2018) Quality assessment of fresh meat from several species based on free amino acid and biogenic amine contents during chilled storage. Foods 7:132
- Lázaro CA, Conte-Júnior CA, Canto AC et al (2015) Biogenic amines as bacterial quality indicators in different poultry meat species. LWT-Food Sci Tech 60:15–21
- Mah J-H, Park YK, Jin YH et al (2019) Bacterial production and control of biogenic amines in Asian fermented soybean foods. Foods 8:85

- Barbieri F, Montanari C, Gardini F et al (2019) Biogenic amine production by lactic acid bacteria: a review. Foods 8:17
- 12. FDA (Food and Drug Administration) (2021) Fish and fishery products hazards and controls guidance. 4th edn. FDA. https://www.fda. gov/food/seafood-guidance-documents-regu latory-information/fish-and-fishery-productshazards-and-controls. Accessed 6 July 2021
- 13. European Commission Regulations (2073/2005; 1441/2007; 1019/2013; 443/2020)
- 14. https://eur-lex.europa.eu/legal-content/ EN/TXT/PDF/?uri=CELEX:32005R20 73&from=EN; https://eur-lex.europa.eu/ legal-content/EN/TXT/PDF/? uri=CELEX:32007R1441&from=EN; https://eur-lex.europa.eu/legal-content/ EN/TXT/PDF/?uri=CELEX:32013R1019; https://eur-lex.europa.eu/legal-content/ EN/TXT/PDF/?uri=CELEX:32020 R0443&from=EN. Accessed 16 July 2021
- 15. Önal A (2007) A review: current analytical methods for the determination of biogenic amines in foods. Food Chem 103:1475–1486
- 16. Triki M, Jiménez-Colmenero F, Herrero AM et al (2012) Optimisation of a chromatographic procedure for determining biogenic amine concentrations in meat and meat products employing a cation-exchange column with a post-column system. Food Chem 130: 1066–1073
- Ruiz-Capillas C, Herrero AM, Jiménez-Colmenero F (2016) Determination of biogenic amines. In: Ruiz-Capillas C, Nollet LML (eds) Flow injection analysis of food additives. CRC Press, Boca Raton, FL, pp 675–690
- Liu SJ, Xu JJ, Ma CL et al (2018) A comparative analysis of derivatization strategies for the determination of biogenic amines in sausage and cheese by HPLC. Food Chem 266: 275–283



# **Chapter 10**

# **Spectrophotometric Analysis of Protein Carbonyls**

## Francesca Soglia, Giulia Baldi, Alberto González-Mohino, Silvia Díaz-Velasco, Massimiliano Petracci, and Mario Estévez

#### Abstract

Oxidation of proteins is a major threat to their functionality and to the sensory properties and nutritional value of meat and meat products. Moreover, the intake of protein oxidation products may also involve a potential health risk for consumers. The accurate analysis of food protein oxidation seems to be an issue of scientific and technological relevance. Protein carbonylation is one of the most remarkable modifications in oxidized proteins and protein carbonyls are commonly used as markers of the oxidative damage to meat proteins. Regardless of its documented drawbacks, the dinitrophenylhydrazine (DNPH) method is the most widely used among food scientists for the quantification of protein carbonyls. Therefore, this chapter aims to provide a detailed description of the abovementioned determination which involves a simultaneous quantification of total protein carbonyls and total protein content in a given sample.

Key words Protein oxidation, Carbonyls, Hydrazones, 2,4-Dinitrophenylhydrazine, DNPH, Spectrophotometric method

#### 1 Introduction

Proteins play a crucial role in foods from various perspectives including functionality (i.e., water absorption and retention, gelation, viscosity, emulsification, foam formation, etc.), sensory properties (i.e., texture and rheological properties, flavor-binding properties, enzymatic browning), and nutritional value (e.g., contribution with essential amino acids) [1]. Yet, a number of post-translational modifications in proteins affect their composition and native structure, leading to impairments in their functional properties and nutritional value.

Oxidation is a major threat to protein stability and functionality as the attack of reactive oxygen species (ROS) or reactive Maillarddicarbonyls such as glyoxal and methylglyoxal, among others, leads to severe chemical changes in proteins, namely, peptide fragmentation, amino acid side chains oxidation, and formation of protein cross-links [2]. These chemical modifications occur all the way from

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_10,

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

food collection/harvesting until culinary treatment prior to consumption [3]. Yet, protein oxidation is particularly promoted during severe and recurring processing such as the combination of some of the following treatments: mincing, slicing, application of high temperatures, radiation, high-pressure, microwaves, packaging in high-oxygen atmospheres, etc. [1]. The consequences of severe protein oxidation are commonly negative as it is linked to increased toughness in meats, loss of essential amino acids and reduced protein digestibility, among others [4]. Additionally, recent reports emphasized the potential toxicity of oxidized proteins and amino acids and therefore, the intake of protein oxidation products may also involve a potential health risk [5].

In view of the aforementioned statements, the analysis of food protein oxidation seems to be an issue of scientific and technological relevance. The quantification of protein oxidation may be essential to identify the extent to which food proteins have suffered oxidative damage; the effectiveness of a given antioxidant strategy may be assessed by confirming its ability to reduce the amount of protein oxidation products [6]. Since protein oxidation is a complex phenomenon and is manifested as manifold chemical changes, an assortment of analytical procedures has been proposed to evaluate its occurrence in food systems [2]. Among all of them, the quantification of protein carbonyls is, without any doubt, the most common method to assess protein oxidation in biological samples. The occurrence of carbonyls in proteins is an unquestionable indication of an oxidative damage as the oxidative deamination of certain alkaline amino acids, such as lysine, arginine, and proline, leads to the formation of protein-bound carbonyls [7]. While several analytical procedures have been proposed to detect and quantify protein carbonyls, the spectrophotometric method that involves prior derivatization of the protein sample with dinitrophenylhydrazine (DNPH), is, again, the most widely used among food scientists (Fig. 1). The key to this procedure is to facilitate the exposure of protein-bound carbonyls to the DNPH and remove all exceeding reagent so that the numbers from quantification are as accurate as possible. To fulfill this objective, some cleaning steps are required to remove lipids and any other substrate and/or fluorophore that may interfere with the spectrophotometric measurement at 370 nm. The method has been criticized for being time-consuming, poorly accurate and intricate in the way samples have to be handled during the entire method, which facilitates a failed sampling from homogenates, loss of sample during washing steps or a failed derivatization. Even if the method is applied correctly, the nature and origin of the protein carbonyls remains unknown as these can derive from the oxidation of amino acid side chains (primary carbonyls) or from the oxidation of unsaturated lipids (i.e., malondialdehyde, MDA) which appear as proteinbound carbonyls upon reaction and covalent linkage with protein



**Fig. 1** Schematic representation of the DNPH-derivatization of primary (allysine) and secondary (protein-bound MDA) protein carbonyls (Adapted from Estévez et al. [8])

amines [8]. Figure 1 shows derivatization of DNPH with one primary carbonyl (allysine) and one secondary carbonyl (proteinbound MDA). The method, that involves a simultaneous quantification of total protein carbonyls and total protein content in a given sample, is described in the following sections.

### 2 Materials

	Prepare all solutions using distilled water and/or analytical grade reagents. Prepare and store all the solutions and reagents at $4 \pm 1$ °C (unless indicated otherwise) with the only exception of Sodium Dodecyl Sulfate (SDS) 5% (w/v) which should be stored at room temperature to avoid crystallization. Diligently follow all waste disposal regulations when disposing waste materials.
2.1 Protein	<ol> <li>KCl 0.15 M: Dissolve 11.18 g of potassium chloride (≥98%) in</li></ol>
Extraction	1 L of distilled water ( <i>see</i> Note 1).
	<ol> <li>TCA 10% (w/v): Dissolve 100.0 g of trichloroacetic acid (≥98%) in 1 L of distilled water (see Note 2).</li> </ol>
2.2 Carbonyl Groups	<ol> <li>SDS 5% (w/v): Dissolve 50 g of sodium dodecyl sulfate</li></ol>
Exposure	(≥98%) in 1 L of distilled water (see Note 3).

2.3	Derivatization	1. HCl 3 M: Mix 248 mL of hydrochloric acid ( $\geq$ 37%) with 752 mL of distilled water ( <i>see</i> <b>Notes 2</b> and <b>4</b> ).
		2. DNPH 0.3% (w/v) in 3 M HCl: Dissolve 6.0 g of 2,4-Dinitrophenylhydrazine (~50%) in 1 L of HCl 3 M ( <i>see</i> <b>Notes 2</b> and <b>5</b> ).
2.4	Washing	<ol> <li>TCA 40% (w/v): Dissolve 400.0 g of trichloroacetic acid (≥98%) in 1 L of distilled water (see Note 2).</li> </ol>
		2. Ethanol-ethyl acetate (1:1): Mix 500 mL ethanol with 500 mL ethyl acetate ( <i>see</i> <b>Note 6</b> ).
		3. Guanidine hydrochloride 6 M in NaH <sub>2</sub> PO <sub>4</sub> 20 mM: Add 573.18 g of guanidine hydrochloride ( $\geq$ 99%) to 1 L of sodium phosphate buffer ( <i>see</i> the following <b>item 4</b> ) and adjust the final pH of the solution to 6.5 ( <i>see</i> <b>Note</b> 7).

4. NaH<sub>2</sub>PO<sub>4</sub> 20 mM: Add 2.40 g of Sodium phosphate monobasic ( $\geq$ 98%) to 1 L of distilled water.

## 3 Methods

	Carry out all procedures at room temperature, unless otherwise specified. Carbonyl content should be measured following the traditional spectrophotometric DNPH-based method described by Levine et al. [9] with the modifications proposed by Soglia et al. [10]. As aforementioned, the spectrophotometric quantifica- tion of protein carbonyls at 370 nm is made simultaneously to the spectrophotometric quantification of total protein content at 280 nm since results are commonly expressed as per protein unit. All the steps of protein extraction, carbonyl groups exposure and subsequent quantification are outlined in Fig. 2.
3.1 Protein Extraction	1. Weigh 1.0 g of sample in a 50 mL centrifuge tube and add 10 mL of ice-cold 0.15 M KCl solution ( <i>see</i> <b>Notes 1</b> and <b>8</b> ).
	2. Homogenize the samples by Ultra-Turrax (IKA-WERKE, Labortechnik, Staufen, Germany) or similar apparatus at 9500 rpm for 30 s ( <i>see</i> <b>Note 8</b> ).
	3. Per each sample, prepare five microcentrifuge tubes of 2 mL in which aliquots of homogenate (100 $\mu$ L/each) are mixed with 1 mL 10% TCA ( <i>see</i> <b>Notes 1</b> and <b>9</b> ).
	4. Centrifuge the samples at $5000 \times g$ for 5 min at room temper- ature ( <i>see</i> <b>Note 10</b> ).
3.2 Carbonyl Groups Exposure	<ol> <li>After removing and discarding the supernatant, add 400 μL of 5% SDS to the pellet.</li> </ol>



Fig. 2 Schematic and simplified representation of the steps of carbonyls quantification involving protein extraction, carbonyl groups exposure, derivatization, and quantification in the presence of DNPH

- Heat the samples in a dry bath at 100 °C for 10 min (see Note 11).
- 3. Sonicate the samples in an ultrasonic cleaner at 40 °C for 30 min.
- 4. The implementation of the abovementioned steps has been recently studied and introduced by Soglia et al. [10]. In detail, these steps, inducing protein unfolding, result in the exposure of even the carbonyl groups that being buried in the inner core of the proteins' structure would have not react with DNPH, thus leading to an underestimation of the carbonyls development following protein oxidation.
- 3.3 Derivatization
   1. Add 800 μL of 0.3% DNPH in 3 M HCl to each sample, while the same volume (800 μL) of 3 M HCl is added to the blanks (*see* Note 2).
  - 2. Incubate the samples for 30 min at room temperature.
  - Following incubation, add 400 μL 40% TCA to each tube (see Note 2).

# **3.4 Washing 1.** Centrifuge the samples at $5000 \times g$ for 5 min (at room temperature) to precipitate the proteins and separate (and discard) the supernatant (*see* Note 10).

2. Add 1 mL of ethanol:ethyl acetate solution (1:1) to each tube (*see* **Note 6**).

- 3. Centrifuge the samples at  $10,000 \times g$  for 5 min (at room temperature) and discard the supernatant (*see* Notes 6 and 10).
- 4. Repeat steps 2 and 3 two more times in order to remove unbound DNPH reagent.
- 5. After removing the supernatant resulting from the third washing step, evaporate any residual solvent under  $N_2$  flux or allow the pellet to dry under the fume hood (*see* Note 6).
- 6. Resuspend the pellet in 1.5 mL 6 M guanidine hydrochloride in NaH<sub>2</sub>PO<sub>4</sub> 20 mM (pH 6.5) (*see* **Note 2**).
- 7. Incubate the samples at  $4 \pm 1$  °C, overnight.
- **3.5 Quantification** 1. Transfer the samples into semi-micro, UV-grade cuvettes (1.5 mL).
  - 2. Read the absorbance of the samples and their respective blanks at 370 nm.
  - 3. Read the absorbance of the samples (only) at 280 nm.
  - 4. Calculate carbonyl content (expressed as nmol/mg of protein) according to the formula:

 $Carbonyl \text{ content} = \frac{Abs \ 370 - Abs \ 370_{blank}}{22,000 \times [Abs \ 280 - (Abs \ 370 - Abs \ 370_{blank}) \times 0.43]} \times 10^6$ 

#### 4 Notes

- 1. Depending on the nature of the sample (liquid, solid, semisolid) and the solubility of the sample's proteins in water, the use of specific buffers and sodium chloride (>2 M) to increase the ionic strength of the solution may be required. In very dry samples (i.e., dry-cured products) and very low protein solubility (i.e., collagen/elastin), the homogenization parameters may be properly modified (e.g., longer time and/or higher speed of homogenization), and the application of urea (>2 M) may be required. The protein extraction should be optimized to guarantee that each sampling aliquot from the homogenate (1) accurately represents the original sample and (2) contains the same protein content.
- 2. Due to the toxic and/or corrosive nature of the reagents used for protein precipitation, it is necessary for the operator to take the required protective measures (gloves, glasses, etc.) as well as to carry out all the operations in laboratory fume hoods.
- 3. SDS is an anionic surfactant capable of denaturing proteins' structure. Due to its detergent nature, air bubbles easily develop when the solution is transferred from the beaker, used to dissolve it, to a volumetric flask. To avoid/limit the

formation of air bubbles, gently heat up the solution to a temperature lower than 68  $^{\circ}$ C while stirring with a magnetic stirrer to assist dissolution.

- 4. Exothermic reaction. Insert 500 mL of distilled water into a 1 L beaker and slowly add 248 mL of hydrochloric acid ( $\geq$ 37%) while stirring with a magnetic stirrer. Wait for the mixture to cool down to room temperature, then add the remaining 252 mL of distilled water.
- 5. The eventual water added to moisten and stabilize the reagent should be considered in the calculation. As an example, if about 50% of water is used to moisten DNPH, then 6.0 g (instead of 3.0 g) of reagent should be dissolved in 1 L of HCl 3 M to achieve the final concentration of 0.3%.
- 6. Due to the irritant nature of the reagents used for washing the pellet, it is necessary for the operator to take all the required protective measures (gloves, glasses, etc.) as well as to carry out all the operations in laboratory fume hoods.
- 7. Endothermic reaction. The solution should be prepared with constant magnetic stirring and gentle heating (<40 °C) and allowed to equilibrate to room temperature before transferring it to a volumetric flask and fill it up to the mark.
- 8. Depending on the concentration of protein in the sample as well as on the extent of the oxidative damage, the amount of sample, the sample:dispersing solution ratio, and the homogenization intensity and time may vary.
- 9. Before pipetting the aliquots in the microcentrifuge tubes, the homogenate should be vortexed (few seconds) to have a homogeneous sample.
- 10. Depending on the protein nature and concentration, the centrifugation force (× g) as well as the TCA concentration added may be adjusted. The pellet formed upon protein precipitation in these steps should be (1) strong enough to allow the easy removal of supernatant but (2) mild enough to guarantee a complete dissolution in the next step for carbonyl exposure, derivatization, or washing.
- 11. Once that the microcentrifuge tubes are placed in the dry bath, gently individually raise and shake them to ensure that the pellet is located in the bottom of the tube and submerged with the SDS solution.

#### References

1. Soladoye OP, Juárez ML, Aalhus JL et al (2015) Protein oxidation in processed meat: mechanisms and potential implications on

human health. Compr Rev Food Sci Food Saf 14:106–122

- Hellwig M (2020) Analysis of protein oxidation in food and feed products. J Agricult Food Chem 68:12870–12885
- 3. Estévez M (2015) Oxidative damage to poultry: from farm to fork. Poultry Sci 94: 1368–1378
- 4. Xiong YL, Guo A (2021) Animal and plant protein oxidation: chemical andf property significance. Foods 10:40
- Estévez M, Xiong Y (2019) Intake of oxidized proteins and amino acids and causative oxidative stress and disease: recent scientific evidences and hypotheses. J Food Sci 84:387–396
- 6. Estévez M (2021) Critical overview of the use of plant antioxidants in the meat industry: opportunities, innovative applications and future perspectives. Meat Sci 181:108610

- 7. Estévez M (2011) Protein carbonyls in meat systems: a review. Meat Sci 89:259–279
- 8. Estévez M, Padilla P, Carvalho L et al (2019) Malondialdehyde interferes with the formation and detection of primary carbonyls in oxidized proteins. Redox Biol 26:101277
- Levine RL, Garland D, Oliver CN et al (1990) Determination of carbonyl content in oxidatively modified proteins. In: Packer L, Glazer AN (eds) Oxygen radicals in biological systems part B: oxygen radicals and antioxidants, vol 186. Academic, pp 464–478
- Soglia F, Petracci M, Ertbjerg P (2016) Novel DNPH-based method for determination of protein carbonylation in muscle and meat. Food Chem 197:670–675



# **Chapter 11**

## Lipid Oxidation (Primary and Secondary Products)

## Yasmim S. V. Leães, José Manuel Lorenzo, Alexandre J. Cichoski, Roger Wagner, Eva María Santos, Jorge F. Reyes, and Paulo C. B. Campagnol

#### Abstract

Lipid oxidation is one of the major causes of deterioration in the quality of meat and meat products. In addition to impairing the sensory quality, it is also responsible for reducing the nutritional quality and producing toxic compounds. The quantification of both the primary and secondary oxidation products is essential to establish strategies to prevent quality losses.

Several methodologies have been used to assess the lipid oxidation in foods, and the methods most commonly used to determine the primary and secondary products of lipid oxidation of meat and meat products will be discussed in this chapter. The chapter will describe in detail the procedures for determining the peroxide values, conjugated compounds, TBARS, and hexanal levels of meat and meat products.

Key words Lipid oxidation, Meat, Meat products, Peroxide values, Conjugated compounds, TBARS, Hexanal

#### 1 Introduction

Lipid oxidation is one of the major causes of deterioration of meat and meat products. In addition to reducing the nutritional quality due to the degradation of vitamins and essential fatty acids, the lipid oxidation reactions lead to the appearance of undesirable compounds that impair the sensory quality of the products. The main sensory modifications include rancid flavor and aroma, unpleasant color, and texture changes in some cases [1]. Furthermore, the oxidation reactions produce toxic compounds that are associated with the emergence of various human pathologies, including various types of cancer [2].

The most important mechanism of lipid oxidation in meat is auto-oxidation, which is triggered by the reaction of oxygen with the double bonds of fatty acids, with the free radical's formation as the main initiator [3]. This mechanism is divided into three steps:

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_11,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

(a) initiation, producing peroxides and hydroperoxides as the main compounds, which will later originate oxidation products responsible for the rancid flavor and aroma; (b) propagation, which consists of the displacement of double bonds during the initiation step, with the formation of conjugated dienes and trienes. During this step, hydrogens are subtracted from fatty acids and transformed into free radicals, triggering an autocatalytic process that accelerates the peroxidation reaction, with structural changes in the fatty acid molecules; (c) and termination, which consists of the reaction of free radicals with each other and with primary compounds, originating stable, non-reactive, and potentially toxic secondary products, such as aldehydes, especially malonaldehyde, and ketones. At this step, the rancid flavor and aroma are accentuated and perceptible to human senses [4].

Considering the importance of lipid oxidation reactions for the quality of meat and meat products, it is necessary to determine both the primary and secondary products, aimed to establish strategies to prevent quality losses [2]. Several methodologies have been used to assess the lipid oxidation of foods, and this chapter will discuss the protocols most commonly used to determine the primary and secondary products of this reaction in meat and meat products.

Hydroperoxides are the main primary products of lipid oxidation. The determination of hydroperoxides, also called peroxide value (PV), is one of the indicators used to evaluate the formation of primary oxidation compounds in meat and meat products. An increase in PV is observed in the early stages of lipid oxidation, which can decrease in the more advanced stages due to the decomposition of hydroperoxides [2]. Thus, a low PV may not be indicative of good oxidative stability. PV can be determined by volumetric methods, UV-visible spectral methods, chromatographic techniques, iodide values, and ferrous oxidation method. Among these methods, ferrous oxidation, proposed by Shanta and Decker [5], is considered a relatively simple method with greater stability, as it has lower sensitivity of ferrous ions to spontaneous oxidation by oxygen present in the air. The determination of conjugated dienes and trienes has also been used to monitor the formation of primary compounds of lipid oxidation, once it is a simple and fast technique that uses a small amount of organic solvents [2]. In this technique, first, the conjugated compounds are extracted from the sample using organic solvents, and then the concentration of conjugated dienes and trienes in the organic phase is measured at 234 and 266 nm, respectively, using a UV-Vis spectrophotometer [6]. Recent studies have proven that the combined determination of PV and conjugated compounds can be a very effective strategy to monitor the changes in the early stages of lipid oxidation of meat and meat products [7–10].

Malonaldehyde (MDA) is one of the major by-products of lipid oxidation and is considered a marker of lipid oxidation, as it confers a rancid aroma even at low concentrations (>2 mg MDA/kg of sample) [2]. The TBARS (Thiobarbituric Acid Reactive Substances) assay is used to quantify MDA. First, MDA is extracted from the sample, and then MDA reacts with the TBA reagent (2-thiobarbituric acid), forming a red complex that is measured at 532 nm in a UV-Vis spectrophotometer. Whereas the TBA reagent can also react with other compounds, this methodology is not specific to determine MDA. However, the TBARS assay is one of the most used methodologies to assess the lipid oxidation of meat and meat products, once it is a simple methodology and the results are well correlated with deterioration in sensory quality.

Hexanal has been reported for its positive correlation with MDA concentrations [11], and quantitatively figures as the main volatile compound of lipid oxidation in meat and meat products, making it an important oxidative marker. The increase in hexanal concentration, as well as MDA concentrations, has an inverse relationship with the sensory acceptance of meat and meat products. The main precursors of hexanal are the polyunsaturated fatty acids (PUFA), including linoleic, linolenic, and arachidonic acids [12]. Other straight-chain aldehydes (C5–C9), saturated or unsaturated, also contribute to the same purpose but are often found in lower concentrations. Both hexanal and other volatile oxidation products can be selectively determined by gas chromatographymass spectrometry (GC-MS). Although different procedures are used for sample preparation, headspace solid-phase microextraction (HS-SPME) has proven to be a viable alternative for volatiles extraction, as it is an easy-to-apply, solvent-free sampling technique, using mild temperature conditions, which minimize the degradation reactions of the meat matrix during extraction. In this context, the combination of HS-SPME and GC/MS for separation and quantification of hexanal, as well as other volatile compounds, has been used as an important tool in monitoring the lipid oxidation reactions in meat and meat products.

Given the above, this chapter will discuss the procedures for (1) determination of PV and conjugated compounds; (2) determination of TBARS; and (3) determination of hexanal in meat and meat products, with emphasis on the preparation of solutions, materials, care, and protocols.

#### 2 Peroxide Values and Conjugated Dienes and Trienes

2.1 Materials

All solutions should be prepared using distilled water and volumetric glassware established in the protocols and/or according to the amount of solution necessary for the analysis. For different volumes, the rule of three should be used to calculate the amount of reagent to be used. Although the solutions have no expiration date, a short-term estimate of the volume required should be performed. The solutions should be stored at room temperature except those requiring refrigeration, which should be informed in the protocols.

- 2.1.1 Lipid Extraction 1. BHT-stabilized chloroform solution: Dissolve 0.2 g of butylated hydroxytoluene (BHT) in 1000 mL of chloroform. The procedure must be carried out in a fume cupboard. Store the solution at room temperature.
  - 2. 1.5% sodium sulfate solution: Dissolve 15 g of anhydrous sodium sulfate in 1000 mL of distilled water. Store the solution at room temperature.
  - 3. Sodium sulfate anhydrous.
  - 4. Methyl alcohol.
  - 1. Barium chloride solution: Dissolve 0.4 g of barium chloride dihydrate in 50 mL of distilled water.
    - 2. Ferrous sulfate solution: Dissolve 0.5 g of iron (II) sulfate heptahydrate in 50 mL of distilled water.
    - 3. 10 N Hydrochloric Acid: In a 50 mL volumetric flask, add 8.2 mL of distilled water and top up with 37% hydrochloric acid. Store the solution in an amber bottle, under refrigeration  $(\pm 5 \ ^{\circ}C).$
    - 4. Ferrous chloride solution: In a beaker, mix 50 mL of barium chloride solution and 50 mL of ferrous sulfate solution under constant agitation. Add 2 mL of 37% hydrochloric acid and wait for the precipitation of barium sulfate. Filter the solution and store in an amber bottle, under refrigeration ( $\pm 5$  °C).
    - 5. 30% ammonium thiocyanate solution: Dissolve 30 g of ammonium thiocyanate and in 100 mL of distilled water. Store the solution in an amber bottle, under refrigeration ( $\pm 5$  °C).
    - 6. Chloroform: methanol (7:3): For 1000 mL solution, mix 700 mL of chloroform and 300 mL of methanol. Store the solvent mixture in an amber bottle under refrigeration ( $\pm 5$  °C).
- Solutions and Reagents for Solution 1: Dissolve 0.2421 g of ferric chloride hexahydrate in 50 mL of chloroform: methanol (7:3). The iron concentration in the mixture is 1.0 mg/L. Store the solvent mixture under refrigeration (5 °C).
  - Solution 2: Dilute 1 mL of Solution 1 to 100 mL using the mixture of chloroform and methanol (7:3). The iron concentration in the mixture is 1.0  $\mu$ g /L. Store the solvent mixture under refrigeration (5 °C).
  - Solutions 5 and 6 described above are also used to prepare the standard curve.

2.1.2 Determination of Peroxide Value

Solutions and Reagents

the Curve

2.1.3 Determination of This analysis requires only the cyclohexane ACS reagent.

Conjugated Dienes and Trienes

2.2 Methods All procedures must be performed at controlled room temperature  $(\pm 20 \ ^{\circ}C)$  and low lighting. Due to the use of organic solvents, it is necessary to use an appropriate chemical resistant mask with a filter for organic gases, nitrile gloves, and goggles. All chemical procedures must be carried out in a fume cupboard.

2.2.1 Lipid Extraction The determinations of PV and conjugated dienes/trienes require at least 100  $\mu$ g of fat per sample. To extract the necessary amounts of fat from the samples, cold extraction is performed according to the method of Bligh and Dyer [13]. Figure 1 shows the simplified scheme with a brief description of each step.

The technique to determine PV only requires fat extraction. However, the determination of fat content is necessary to determine conjugated dienes and trienes, as the fat content is used to calculate the results. Thus, the drying and weighing steps will be described, as well as other necessary protocols before analysis.

- $\label{eq:preparation of Materials} 1. \ Identify \ and \ place \ beakers \ in \ an \ oven \ set \ at \ 105 \ ^\circ C \ for \ 2 \ h.$ 
  - 2. Remove the beakers from the oven using tongs and place them into a desiccator. Wait 30 min or until completely cooled.
  - 3. On an analytical balance, weigh the beakers and put them back into the desiccator until analysis.

#### Lipid extraction method 1. 5 g of the sample 8. Centrifugation (2.500 rpm 10') 2. Add 4 mL of distilled water + 16 mL of methanol 9. Fat + chloroform phase extraction 3. Homogenize 30 s (at 13.500 rpm) 10. Mix the Fat + chloroform phase extraction with 1 g of sodium sulfate in vortex for 10" 4. Add 8 mL of chloroform 11. Filtration 5. Shake the tubes horizontally (70 rpm, 20 °C, 30') 12. 2 mL of filtrate for oven-drying at 105°C 2 hours 6. Add 8 mL of chloroform + 8 mL of 1.5% (see the material preparation and oven-drying method sodium sulfate solution steps ) 7. Shake the tubes manually for 30 s 13. Storage the filtrate at -18 °C in an amber flask to determination of peroxide values and conjugated diene and triene

Fig. 1 Schematic and simplified representation of the steps of the fat extraction

- 4. In 15 mL screw cap test tubes, add approximately 1 g of sodium sulfate. Close the tubes and reserve for the following steps.
- Determination Procedures 1. On an analytical balance, weigh 5 g of sample into 50 mL polypropylene centrifuge tubes and note the exact weight of the samples. It is desirable to respect a maximum deviation of  $\pm 0.03$  g.
  - 2. Add 4 mL of distilled water.
  - 3. Add 16 mL of methanol, and close tubes to prevent volatilization. Remove tube caps only in **step 4**.
  - 4. Homogenize the samples in an ultraturrax at 13,500 rpm for 30 s, with the tube immersed in ice water to prevent heating and increasing the oxidation rate. Clean the dispersion tool shaft between samples (*see* Note 1 of Subheading 2.3).
  - 5. Add 8 mL of BHT-stabilized chloroform to each tube and close the cap to prevent volatilization.
  - 6. Shake the tubes horizontally on a shaking table at 70 rpm for 30 min at 20 °C. Make sure the tubes are properly closed to prevent sample leakage.
  - 7. Add 8 mL of chloroform and 8 mL of 1.5% sodium sulfate solution, and close the lid. Shake the tubes manually for 30 s.
  - 8. Centrifuge the tubes at 2500 rpm for 10 min. It is not necessary to use a refrigerated centrifuge. It is important to balance the tubes before centrifugation, by forming pairs with similar weights (*see* **Note 2** of Subheading 2.3). Centrifugation allows the separation of the contents into two phases, consisting of a phase with methanol plus the food matrix and another phase at the bottom of the tube corresponding to the chloroform containing the extracted fat.
  - 9. Pipette the chloroform + fat phase into the test tube containing the sodium sulfate, obtained in **step 4** of Subheading "Preparation of Materials", and homogenize by vortexing for 10 s. Insert the pipette carefully, pressing it into the wall of the Falcon tube until it reaches the bottom. At this step, approximately 10 mL of the chloroform + fat phase can be pipetted.
  - 10. Filter the samples. Place 2 mL in the beakers separated in **step 3** of Subheading "Preparation of Materials". Store the remaining filtrate in amber vials sealed with a stopper and lid. Apply a layer of Parafilm<sup>®</sup> around the entire bottle cap to avoid the volatilization of chloroform. Store the filtrate under freezing until analysis of the peroxide values ( $\pm -18$  °C).
  - 11. Preheat the oven at 105 °C so that it reaches the desired temperature for the drying step.

- 12. Before placing the beakers containing 2 mL of sample in the oven, it is important to place them in the fume cupboard for chloroform evaporation. Then, place the samples in the oven at 105 °C for 2 h.
- Remove the beakers from the oven using tongs and place them into a desiccator. Wait 30 min or until completely cooled. Weigh the beakers containing the dried samples.

The fat content is calculated considering the difference between the weight of the empty beaker and the beaker containing the dried sample. The results are expressed in g of fat/100 g of sample.

2.2.2 Determination of Peroxide Values The determination of peroxide values described below is based on the methodology of Shanta and Decker [5], with some modifications. It is important to emphasize that there is no need to perform a standard curve for each repetition when the same standard  $Fe^{3+}$  solution (Solution 2, Subheading 2.1.2) is used. When a new standard solution is required, a new standard curve must be performed. The procedures to prepare the standard curve and to determine the peroxide values are described below, and a simplified scheme with a brief description of each step is shown in Fig. 2a.

Preparation of the Standard 1. For the solution

1. For the standard curve, a blank without the Fe<sup>3+</sup> standard solution, containing the mixture of chloroform and methanol



Fig. 2 Simplified scheme with a brief description of each step of the determination of peroxide values (a) and conjugated dienes and trienes (b)

#### Table 1

Concentrations of each reagent used to construct the standard curve of the determination of peroxide values

	Fe <sup>3+</sup> concentration (μg)	Volume of Fe <sup>3+</sup> standard solution	Volume of chloroform: methanol (mL)	Volume of ammonium thiocyanate (µL)
Blank	0	0	9.8	50
1	0.5	50 μL	9.7	50
2	1	100 µL	9.7	50
3	5	0.5 mL	9.3	50
4	10	l mL	8.8	50
5	15	1.5 mL	8.3	50
6	20	2 mL	7.8	50
7	30	3 mL	6.8	50
8	40	4 mL	5.8	50

and ammonium thiocyanate should be made, as well as eight points with different standard solution concentrations (0.5, 1, 5, 10, 15, 20, 30, and 40  $\mu$ L). The volume of chloroform: methanol decreases with increasing the volumes of standard solution, and the volume of ammonium thiocyanate solution is fixed at 50  $\mu$ L. In the end, a colorimetric curve is obtained, in which the greater the pink color, the greater the representation of the oxidized sample. Table 1 shows the concentrations of each reagent to construct the standard curve. Absorbance readings should be performed at 500 nm.

- 2. The absorbance values obtained for each  $\text{Fe}^{3+}$  concentration are used to calculate the standard curve, and an ideal  $R^2$  value should be above 0.99.
- Determination Procedures1. Prepare the water bath at 30 °C. This temperature is indicated<br/>to accelerate the chloroform volatilization process, and cannot<br/>be exceeded, aiming to prevent the fat oxidation of the sample.
  - 2. In a test tube, add 4 mL of the phase containing chloroform + fat, extracted by the Bligh and Dyer method [13] previously described.
  - 3. Use a nitrogen flow to dry the sample, until only a light-colored viscous liquid remains at the bottom of the test tube (fat) (*see* **Note 3** of Subheading 2.3).
  - 4. In a test tube, weigh 60  $\mu$ g of sample using an analytical balance.

- 5. Add 9.8 mL of chloroform: methanol (7:3) mixture. Close the tube cap and homogenize by vortexing for 10 s.
- 6. Add 50  $\mu$ L of 30% ammonium thiocyanate solution. Close the tube cap and homogenize by vortexing for 10 s.
- 7. Add 50  $\mu$ L of ferrous chloride solution. Close the tube cap and homogenize by vortexing for 10 s.
- 8. Place the samples in a dark environment for 5 min at controlled room temperature ( $\pm 20$  °C).
- Prepare the blank with all reagents except the sample. Absorbance readings should be performed in a spectrophotometer at 500 nm. A glass cuvette can be used due to the absorbance range used.
- 10. The results are calculated according to Eq. 1 and expressed in milliequivalent of peroxides/kg of lipids (meq peroxides/kg of lipids).

Peroxide value =  $(Abs 500 \text{ nm} \times m0)/(55.84 \times m1 \times 2)$  (1)

where: Abs 500 nm = absorbance of the sample in the spectrophotometer; m0 = 1/slope of the standard curve/55.84 = molar weight of iron; m1 = weight of fat (g); 2 = peroxide meq factor.

2.2.3 Determination of Conjugated Dienes and Trienes and Trienes
 The determination of conjugated dienes and trienes is based on the methodologies of Semb [14] and Recknagel and Glende [6]. Figure 2b shows a simplified scheme with a brief description of each step.

- Determination Procedures
- 1. Repeat step 1 of the peroxide determination.
- 2. In a test tube, place 70  $\mu$ L of the chloroform + fat phase, extracted by the Bligh and Dyer method as previously described.
- 3. Dry the samples in nitrogen flow (*see* Note 3 of Subheading 2.3).
- 4. In a fume cupboard, add approximately 3 mL of cyclohexane to the tubes containing the sample and homogenize by vortexing. Prepare the blank using only 3 mL of cyclohexane.
- 5. Absorbance readings should be performed simultaneously in a spectrophotometer, using a quartz cuvette, at 234 (dienes) and 266 nm (trienes).
- 6. The results are calculated according to Eqs. 2 and 3, and expressed in mg of lipids/mL of cyclohexane). It is worth noting that the results of the Bligh and Dyer analysis are used for the calculation.

Conjugated dienes = 
$$\frac{\text{Abs } 234 \text{ nm} \times \text{mg fat of the sample}}{3 \text{ mLcyclohexane}}$$
 (2)

Conjugated trienes = 
$$\frac{\text{Abs } 266 \text{ nm} \times \text{mg fat of the sample}}{3 \text{ mL cyclohexane}}$$
 (3)

- 2.3 Notes
  1. Before cleaning the dispersion tool shaft of the ultraturrax, it is necessary to ensure that no sample residue is retained. If there is any residue on the shaft, remove the sample with the aid of a needle and place it in the Falcon tube, since sample loss affects the result and the fat yield for the determination of peroxides and conjugated dienes/trienes. Then, wash the shaft with distilled water and dry it with a paper towel. It is not necessary to wash each replicate. At each different treatment, washing must be done to avoid interference of one sample with another. At the end of use, sanitize the stem according to the manufacturer's instructions.
  - 2. When using the centrifuge, there may be a variation in the weight of the tubes due to the lack of standardization (size, weight, tube model), as well as a difference in the weight of the samples. Therefore, before inserting the tubes into the centrifuge rotor, weigh each one and note the weight, grouping the tubes with approximate weight, and adjusting the weights with distilled water when necessary, for proper use of the centrifuge.
  - 3. It is important to check whether the amount of fat in the food matrix using 4 mL of the chloroform: fat phase will be sufficient to carry out the analysis. Therefore, it is important to standardize the amount of sample subjected to drying so that all samples have the minimum μg needed for analysis.

#### 3 Thiobarbituric Acid Reactive Substances (TBARS)

The simplified steps of the TBARS assay are shown in Fig. 3. The analytical protocol was based on the methodology of Bruna et al. [15] and Ripollés et al. [16].

- **3.1** *Materials* Use distilled water and analytical grade reagents to prepare all solutions (*see* **Note 1** of Subheading 3.3).
  - 1. 5% TCA solution: Dissolve 5.0 g of trichloroacetic acid (TCA) in 100 mL of distilled water.
  - 2. 0.08 M TBA solution: Dissolve 1.3 g of thiobarbituric acid (TBA) and 5.0 g of sodium hydroxide in 100 mL of a 50% acetic acid solution (*see* **Note 2** of Subheading 3.3).



Fig. 3 Schematic and simplified representation of the steps of the TBARS assay

- 3. 0.15% BHT solution: Dissolve 0.75 g of butylated hydroxytoluene (BHT) in 500 mL of ethanol.
- 4. 1 mM TEP standard solution (220.3 μg/mL): Exactly weigh 227.1 mg of 1,1,3,3-tetraethoxypropane (TEP) (97% purity) and dissolve in distilled water using a 1 L volumetric flask.
- 5. TEP working solution (22.03  $\mu$ g/mL): Transfer 10 mL of the standard TEP solution (220.3  $\mu$ g/mL) to a 100 mL volumetric flask and complete the volume with distilled water.

#### 3.2 Methods

3.2.1 Extraction of Malonaldehyde from the Samples

- 1. Weigh 5 g of the previously homogenized sample (note the weight) into 50 mL centrifuge tubes.
- 2. Add 1 mL of 0.15% BHT solution and 20 mL of 5% TCA solution.
- 3. Homogenize the tubes for 60 s in an ultraturrax at 12,000 rpm (*see* **Note 3** of Subheading 3.3).
- 4. Centrifuge at 4 °C for 10 min at 3000 rpm and filter the supernatant.

3.2.2 Preparation of the	1. Add 2 mL of 5% TCA solution into six 15 mL tubes.			
TEP Standard Curve	2. Add 2 mL of TEP working solution (22.03 $\mu$ g/mL) and homogenize by vortexing. This tube contains 11.015 $\mu$ g/mL of TEP.			
	3. Make successive dilutions to the desired concentration. For example, remove 2 mL from the tube with the concentration of 11.015 $\mu$ g/mL and place in another tube containing 2 mL of 5% TCA solution (Concentration of 5.50 $\mu$ g/mL). Discard 2 mL from the last tube.			
3.2.3 Colorimetric	1. In 15 mL tubes add:			
Reaction	(a) 2 mL of 5% TCA solution for the blank			
	(b) 2 mL of standard curve dilutions			
	(c) 2 mL of centrifuged and filtered sample.			
	2. Add 2 mL of 0.08 M TBA solution to all tubes.			
	3. Allow reacting for 5 min at 95 °C (see Note 4 of Subheading 3.3).			
	4. Cool to room temperature.			
	5. Measure the absorbance at 532 nm in a spectrophotometer using the blank to zero the equipment.			
3.2.4 Calculation of Malonaldehyde Concentration	1. The acid hydrolysis of 1 mol of TEP generates 1 mol of mal- onaldehyde, thus 220.3 g of TEP releases 72.03 g of malonal- dehyde. This relationship is used to calculate the malonaldehyde concentration for each point of the standard curve by dividing the TEP concentration by 3.057.			
	2. Using specific software, construct a graph of the standard curve, by placing the malonaldehyde concentration on the X-axis and the absorbance on the $\Upsilon$ -axis and generate the equation of the line.			
	<ol> <li>Substitute the absorbance of the samples in the equation of the line.</li> </ol>			
	For example : $y = 1.1419x + 0.024$ Sample absorbance : 0.30			
	0.30 - 0.024 = 1.1419x			
	$x = 0.24 \ \mu g$ of malonaldehyde ( <i>see</i> <b>Note 5</b> of Subheading 3.3)			
	4. Calculate the malonaldehyde concentration in the extract			
	$Z1 = (x \times (Volume TCA used in the extraction)/$			
	(Volume of TCA used in the colorimetric reaction)			
	$Z1=(0.24\times 20)/2$			

 $Zl = 2.4 \ \mu g$  of malonal dehyde (*see* **Note 6** of Subheading 3.3)

5. Calculate the malonaldehyde concentration in the sample

Malonaldehyde concentration = Z1/sample weight in grams Malonaldehyde concentration = 2.4/5

Malonaldehyde concentration =  $0.48 \ \mu g$  of malonaldehyde/g of sample (*see* **Note** 7 of Subheading 3.3).

- 3.3 Notes1. Use personal protective equipment and prepare the solutions in a fume cupboard due to the toxic and corrosive nature of the reagents.
  - 2. Use magnetic stirrer to dissolve TBA. The dissolution of TBA must not be carried out with heating and the presence of light to prevent degradation. The TBA solution should be used immediately after preparation.
  - 3. During homogenization, the tubes must be kept in an ice bath to prevent heating of the sample. It is recommended to stop the agitation every 20 s to remove the sample residues from the dispersion tool shaft of the ultraturrax.
  - 4. The colorimetric reaction must be carried out at 40 °C for 80 min for samples containing sugar in its composition.
  - 5. Amount of malonaldehyde present in 2 mL of filtrate added to the colorimetric reaction tube (Subheading 3.2.3).
  - 6. Amount of malonaldehyde present in 20 mL of TCA used to extract malonaldehyde from the sample (5 g) in the extraction step (Subheading 3.2.1).
  - 7. Final result of the TBARS values of the sample. The result can be expressed as  $\mu$ g malonaldehyde/g sample or per mg malonaldehyde/kg sample.

#### 4 Hexanal Determination

The steps of hexanal determination in meat and meat products through headspace solid-phase microextraction (HS-SPME) and gas chromatograph-mass spectrometry (GC/MS) are shown in Fig. 4. The analytical protocol is based on the methodologies of Wagner and Franco [17] and Domínguez et al. [12]. This protocol allows obtaining quantitative information of hexanal, and the result is expressed as area units of extraction ion chromatogram (AU-EIC) per gram of sample (AU-EIC × 10<sup>4</sup> g of sample). Thus, this technique allows evaluating the oxidation products in different meat matrices for various purposes, including manufacturing processes, or evaluation of raw materials or new

### Sample preparation for hexanal analysis





ingredients, as well as the shelf life of meat products. Additionally, the protocol comprises established chromatographic conditions, which allow the analyst to investigate other metabolites of the lipid or protein oxidation, through the identification of the mass spectra of other analytes, if applicable.

# **4.1** *Materials* Use analytical grade reagents to prepare the methanolic hexanal solutions (*see* **Note 1** of Subheading 4.3).

1. Hexanal solutions: Prepare a methanolic solution of hexanal (S1) by mixing 10  $\mu$ L of hexanal and 1 mL of methanol. Then,

prepare the qualitative solution using 100  $\mu$ L of methanolic solution S1 and 10 mL of methanol (S2).

- 2. SPME fiber installed in a GC-autosampler or a manual holder: Use SPME fused-silica fiber (10 mm length) coated with a 50/30-μm thickness of DVB/Car/PDMS (divinylbenzene/ carboxen/polydimethylsiloxane) (Supelco, Bellefonte, PA, USA) (*see* **Note 2** of Subheading 4.3).
- 3. The volatile compounds of meat samples should be analyzed in a gas chromatograph coupled to a mass spectrometer Shimadzu GC/MS QP2010 (Shimadzu Corporation, Japan). The GC injection port should be equipped with an SPME liner (internal diameter of 0.75 mm). The separation of analytes should be carried out in a ZB-Wax fused-silica capillary column (60 m  $\times$  0.25 mm i.d., 0.25 µm film thickness; Phenomenex, USA).

Meat samples should be ground in a food multiprocessor to enhance the sample homogeneity and particle size reduction.

- 1. Weigh 1 g of homogenized sample in a 20-mL extraction vial and immediately screw-capped with a laminated Teflon-rubber septum (Fig. 4a). For the analysis of the hexanal standard, 1  $\mu$ L of the methanolic hexanal solution (S2) should also be placed in a 20-mL vial.
- 2. The extractions of volatile compounds should be carried out at 40 °C for 50 min of fiber exposition in the sample headspace. To obtain this temperature, it is recommended to use a water bath with a controlled temperature and accuracy of 0.2 °C. Before extraction, the sample remains at this same temperature for 15 min, to reach a homogeneous temperature inside the vial. After the time of extraction, the fiber should be withdrawn into the needle, and then posteriorly inserted into the GC injection port for volatile compounds analysis (Fig. 4b).

The hexanal as well as other volatile compounds extracted by SPME fiber are separated, identified, and quantified using a gas chromatograph as following:

1. The volatile compounds are thermally desorbed from SPME fiber by inserting it into the injection port of the GC/MS at 260 °C for 10 min (Fig. 4c). The thermal desorption of the volatile compounds occurs in a split/splitless injector operating in a splitless mode for 1 min (after the splitter valve is open at 20:1). Helium is used as the carrier gas under a constant flow of 1.2 mL/min. After each injection, the SPME fiber should be cleaned in the SPME conditioning station at 260 °C for 2 min to ensure that the fiber is completely clean before the next extraction.

#### 4.2 Methods

4.2.1 Sample Preparation

4.2.2 Hexanal Determination



Fig. 5 Chromatograms of hexanal standard (a) and a pepperoni sausage sample red line, and hexanal standard—black line (b)

- 2. The hexanal and other volatile compounds are separated in a capillary column of fused-silica ZB-Wax. The oven temperature is programmed as initial oven temperature of 35 °C, remaining for 5 min, increasing at 2 °C/min until 80 °C, followed by an increase of 5 °C/min until 230 °C, held for 3 min.
- 3. The GC/MS interface temperature and ion source of MS detector are maintained both at 230 °C. The instrument is operated in the electron ionization mode (+70 eV) with a single quadrupole mass analyzer operating in full scan mode, collecting 2 scans/s over the range of 35-350 m/z.
- 4. The hexanal identification in the sample is done by comparing the retention times and the spectra of the chromatographic peak of the sample and hexanal standard (*see* Note 3 of Subheading 4.3). The hexanal mass spectrum is shown in Fig. 5a. Due to greater selectivity, it is suggested to use the ion m/z56 (82% of intensity) as quantitative ion and ions 57 (39%), 72 (18%), and 82 (17%) as qualitative ions.

	5. The final results are expressed as area units of the extraction ion chromatogram (AU-EIC) per gram of sample (AU-EIC $\times 10^4$ g of sample) using the quantifier ion of $m/z$ 56 (Fig. 5b).
4.3 Notes	1. Use personal protective equipment and prepare the solutions in the fume cupboard due to the toxic and corrosive nature of the reagents.
	2. Before the analysis, the SPME fiber should be conditioned by heating in an SPME fiber conditioning station or a GC injection port (in split mode) at 260 °C for 30 min, according to the manufacturer's instructions.
	3. Chromatographic selectivity must be verified when determin- ing the hexanal. This can be assessed by spectral comparisons of ions of the hexanal standard and the corresponding peak of the samples. The similarity index between the spectra must be greater than 95%.

#### Acknowledgments

This work was supported by National Council for Scientific and Technological Development (CNPq) and Coordination for the Improvement of Higher Education Personnel (CAPES finance code 001). Authors are members of the HealthyMeat network, funded by CYTED (ref. 119RT0568).

#### References

- Amaral AB, Silva MV, Lannes SCS (2018) Lipid oxidation in meat: mechanisms and protective factors—a review. Food Sci Technol 38: 1–15
- 2. Domínguez R, Pateiro M, Gagaoua M et al (2019) A comprehensive review on lipid oxidation in meat and meat products. Antioxidants 8:429
- 3. Cheng JH (2016) Lipid oxidation in meat. J Nutr Food Sci 6:3
- Huang X, Ahn DU (2019) Lipid oxidation and its implications to meat quality and human health. Food Sci Biotechnol 28:1275–1285
- 5. Shanta NC, Decker EA (1994) Rapid, sensitive, iron-based spectrophotometric methods for determination of peroxide values of food lipids. J AOAC Int 77:2
- 6. Recknagel RO, Glende EA (1984) Spectrophotometric detection of lipidic conjugated dienes. Methods Enzymol 105:331–337
- 7. Alves LL, Silva MS, Flores DRM et al (2018) Effect of ultrasound on the physicochemical and microbiological characteristics of Italian salami. Food Res Int 106:363–373

- Athayde DR, Flores DRM, Silva JS et al (2017) Application of electrolyzed water for improving pork meat quality. Food Res Int 100: 757–763
- Cichoski AJ, Flores DRM, Menezes CR et al (2019) Ultrasound and slightly acid electrolyzed water application: An efficient combination to reduce the bacterial counts of chicken breast during pre-chilling. Int J Food Microbiol 301:27–33
- 10. Silva JS, Voss M, Menezes CR et al (2020) Is it possible to reduce the cooking time of mortadellas using ultrasound without affecting their oxidative and microbiological quality? Meat Sci 159:107947
- 11. Brunton NP, Cronin DA, Monahan FJ et al (2000) comparison of solid-phase (SPME) fibres for measurement of hexanal and pentanal in cooked turkey. Food Chem 68:339–345
- Domínguez R, Purriños L, Pérez-Santaescolástica C et al (2019) Characterization of volatile compounds of dry-cured meat products using HS-SPME-GC/MS technique. Food Anal Methods 6:1263–1284

- Bligh EG, Dyer WJ (1959) A rapid method of total lipid extraction and purification. Can J Biochem Physiol 37:8
- 14. Semb TN (2012) Analytical methods for determination of the oxidative status in oils. Doctoral thesis, Norwegian University of Science and Technology
- 15. Bruna JM, Ordonez JA, Fernández M et al (2001) Microbial and physico-chemical changes during the ripening of dry fermented sausages superficially inoculated with or having

added an intracellular cell-free extract of Penicillium aurantiogriseum. Meat Sci 59:87–96

- 16. Ripollés S, Campagnol PCB, Armenteros M et al (2011) Influence of partial replacement of NaCl with KCl, CaCl<sub>2</sub> and MgCl<sub>2</sub> on lipolysis and lipid oxidation in dry-cured ham. Meat Sci 89:58–64
- 17. Wagner R, Franco MRB (2012) Effect of the variables time and temperature on volatile compounds extraction of salami by solid phase microextraction. Food Anal Methods 5: 1186–1195



# **Chapter 12**

## **Volatile Organic Compound Profile**

## Olalla López-Fernández, Rubén Domínguez, Laura Cutillas, Paulo E. S. Munekata, Laura Purriños, José Manuel Lorenzo, Nestor Sepúlveda, Alfredo Teixeira, and Mirian Pateiro

#### Abstract

The intake of meat and meat products provide certain benefits to the consumer as these foods are an important source of vital nutrients such as minerals, fatty acids, vitamins, essential amino acids and proteins. However, it must be taken into account that due to their characteristics, this type of foods is susceptible to deterioration, either by enzymatic degradation, lipid oxidation or by the action of the microorganisms. The spoilage reactions can produce textural and color changes and also the formation of volatile organic compounds that could origin nasty odors in meat which can lead to rejection by consumers. To avoid this, it is necessary to develop new faster and easier analytical techniques to determine volatile organic compounds in meat and meat products.

A procedure for the analysis of volatile organic compounds in meat and meat products is detailed below, which includes an extraction stage by solid-phase microextraction with fused-silica fiber coated with divinylbenzene/carboxen/polydimethylsiloxane and its subsequent analysis by gas chromatography with mass spectrometry. All steps and conditions proposed by us are described clearly with the aim reproduced by other researchers.

Key words Volatile compounds, Meat, Meat products, SPME extraction, Gas chromatography-mass spectrometry, Food analysis

#### 1 Introduction

Meat and meat products are foods very consumed. This is consumed largely because they constitute an important source of protein in the diet [1, 2]. However, their consumption is in great part affected by their nutritional quality and sensory perception [3, 4]. These characteristics specifics of each kind of meat and meat products are greatly influenced by diverse factors such as the animal characteristics (sex, feed, breed, muscles, rearing systems, genotypes, etc.) [5, 6], the meat processing (freezing, cooking, smoking process, mincing degree, etc.), the ingredients (salt amount, adding of species, meat/fat ratio, etc.), and the ripening

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_12,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

conditions (time, temperature, and relative humidity) [5, 7, 8]. All these factors have an influence over the meat and meat products aroma.

The aroma is formed by a hundred molecules called volatile organic compounds (VOCs) [7]. The principal VOCs that can be found in meat and meat products can be separated into different chemical families between including alcohols, aldehydes, esters, furans, hydrocarbons, ketones, lactones, organic acids, and terpenes [8]. The importance of each compound in the aroma depends on the concentration of each VOC and the olfactory threshold [8, 9].

For their features, meat and meat products are foods highly susceptible to spoilage and the formation of unwanted odors and flavors that are produced as a result of microorganisms degradation or the lipid oxidation and enzymatic processes [4]. Some volatile organic compounds (VOC) are created through these types of reactions that are considered a critical factor of food flavor quality [10, 11]. In addition, it should be noted that not only do these processes give rise to unwanted tastes/odors, but in the case of cured meat, for example, these processes release desirable volatile compounds, which are part of the characteristic flavor of meat products. Therefore, it is vital to correctly monitor the evolution and release of volatile compounds, both from the point of view of seeing a possible degradation and to monitor the correct development of the desired aroma/flavor [8].

Nowadays, is necessary the development of new methods of analysis of volatile organic compounds in meat and meat products to control their quality due to increasing consumer demands for safe and high-quality food [4, 12]. However, the quantification of these compounds in this kind of food is not easy because the sampling techniques and analytic instrumentation used are complex [11]. Several sampling techniques are available for the analysis of VOCs released during meat and meat products manufacture, such as the solvent extraction, simultaneous steam distillation-extraction, static and dynamic headspace and solid-phase microextraction (SPME) [4]. However, the most common technique used in meat and meat products is solid-phase microextraction (SPME) [13]. This fact is caused for SPME a rapid and inexpensive analytical technique enabling VOCs sampling in the headspace of a sample [12]. Nevertheless, to insure a high efficiency of SPME is necessary to control several parameters as the type of fiber, the sample volume, time, temperature, and stirring [14]. Regarding the analysis technique, the gas chromatography (GC) coupled to mass spectrometry (MS) has been the most used technique for VOCs analysis in different meat samples and meat products. Recent studies such as those cited below in beef [15, 16], pork [17, 18], lamb [10, 19], bovine [11], chicken [12], and meat products [8, 20-22] evidence this.

With this in mind, the objective of this book chapter is to clearly and fully explain a procedure for determining and quantifying volatile organic compounds in meat and meat products so that they can be reproduced in the future by other researchers. In this case, two stages are specified, the solid-phase microextraction extraction (SPME) of the volatile organic compounds and the subsequent chromatographic analysis by gas chromatographymass spectrometry (GC-MS).

2	Μ	ate	eria	ls

2.1 Volatile	Organic •	Fused-silica fiber (10 mm length) coated with a 50/30-mm
Compounds E	xtraction	thickness of DVB/CAR/PDMS (divinylbenzene/carboxen/
		polydimethylsiloxane) (Supelco, Bellefonte, PA, USA).

• 20 mL vials with screw-capped with a laminated Teflon-rubber disc (Agilent Technologies, Santa Clara, CA, USA).

**2.2 Gas** The extraction of volatile organic compounds (VOC) was performed using solid-phase microextraction (SPME) with an autosampler Pal RTC-120 (Agilent Technologies) and the separation, identification, and quantification of them were carried out using a gas chromatograph Agilent mod. 7890B (Agilent Technologies) equipped with a mass selective detector Agilent mod. 5977B (Agilent Technologies). For the separation of the volatile organic compounds, a DB-624 capillary column (30 m, 250 µm i.d., 1.4 µm film thickness; Agilent Technologies) was used.

#### 3 Methods

**3.1 SPME Extraction** Prior to the analysis, the fiber (fused-silica fiber DVB/CAR/ PDMS) has conditioned by heating in an SPME Fiber Conditioning Station at 270 °C for 30 min.

For headspace SPME (HS-SPME) extraction, weigh  $1 \pm 0.02$  g of each sample in a 20-mL vial and then screw-capped with a laminated Teflon-rubber disc (*see* Notes 1–3).

The SPME extraction was carried out at  $37 \,^{\circ}$ C for  $30 \,^{\circ}$ min, after sample equilibration time at the same temperature for  $15 \,^{\circ}$ min, to ensure a homogeneous temperature for the sample and headspace. A resume of SPME extraction conditions is shown in Table 1.

Figure 1 shows a summary of the extraction process of volatile compounds in meat samples and meat products by SPME extraction.
Table 1				
Summary of SPME extraction	for the	volatile	compound	analysis

SPME extraction	
Fiber type	Fused-silica fiber (10 mm length) coated with a 50/30-mm thickness of DVB/CAR/ PDMS
Fiber conditioning	270 °C during 30 min
Sample incubation	37 °C during 15 min
Sample extraction	37 °C during 30 min

#### 3.2 Volatile Organic Compounds Identification and Quantification (GC-MS)

Separation, identification, and quantification of volatile organic compounds were carried out using a gas chromatograph-mass spectrometer, following the next conditions (Table 2) (*see* Notes 4 and 5):

- After the extraction process, the fiber was transferred to the injection port of the gas chromatograph-mass spectrometer (GC-MS) system. Then the volatile organic compounds were desorbed from the SPME fiber in the injection port at 260 °C for 8 min (splitless mode; helium pressure 9.59 psi). After each injection, the SPME fiber was clean for 2 min at 270 °C in the SPME Conditioning Station to guarantee that fiber is completely clean before the next sample extraction. Helium was used as carrier gas at a constant flow of 1.2 mL/min (9.59 psi). A DB-624 capillary column (30 m, 0.25 mm i.d., 1.4 µm film thickness; Agilent Technologies) was used for volatile compound separation. This column has excellent resolution and a wide operating temperature range.
- 2. The oven temperature program was isothermal for 10 min at 40 °C, raised to 200 °C at 5 °C/min, then raised to 250 °C at 20 °C/min and held for 5 min.
- 3. The mass detector transfer line was maintained at 260 °C. The ion source used was an Extraction Source Xtr EI 350 (Agilent Technologies, Santa Clara, CA, USA). The mass spectra were obtained using a 5977B mass selective detector working in electronic energy at 70 eV, with an electron multiplier voltage of about 900 V (gain factor = 1) and collecting data at 2.9 scans/s over the range m/z 40–550 in scan acquisition mode. The mass source was maintained at 230 °C while the mass quad was set at 150 °C. The total time for chromatographic analysis is 49.5 min. Figure 2 shows an example of volatile organic compounds chromatogram of *Cecina*.





**3.3 Data Processing** Data acquisition and equipment control was carried out using GC software MassHunter Acquisition B07.05, while data were analyzed with MassHunter Quantitative analysis B.07.01 (Agilent Technologies).

#### Table 2

Summary of the chromatographic and mass spectrometer conditions used for the volatile compound analysis

Chromatographic and mass spectrometer conditions				
Inlet	Temperature (° C)	Pressure (psi)	Total flow (mL/min)	Desorption time
Splitless mode	260	9.59	19.2	8 min
Oven	Rate (° C/min)	Temperature (° C)	Hold time (min)	Total time (min)
Initial	-	40	10	10
Ramp 1	5	200	-	42
Ramp 2	20	250	5	49.5
Column	Characteristics	Temperature (° C)	Pressure (psi)	Flow (mL/min)
DB-624	$30 \text{ m} \times 250  \mu\text{m} \times 1.4  \mu\text{m}$	40 (Initial)	9.59	1.2
Transfer line	e temperature	260 °C		
Detector	Temperature (° C)	m/z range	Mass source temperature	Mass quadrupole temperature
MS	260	40-550	230	150



Fig. 2 Volatile organic compounds chromatogram of Cecina

After samples analysis, a new method from acquired scan data with library search was created. For the integration Agile2 algorithm was used, while the peak detection was carried out by deconvolution. For method creation, a representative number of samples must be used. The compounds identification was carried out by comparison of their mass spectra with those contained in the NIST14 library (National Institute of Standards and Technology, Gaithersburg). The identification was considered correct only for the peaks with a match factor > 85%. After integration, peak detection and identification of each compound, the extracted ion chromatogram (EIC) from the quantifier ion was obtained from each peak. The final results were expressed as area units of the EIC ×  $10^4$  per gram of sample (AU-EIC ×  $10^4$ /g of sample).

#### 4 Notes

- 1. Homogenize the sample well to ensure a correct extraction of all the volatile compounds. In samples without chopping, it is necessary to grind (using a mincer) to achieve greater homogenization. However, it is necessary to avoid overheating of the sample since increasing the temperature of the sample could generate new and unwanted volatile compounds (*see* **Note 6**).
- 2. Make sure that the sample is well homogenized and that it remains at the bottom of the vial to prevent the fiber from touching the sample.
- 3. It the samples are not analyzed immediately, it is necessary to preserve them under vacuum and keep them frozen until use. However, it is advisable to analyze them fresh, since these processes can lead to unwanted changes.
- 4. A blank should be injected to compare the peaks/spectra with the injected samples, and to be able to detect potential equipment or SPME fiber problems.
- 5. To ensure a correct operation of the mass spectrometer, it is necessary to carry out a Tune Evaluation approximately once a week (to check the mass properties) and an Etune once a month to calibrate the masses. These tests apply different voltages to adjust various parameters of the mass spectrometer through the infusion of a tuned compound, commonly perfluorotributylamine (PFTBA). With this, it is possible to increase the sensitivity of the equipment and the signal/noise ratio.
- 6. Take into account that the fiber supports approximately between 90 and 100 injections, from that number it is necessary to replace it with another.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568).

#### References

- Pateiro M, Borrajo P, Campagnol PCB et al (2019) Extraction of valuable compounds from meat by-products. In: Barba FJ, Roselló-Soto E, Brncic M et al (eds) Green extraction and valorization of by-products from food processing. CRC Press, Boca Raton, FL, pp 55–90
- Kausar T, Hanan E, Ayob O et al (2019) A review on functional ingredients in red meat products. Bioinformation 15:358–363
- Bosse (née Danz) R, Wirth M, Konstanz A et al (2017) Determination of volatile marker compounds in raw ham using headspace-trap gas chromatography. Food Chem 219:249–259
- 4. Met A, Şahin Yeşilçubuk N (2017) Comparison of two volatile sampling techniques based on different loading factors in determination of volatile organic compounds released from spoiled raw beef. Food Anal Methods 10: 2311–2324
- Leroy F, Vasilopoulos C, van Hemelryck S et al (2009) Volatile analysis of spoiled, artisan-type, modified-atmosphere-packaged cooked ham stored under different temperatures. Food Microbiol 26:94–102
- 6. Cartoni Mancinelli A, Silletti E, Mattioli S et al (2021) Fatty acid profile, oxidative status, and content of volatile organic compounds in raw and cooked meat of different chicken strains. Poult Sci 100:1273–1282
- Kosowska M, Majcher MA, Fortuna T (2017) Volatile compounds in meat and meat products. Food Sci Technol 37:1–7
- Domínguez R, Purriños L, Pérez-Santaescolástica C et al (2019) Characterization of volatile compounds of dry-cured meat products using HS-SPME-GC/MS technique. Food Anal Methods 12:1263–1284
- Rivas-Cañedo A, Nuñez M, Fernández-García E (2009) Volatile compounds in Spanish dry-fermented sausage "salchichón" subjected to high pressure processing. Effect of the packaging material. Meat Sci 83:620–626
- 10. Gravador RS, Serra A, Luciano G et al (2015) Volatiles in raw and cooked meat from lambs fed olive cake and linseed. Animal 9:715–722
- Acevedo CA, Creixell W, Pavez-Barra C et al (2012) Modeling volatile organic compounds released by bovine fresh meat using an integration of solid phase microextraction and databases. Food Bioprocess Technol 5:2557–2567

- Mikš-Krajnik M, Yoon YJ, Yuk HG (2015) Detection of volatile organic compounds as markers of chicken breast spoilage using HS-SPME-GC/MS-FASST. Food Sci Biotechnol 24:361–372
- Balasubramanian S, Panigrahi S (2011) Solid-Phase Microextraction (SPME) techniques for quality characterization of food products: a review. Food Bioprocess Technol 4:1–26
- Iglesias J, Medina I (2008) Solid-phase microextraction method for the determination of volatile compounds associated to oxidation of fish muscle. J Chromatogr A 1192:9–16
- 15. Bueno M, Resconi VC, Campo MM et al (2019) Development of a robust HS-SPME-GC-MS method for the analysis of solid food samples. Analysis of volatile compounds in fresh raw beef of differing lipid oxidation degrees. Food Chem 281:49–56
- 16. Argyri AA, Mallouchos A, Panagou EZ et al (2015) The dynamics of the HS/SPME-GC/ MS as a tool to assess the spoilage of minced beef stored under different packaging and temperature conditions. Int J Food Microbiol 193: 51–58
- 17. Sun Y, Fu M, Li Z et al (2018) Evaluation of freshness in determination of volatile organic compounds released from pork by HS-SPME-GC-MS. Food Anal Methods 11:1321–1329
- 18. Zareian M, Böhner N, Loos HM et al (2018) Evaluation of volatile organic compound release in modified atmosphere-packaged minced raw pork in relation to shelf-life. Food Packag Shelf Life 18:51–61
- 19. Vasta V, Ventura V, Luciano G et al (2012) The volatile compounds in lamb fat are affected by the time of grazing. Meat Sci 90:451–456
- 20. Lorenzo JM, Carballo J, Franco D (2013) Effect of the inclusion of chestnut in the finishing diet on volatile compounds of dry-cured ham from celta pig breed. J Integr Agric 12: 2002–2012
- 21. Marušić N, Petroví M, Vidaček S et al (2011) Characterization of traditional Istrian dry-cured ham by means of physical and chemical analyses and volatile compounds. Meat Sci 88:786–790
- 22. Wang J, Jin G, Zhang W et al (2012) Effect of curing salt content on lipid oxidation and volatile flavour compounds of dry-cured turkey ham. LWT-Food Sci Technol 48:102–106



# **Chapter 13**

## **Proteomics**

## María López-Pedrouso, José Manuel Lorenzo, and Daniel Franco

#### Abstract

This text provides a brief overview of specific issues in the meat industry which may be resolved using the current proteomics-based methodology. Proteins as key to understanding biological processes as well as to change the characteristics of meat products offer enormous potential in the context of meat science. The quantification, analysis and curation of proteomics data, and particularly, the industrial applications of these new findings will become a more essential tool in the coming years.

The methodology described in this chapter presents the basis for quantification, analysis, and curation of proteomics data, representing the pillars of main proteomic methods. The main steps include 1. Extraction and quantification of proteins; 2. Separation of proteins by isoelectric point; 3. Separation of proteins by molecular weight; 4. Staining of proteins; 5. Tryptic digestion with enzymes; and 6. identification of proteins by mass spectrometry.

More recent proteomics approaches are beyond the scope of this chapter. Therefore, all steps are described in detail so that the conditions proposed by us can be reproduced by other researcher groups. Although protocols and methodology are specifically obtained from our lab, these can be easily adapted for every laboratory.

Key words 2-DE, Protein isoelectric point, MALDI-TOF, Meat products

#### Abbreviations

Two-dimensional electrophoresis
(3-((3-cholamidopropyl) dimethylammonium)-1-propanesulfonate)
Dithiothreitol
Immobilized pH gradient
Matrix-assisted laser/desorption ionization-time of flight
Molecular weight
Isoelectric point
Peptide Mass Fingerprint
Persulfate ammonium
Sodium dodecyl sulfate

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_13, © The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

SDS-PAGE	Sodium dodecyl sulfate polyacrylamide gel electrophoresis
TCA	Trichloroacetic
TEMED	Tetramethylethylenediamine
Tris	Tris(hydroxymethyl)aminomethane

#### 1 Introduction

The protein content of meat products has a critical influence on nutritional quality, sensory and textural properties. Important quality traits of fresh meat, like tenderness, color, and water holding capacity are closely linked to their proteins and posttranslational modifications. Indeed, during meat aging, the rate of pH fall is influencing the protein denaturation, myofibrillar structures, and cellular shrinkage causing variations in meat quality [1, 2]. Moreover, protein changes such as protein oxidation has an important role in sensory attributes and other quality traits of the meat products [3], as well as proteolysis caused by enzymes during the ripening of dry-cured meat products [4]. Other processes like protein carbonylation, breakdown, and aggregation also imply solubility and functionality changes that lead to a reduction of digestibility, gelation, emulsification, and water holding capacity [5]. On the other hand, the quality of processed meats is particularly linked to the nutritional value of proteins, composed of essential amino acids and bioactive peptides. Indeed, it has been reported that peptides released by digestion gastric or enzymatic hydrolysis from animal proteins may achieve a high biological activity and bioavailability [6]. Within the meat industry, quality control should be performed using protein biomarkers determining phenotypes and predicting meat quality features. The aim is to identify proteins whose expression or abundance is linked to a phenotypic trait [7], influencing the final meat product quality. Briefly, it has been described an overview of specific issues in the meat industry which may be resolved using the current proteomics-based methodology. For the above reasons, the knowledge of meat proteins and their modifications need to be investigated further for the improvement of meat quality in all these issues.

The growing use of omics technologies including proteomics brings significant benefits to food sciences. In this regard, proteomics as the large-scale analysis of proteins has mainly been developed for gel-based techniques and mass spectrometry in these years. The identification of protein patterns using two-dimensional electrophoresis (2-DE) for protein separation implies a first separation based on their isoelectric point (pI) and a second step separated based on molecular weight (Mr). These two parameters are independent resulting in an effective separation method. The analysis of 2-DE gel is performed by software comparing the images to



Fig. 1 Main steps to carry out separation and identification of proteins

identify different spots in quantitatively and qualitatively terms. Afterwards, the preferred method to identify the proteins in gels is Peptide Mass Fingerprint (PMF) using matrix-assisted laser/desorption ionization-time of flight (MALDI-TOF) mass spectrometry. For this purpose, it is necessary to carry out enzymatic digestion that breaks down protein with great specificity. This specific fingerprint is used to achieve the identification of the protein searching in databases.

The in-depth knowledge of protocols for each food matrix is a key step towards reliable data of proteins. 2-DE is the most common technique used in food proteomics as it can separate different isoforms or even detecting post-translation modifications of proteins by using commercial dyes. However, variability associated with the technique should be minimized through a rigorous procedure in the laboratory. The main steps (Fig. 1) are: (1) Extraction of proteins and initial quantification by commercial kit; (2) First dimension or separation of proteins by p*I*; (3) Second dimension or separation of proteins by Mr.; (4) Staining of gels and comparison of gel images using a specific software package; (5) Tryptic digestion with commercial enzymes; and (6) Identification of proteins by mass spectrometry (e.g., MALDI-TOF).

Particularly in muscle sample tissues, before the proteomic analysis, a protein extraction protocol for 2-DE analysis should be carried out by TCA-acetone precipitation avoiding contaminants (mainly salts and detergents) and increasing concentration of the proteins in solution [8]. The proteins should be solubilized in a denaturing buffer with neutral chaotropic, zwitterionic or neutral detergent and reducing agents. Currently, the first dimension (known as isoelectric focusing) is often performed on an immobilized pH gradient (IPG) submitted to a high voltage where the proteins can move towards the pH produces a zero-net charge on protein. The second-dimension gel is based on the movement of the unfolded protein which contains SDS through a gel pore resulting in a separation by protein size. Inside the gel, the proteins are stained using Coomassie blue, SYPRO, or silver stain producing gel imaging. Thus, the protein spots could be detected by gel image analysis and the quantitative changes measured reflecting the expression of the proteins. After all, mass spectrometry as MALDI-TOF MS is being used to identify the proteins after tryptic digestion.

To sum up, it has been presented an overview of proteomic methodology to identify and quantify proteins in meat products. In the next sections, detailed information about the main steps will be provided. The methodology described in this chapter presents the basis for quantification, analysis, and curation of proteomics data, representing the pillars of main proteomic methods. Thus, protocols and methodology were specifically obtained from our lab, these can be easily adapted for every laboratory.

#### 2 Material and Methods

In the next sections, we will provide a detailed description of the main steps indicated in Fig. 1, developed in our lab.

2.1 Extraction and Measurement of Protein Content in the Solution After the collection of meat samples, they must be excised in small pieces, lyophilized and maintained at -80 °C to ensure the conservation before protein extraction. Afterwards, meat proteins will be precipitated using the TCA-acetone protocol. The protein content in the solution will be carried out to know exactly the amount of protein necessary to carry out the proteomic experiment. In our lab, we use the commercial CB-X protein assay kit (GBiosciences, USA) according to the instructions of the manufacturer using a microplate reader, but other protein quantification kits would also be valid.

Solutions and buffers: Prepare all solutions using ultrapure water and analytical grade reagents.

- Acetone (90%).
- *Extraction buffer* (10% TCA and 0.07% dithiothreitol (DTT) in acetone): Dissolve 6 g of TCA in 60 mL of acetone and add 4.2 mg of DTT.
- *Lysis buffer* (7 M urea, 2 M thiourea, 4% CHAPS, 10 mM DTT and 2% Pharmalytes): Dissolve 10.5 g of urea, 3.8 g of thiourea,

1 g of CHAPS, 38 mg of DTT, and 0.5 mL of Pharmalyte<sup>™</sup> pH 3–10 (GE Healthcare, Uppsala, Sweden) in 25 mL and make up to 25 mL with water. Stir for hours to dissolve completely at room temperature.

- *Wash buffer* (0.07% DTT in acetone): Dissolve 14 mg of DTT in acetone.
- 2.1.1 Protocol
  1. An amount of 50 mg of lyophilized meat tissue is suspended in 10 mL of extraction buffer (10% TCA and 0.07% DTT in acetone).
  - 2. A sonication step should be considered for cell disruption and homogenization of the solution in refrigerated conditions. The sonication should be in short pulses at 0  $^{\circ}$ C.
  - 3. The solution is maintained at -20 °C for at least 1 h.
  - 4. The solution is centrifuged at  $4500 \times g$  at  $4 \degree C$  for 30 min and the pellet is recovered.
  - 5. The pellet is rinsed two times with 1 mL of wash buffer (0.07%) DTT in acetone) followed by centrifugation as described in step 4.
  - 6. The last washing is using acetone (90%) and centrifugation (step 4).
  - 7. The final pellet is eluted in 1 mL of lysis buffer (7 M urea, 2 M thiourea, 4% CHAPS, 10 mM DTT, and 2% Pharmalytes) which should be stirred to dissolve completely in an Eppendorf Thermomixer.
  - 8. The concentration of protein in the final solution should be determined using the commercial CB-X protein assay kit (GBiosciences, USA) according to the instructions of the manufacturer in a microplate reader. A microplate reader is necessary to look up the absorbance at 595 nm. This method is compatible with detergents and other substances like reducing sugars, thiols, and chelating agents (protein standard is specifically provided to prepare a calibration plot).
- 2.2 First Dimension or Separation of Proteins by pl Isoelectric focusing with IEF by using Protean IEF Cell (Bio-Rad Laboratories, Inc., USA) and immobilized pH gradients is employed for the first-dimension separation (Fig. 2). The protein can be loaded on 11-cm, 18-cm and 24-cm Ready Strip IPGs (Bio-Rad Laboratories). In this sense, IPG strips with different linear pH linear or non-linear gradients (3–10 and 4–7) can be chosen according to the experiment.

*Solutions and buffers*: Prepare all solutions using ultrapure water and analytical grade reagents.



Fig. 2 Protean IEF Cell (Bio-Rad Laboratories)

- *Tris solution* (50 mM Tris pH 8.8): Dissolve 181.5 g of Tris in 750 mL of ultrapure water, adjust the pH to 8.8, and make up to 1000 mL with water.
- *Equilibration solution* (50 mM Tris pH 8.8, 6 M urea, 30% glycerol, and 2% SDS): 6.7 mL of Tris solution, 72.1 g of urea, and 60 mL glycerol are vigorously stirred for 30 min. Finally, 4 g of SDS is added and made up to 200 mL with ultrapure water. In the case of equilibration solution with 1% of DTT, dissolve 60 mg of DDT in 6 mL of equilibration solution for each strip, and in case of 2.5% of iodoacetamide, dissolve 150 mg of iodoacetamide in each 6 mL of equilibration solution for each strip.
- 2.2.1 Protocol
   1. According to the previous protein quantification, a volume containing 250 μg of total protein is dissolved in lysis buffer up to the total volume recommended by the manufacturer for each strip length (Table 1).
  - 2. The rehydration solution is completed by the addition of 0.6% DTT and 1% IPG buffer (Bio-Rad Laboratories). A complete dissolution takes place after a vigorous agitation.
  - 3. This sample should be distributed along a lane of the focusing tray from one electrode to another. Carefully place the IPG strips contacting the gel side with the sample. Finally, cover with mineral oil along the lane to avoid evaporation during the first electrophoresis according to Table 1. Place the plastic lid onto the focusing tray and introduce it into the Protean IEF Cell (Bio-Rad Laboratories, Inc., USA).
  - 4. The focusing conditions and ramping protocols have been optimized for this quantity of meat samples. A previous step of active rehydration with a voltage of 50 V during 12 h should

Strip length (cm)	Total volume recommended ( $\mu$ L)	Volume of mineral oil (mL)
7	125	1
11	200	1.5
17	300	2
18	315	2
24	450	2.5

# Table 1 Total volume of buffer for rehydration for each strip

#### Table 2

#### Focusing conditions for programming the unit, using a linear ramp in each step

	Voltage (V)	Time (min)
Step 1	250	30
Step 2	500	60
Step 3	1000	60
Step 4	4000	120
Step 5	8000	120
Step 6	10,000 <sup>a</sup>	

<sup>a</sup>A voltage of 10,000 V is applied until to reach final voltage in the last step

be performed and the settings for IEF are shown in Table 2. The final voltage depends on the length and pH of each IPG strip, therefore, necessary to consult the manual.

5. The equilibration of the strips is necessary to run the second dimension ensuring that the cysteines are reduced and alkylated. The strips should be placed in the equilibration tray for better treatment. The focused strips should be incubated firstly in equilibration solution (50 mM Tris pH 8.8, 6 M urea, 30% glycerol, and 2% SDS) with 1% DTT for 15 min and secondly in equilibration solution (50 mM Tris pH 8.8, 6 M urea, 30% glycerol, and 2% SDS) with 2.5% iodoacetamide for 15 min.

2.3 Second
Dimension or
Separation of Proteins
by Mr.
For SDS-polyacrylamide gel electrophoresis (SDS-PAGE), the protein unfolded and charged by anionic detergent SDS is carried out in an Ettan Dalt six gel system (GE Healthcare, Uppsala, Sweden) (Fig. 3). The gel pore size could be differently adjusted by percentages of polyacrylamide and cross-linker. In this regard, the more common percentage ranged from 10% to 15% w/v polyacrylamide gels.

*Solutions and buffers*: Prepare all solutions using ultrapure water and analytical grade reagents.



Fig. 3 Ettan Dalt six gel system (GE Healthcare, Uppsala, Sweden)

Table 3			
Composition (volume) of polyn	nerization solution a	ccording to acrylam	ide percentage

Percent of acrylamide	10%	12%	15%
Acrylamide solution	136 mL	171.3 mL	207 mL
Tris solution (50 mM Tris pH 8.8)	104 mL	103 mL	104 mL
Ultrapure water	162 mL	127 mL	95 mL
SDS (10%)	4.1 mL	4.1 mL	4.1 mL
PSA (persulfate ammonium) (10%)	4.1 mL	4.1 mL	4.1 mL
TEMED (N,N,N',N'-tetramethyl-ethyl-1,2-diamine)	82 µL	64 µL	48 µL

Values are calculated for the preparation of six gels in the Gel Caster (Ettan Dalt six gel system, GE Healthcare, Uppsala, Sweden)

- *Acrylamide solution:* Dissolve 7.8 g of bis-acrylamide and 292.2 g of acrylamide in ultrapure water to reach a total volume of 1 L in dark conditions.
- *Polymerization solution*: The polyacrylamide gels consist of the polymerization from acrylamide and bis-acrylamide initiated by TEMED and PSA to free radicals. Thus, the mixture for different percentages of acrylamide should contain the amounts shown in Table 3. This solution should be prepared with constant agitation and without interruptions. Finally, the solution is poured into the Gel Caster and covered with water-saturated isobutanol. The polymerization occurs for approximately 6 h at room temperature.
- 10× Electrophoresis buffer (50 mM Tris base, 384 mM glycine and 0.2% SDS): Dissolve 30.2 g of Tris, 144.2 g of glycine, and 10 g of SDS in distilled water reaching a total volume of 1 L. Thus,

 $1 \times$  electrophoresis buffer is obtained by adding 1 part of this solution to 9 parts of the distilled water and  $2 \times$  electrophoresis buffer by adding 2 parts of this to 8 parts of the distilled water.

- Agarose solution: Heat to boiling 0.5 g of agarose in 100 mL of 1× electrophoresis buffer for its dissolution and add traces of bromophenol blue.
- 2.3.1 Protocol 1. The IPG strips should be transferred to gels polymerized according to the previous section. It should be embedded the strip on the top of the second dimension and sealing the place, with molten agarose solution and bromophenol blue to ensure good contact between the gel and the IPG strip and to track the electrophoresis with the dye.
  - 2. The electrophoresis unit (Ettan Dalt six gel system, GE Healthcare, Uppsala, Sweden) is assembled according to the manual.
  - 3. The upper chamber is filled with 1 L of  $2 \times$  electrophoresis buffer meanwhile the lower chamber needs 4.5 L of  $1 \times$  electrophoresis buffer.
  - Electrophoresis runs applying 5 mA/gel, 2 h; 10 mA/gel for 3 h and finally 16 mA/gel until the color blue reaches the gel bottom indicating the end of the electrophoresis.
  - 5. After the electrophoresis, the gels are placed in plastic boxes to proceed to the staining.

Gel staining protocols should be chosen depending on the specific needs. Protein staining should be mainly carried out using Coomassie blue, silver staining or Sypro Ruby® stain (Bio-Rad Laboratories). In the case of silver staining, it is a very excellent sensitive method as well as very simple and cheap. The method is based on the selective reduction of silver in the proximity of protein molecules [9]. However, it is less used due to its incompatibility with the mass spectrometry necessary for protein identification. In this sense, coomassie blue has been widely employed to stain the proteins in-gel for many years. The main benefit of this stain is economical, and the equipment used is a simple densitometer, but the protein detection is poor about 30 ng. Hence, this methodology is being implemented to improve these aspects [10]. On the other hand, fluorescent staining is often preferred by its high sensitivity and dynamic range. Sypro Ruby<sup>®</sup> stain (Bio-Rad Laboratories) is the most widely used achieving a protein detection of 2 ng, but it is necessary the use of GelDoc XR System (Bio-Rad Laboratories) (Fig. 4) or similar at least.

2.4.1 SYPRO<sup>®</sup> Ruby Protein Stain According to the Instruction Manual Gels are stained with SYPRO Ruby<sup>®</sup> stain (Bio-Rad Laboratories), following the manufacturer indications, and digitalized using the GelDoc XR System (Bio-Rad Laboratories). The volume of

2.4 Staining of Gels and Comparison of Gel Images Using a Specific Software Package



Fig. 4 GelDoc XR System (Bio-Rad Laboratories)

solutions will be determined by box-sized considering that the gel must be immersed in the solution. Furthermore, a rocker is necessary for gentle mixing of the gel in the solutions.

1. Gel is rinsed in a fixed solution (ethanol 40% and acid acetic 10%) for 3 h.

- 2. Gel is washed with distilled water for 30 min three times.
- 3. In dark conditions, the gel is immersed in fluorescent stain overnight.
- 4. Finally, the gel is rinsed twice with distilled water in darkness for 30 min.
- 5. The gel is digitalized in GelDoc XR System (Bio-Rad Laboratories).

Afterwards, the 2-DE images can be analyzed using PDQuest 2-D Analysis Software (Bio-Rad Laboratories).

In Fig. 5 two examples of polyacrylamide gels (12%), employing strips of 24 cm in the pH range 4-7 stained with SYPRO Ruby and obtained from our lab are displayed. The p*I* and Mr. of spots

Protocol



Fig. 5 Representative 2-DE gel protein profiles of fresh meat from *Longissimus thoracis* of *Bos taurus* (a) and dry-cured ham of *Sus scrofa* (b)

	are determined from their position on the IEF-strips and standard molecular mass markers ranging from 15 to 200 kDa (Fermentas, Ontario) could be added, respectively.
2.5 Tryptic Digestion of Gel Spots	In a comparative proteomic analysis, the protein spots are chosen to be identified by mass spectrometry, thus, the protein spots are manually excised from the gel using a pipette tip or similar. <i>Solution</i> :
	• <i>Ambic buffer</i> : 50 mM ammonium bicarbonate in 50% methanol.
2.5.1 Protocol	1. The excised spots are cut into pieces and washed with Milli-Q water.
	2. The pieces are twice rinsed with Ambic buffer for 20 min after centrifugation (600 rpm), the supernatant is discarded.
	<ol> <li>The gel pieces are dehydrated with 100 μL acetonitrile for 10 min and subsequently dried in a vacuum centrifuge.</li> </ol>
	4. Following a reduction step with 10 mM DTT in Ambic buffer for 30 min at 56 °C after centrifugation (600 rpm), the super- natant is discarded.
	5. Repeat step 2.
	6. The alkylation step is produced by 55 mM iodoacetamide in Ambic buffer freshly prepared for 20 min in darkness.
	7. Repeat <b>step 2</b> .
	8. Modified porcine trypsin (Promega) is added to the dry gel pieces at a final concentration of 20 ng/ $\mu$ L in Ambic buffer incubating it at 37 °C for 16 h. Afterwards, centrifugation (600 rpm) is used to discard the supernatant.
	<ol> <li>Peptides are extracted thrice by 20 min incubation in 40 μL of 60% acetonitrile in 0.5% HCOOH under constant agitation.</li> </ol>

The supernatant is dried and stored at -20 °C for further analysis.

In the next and final main step, a MALDI-TOF MS is used to identify the proteins.

2.6 Identification of Matrix solution is prepared using 0.5 µL of peptides and 3 mg CHCA dissolved in 1 mL of 50% acetonitrile in 0.1% TFA. The Proteins by Mass final solution is placed on the Opti-TOF MALDI plate (Applied Spectrometry Biosystems). The mass spectrometry was carried out using a 4800 MALDI-TOF/TOF analyzer (Applied Biosystems) by reflectron positive-ion mode with a Nd:YAG, 355 nm wavelength laser, averaging 1000 laser shots and at least three trypsin autolysis peaks used as internal calibration. For tandem mass spectrometry (MS/MS), the data are performed by resolution of 300 (FWHM) and metastable suppression. GPS Explorer Software v3.6 using Mascot software v2.1. (Matrix Science) identify the proteins by peptide mass fingerprinting (PMF) and peptide fragmentation spectra data in a protein database. The parameters most used are 30 ppm precursor tolerance, 0.35 Da MSMS fragment tolerance, CAM (carbamidomethyl cysteine) as fixed modification, oxidized methionine as variable modification and allowing 1 missed cleavage. Finally, only the protein identifications are considered in the case of the score are greater than 56 and when protein score CI (Confidence Interval) was above 98%. For MS/MS, the total ion score CI should achieve above 95%.

#### References

- Li X, Zhang D, Ren C et al (2021) Effects of protein posttranslational modifications on meat quality: a review. Compr Rev Food Sci Food Saf 20:289–331
- 2. Hughes JM, Oiseth SK, Purslow PP et al (2014) A structural approach to understanding the interactions between colour, water-holding capacity and tenderness. Meat Sci 98:520–532
- 3. Estévez M (2011) Protein carbonyls in meat systems: a review. Meat Sci 89:259–279
- 4. López-Pedrouso M, Pérez-Santaescolástica C, Franco D et al (2019) Molecular insight into taste and aroma of sliced dry-cured ham induced by protein degradation undergone high-pressure conditions. Food Res Int 122: 635–642
- Berardo A, Claeys E, Vossen E et al (2015) Protein oxidation affects proteolysis in a meat model system. Meat Sci 106:78–84

- 6. Xing L, Liu R, Cao S et al (2019) Meat protein based bioactive peptides and their potential functional activity: a review. Int J Food Sci Technol 54:1956–1966
- 7. Picard B, Lefèvre F, Lebret B (2012) Meat and fish flesh quality improvement with proteomic applications. Anim Front 2:18–25
- 8. Franco D, Mato A, Salgado FJ et al (2015) Tackling proteome changes in the longissimus thoracis bovine muscle in response to pre-slaughter stress. J Proteome 122:73–85
- Kumar G (2012) Principle and method of silver staining of proteins separated by sodium dodecyl sulfate-polyacrylamide gel electrophoresis. In: Kurien BT, Scofield RH (eds) Protein gel detection and imaging. Humana Press, pp 451–464
- Gauci VJ, Padula MP, Coorssen JR (2013) Coomassie blue staining for high sensitivity gel-based proteomics. J Proteome 90:96–106



# **Chapter 14**

## **Antioxidant Capacity**

## Noemí Echegaray, Roberto Bermúdez, Gema Nieto, Rubén Domínguez, Mirian Pateiro, Nestor Sepúlveda, Marco Antonio Trindade, and José Manuel Lorenzo

#### Abstract

The requirements of the meat industry have made the determination of antioxidant capacity acquire special interest today, because these tests help to know the antioxidant status of meat and meat products. For this reason, the analysis of the antioxidant capacity represents a useful tool when studying the influence of diverse procedures in meat production (from animal husbandry to meat products elaboration), since together with other techniques it allows the election of procedures that favor the extension of the shelf life of these foods. However, despite this importance, the antioxidant capacity tests lack adequate standardization, because even using the same assays and fundamentals, different conditions and standards can be employed by researchers. This fact makes it very difficult to compare scientific results. On the other hand, the determination of the antioxidant capacity has been relatively little studied in products of animal origin, making it even more difficult to standardize the methods for a matrix as complex as meat. Taken together, these facts can hinder the advances pursued by the meat industry, generating economic losses in the sector.

Therefore, this book chapter aims to share in detail both the procedure of a general extraction method of antioxidant compounds and the methodology of three of the most used antioxidant capacity assays (FRAP, DPPH, and ABTS), in order to facilitate the comparison of outcomes.

Key words Antioxidants compounds, Antioxidant assays, FRAP, DPPH, ABTS, Absorbance, Spectrophotometric methods

#### 1 Introduction

At present, the determination of the antioxidant capacity plays a considerable role in meat and meat products due to the importance that oxidative deterioration processes have in these foods [1, 2]. In this sense, the complementary use of the traditional oxidation analyses and the antioxidant capacity techniques is of special interest because these last tests allow to know the antioxidant status of meat and meat products [3]. At the same time, the antioxidant capacity assays permit to implement enhancements in the meat industry based on the improvement of the oxidative stability of

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products,

Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1\_14,

<sup>©</sup> The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

these foods through the study of different strategies such as the enrichment in antioxidant substances using certain animal diets [4] and animal breeds [5] or employing particular antioxidant compounds in the preparation of meat-based products [6].

Given that oxidative stability improvement strategies are one of the main approaches addressed in the meat industry with the aim of avoiding the oxidation processes and extending the shelf life of meat and meat products, it seems clear that the determination of antioxidant capacity is an analysis that can be of crucial importance. On this matter, there are different methodologies that allow to know the antioxidant status of foods [7]. However, these analytical techniques are currently far from being standardized. In addition to this disadvantage, the determination of the antioxidant capacity in foods of animal origin has been studied to a lesser extent compared to other foods [8, 9], which makes their suitable application and utility in meat and meat products more difficult. For this reason, the methodology proposed in this book chapter tries to provide analysts with a unifying tool, explaining the possible considerations and differentiations to take into account when performing antioxidant capacity assays in these foods.

Prior to determining the antioxidant capacity of meat and meat products, an extraction of the antioxidant substances present in these products must be carried out. Currently, there is a great variety of methodologies for this purpose in terms of the steps and solvents employed. However, all processes are usually based on a solid-liquid extraction, the main difference being the solvent utilized that affects the extracted compounds [10-13]. Furthermore, the extraction of antioxidant compounds of meat and meatbased products can include some additional step that helps to promote the release of these substances. The present chapter describes a feasible and easy extraction technique with the use of a reagent commonly employed in the laboratory (80% methanol) together with the application of ultrasounds [4]. This simple technology favors the liberation of the antioxidant compounds due to the elongation of the pores and rupture of the cells.

Once the extraction process of the antioxidant compounds has been carried out, the determinations of the antioxidant capacity can be performed in the meat extract obtained. Although there are different methods, the most used in meat and meat products are the ferric ion reducing antioxidant power (FRAP) test, 2,2-diphenyl-1-picrylhydrazyl radical (DPPH) scavenging assay, and 2,2-'-azinobis-(3-ethylbenzothiazoline-6-sulfonic acid) radical cation (ABTS) scavenging probe due to their easy implementation [3]. These three methods are spectrophotometric tests based on the reactions between a colored complex or free radical with the antioxidant molecules capable of reducing an oxidizing substance and/or donating a hydrogen atom. Concretely, the FRAP technique is a typical electron transfer-based method because this test

measures the ability of antioxidants to reduce ferric ion  $(Fe^{3+})$  to ferrous ion (Fe<sup>2+</sup>) employing the ferric 2,4,6-tripyridyl-s-triazine complex  $[Fe^{3+} - (TPTZ)_2]^{3+}$  present in the FRAP reagent [7]. Thus, the colorless ferric complex  $[Fe^{3+} - (TPTZ)_2]^{3+}$  can be reduced in the presence of antioxidant compounds to the navy blue colored ferrous complex  $[Fe^{2+} - (TPTZ)_2]^{2+}$ , which has a maximum of absorbance at 593 nm. In this way, the measurement of the increase in absorbance at this wavelength can be carried out by an UV/Vis spectrophotometer, which allows to monitor the formation of the  $[Fe^{2+} - (TPTZ)_2]^{2+}$  complex and relates this value to the total reducing capacity of electron-donating antioxidants present in the meat sample extract [14]. On the other hand, the DPPH and ABTS methods are based on the combination of electron transfer and hydrogen donation reactions, being known as mixed-based methods, although the dominant reactions in the case of the DPPH assay are the electron transfer reactions [15]. Specifically, the DPPH test is based on the neutralization of the 2,2-diphenyl-1-picrylhydrazyl free radical, which has a violet coloration at 515-528 nm that is lost when contacting with an antioxidant substance. This loss of coloration can be related to the presence of antioxidants in the meat extract by monitoring the absorbance spectrophotometrically at 515 nm, since this parameter depends linearly on the concentration of antioxidants compounds [16, 17]. Similarly, the ABTS assay is based on the neutralization of the ABTS free radical, which has a blue-green color with a maximum absorbance at 730 nm. This test allows to measure the ability of antioxidants to reduce the ABTS radical (previously generated), which can be monitored by loss of blue-green coloration through the measurement of absorbance at 730 nm in an UV/Vis spectrophotometer [18].

With all that previously reported, the intention of this book chapter is to serve as a guide for the determination of the antioxidant capacity in meat and meat products (including the initial extraction step), thus allowing the standardization of the existing tests to favor their use in the laboratory and the correct comparisons in the analysis of these foods for the benefit of the meat industry.

#### 2 Materials

Prepare all solutions using distilled water and/or analytical grade reagents. Prepare and store all reagents at room temperature (unless indicated otherwise). Volumetric flasks were always used to adjust the volume of the reagents and standard solutions (unless indicated otherwise). When samples or reagents must be protected from light during their preparation and storage, they are protected by covering them with aluminum or using opaque material.

Diligently follow all waste disposal regulations when disposing of waste materials (*see* **Note 1**). 2.1 Sample 1. Professional mill KN 295 Knifetec (Foss). Preparation for 2. Homogeniser HM 294 (Foss). Analysis 3. Cutting board. 4. Knife. 2.2 Extraction of 1. Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius). Antioxidant Compounds 2. Polypropylene tubes of 20 and 40 mL. 3. Tube rack. 4. Borosilicate glass graduated pipette of  $25 \pm 0.1$  mL. 5. Vortex stirrer (see Note 2). 6. Rocking shaker, mod. SW-3D-E (OVAN). 7. Ultrasound water bath apparatus, mod. 8510E-DTH (Bransonic<sup>®</sup>). 8. Centrifuge, mod. Allegra™ X-22R (Beckman Coulter). 9. Qualitative filter paper (*see* Note 3). 10. Aluminum foil (see Note 4). 11. 80% Methanol (v/v): Mix 800 mL of pure methanol with 200 mL of distiller water. 2.3 Ferric Ion 1. Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius). Reducing Antioxidant Power (FRAP) Assay 2. Precision balance with a resolution of 0.01 g, mod. TE 612 (Sartorius). 3. Volumetric flasks and beakers of borosilicate glass. 4. Aluminum foil (see Note 4). 5. Microtubes of 1.5 mL. 6. Opaque rack for microtubes (*see* **Note 5**). 7. Magnetic stirrer bars and magnetic stirrer plate. 8. Calibration Check pH Bench Meter, mod. HI 221 (Hanna). 9. Temperature controlled water bath, mod. 6001197 (JP Selecta<sup>™</sup>). 10. Eppendorf Research Micropipettes of 10-100, 20-200,

- 10. Eppendorf Research Micropipettes of 10-100, 20-200100-1000, and 500-5000 µL.
- 11. Quartzglass cuvettes of 1000  $\mu$ L (Hellma Analytics).
- 12. UV/Vis spectrophotometer, mod. UV-1800 (Shimadzu).
- 13. Buffer solution pH 7.00  $\pm$  0.02.

- 14. Buffer solution pH 4.00  $\pm$  0.02.
- 15. 40 mM hydrochloric acid: Mix 331  $\mu$ L of 37% hydrochloric acid with distilled water up to 100 mL.
- 16. FRAP reagent: Mix 100 mL of 0.3 M acetate buffer solution (pH 3.6) with 10 mL of 10 mM 2, 4, 6-tri(2-pyridyl)-1,3,5triazine (TPTZ) solution, and 10 mL of 20 mM ferric chloride solution under darkness (*see* Note 6).
  - (a) 0.03 M acetate buffer solution (pH 3.6): Dissolve 2.46 g of sodium acetate with an approximate volume of 70 mL of distilled water in a 100 mL beaker. Adjust the pH to 3.6 using glacial acetic acid (with the help of a benchtop pH meter, previously calibrated with buffers 4 and 7). After adjusting the pH to 3.6, bring the volume to 100 mL.
  - (b) 10 mM 2, 4, 6-tri(2-pyridyl)-1,3,5-triazine (TPTZ) solution: Dissolve 0.0312 g of 2, 4, 6-tri(2-pyridyl)-1,3,5-triazine in 10 mL of 40 mM hydrochloric acid (measured with a 500–5000  $\mu$ L micropipette).
  - (c) 20 mM ferric chloride solution: Dissolve 0.0540 g of ferric chloride in 10 mL of distilled water (measured with a  $500-5000 \ \mu L$  micropipette).
- 17. Ferrous sulfate stock solution (5 mM): Dissolve 0.0139 g of ferrous sulfate heptahydrate in 10 mL of 40 mM hydrochloric acid (*see* **Note 6**).
- 1. Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius).
- 2. Borosilicate glass weighing funnels (see Note 7).
- 3. Volumetric flasks and beakers of borosilicate glass.
- 4. Aluminum foil (see Note 4).
- 5. Opaque glass test tubes of 12 mL (see Note 4).
- 6. Metal tube rack.
- 7. Magnetic stirrer plate and magnetic stirrer bars.
- Temperature controlled water bath, mod. 6001197 (JP Selecta<sup>™</sup>).
- 9. Eppendorf Research Micropipettes of 20–200, 100–1000 and 500–5000  $\mu L.$
- 10. Quartzglass cuvettes of 3 mL (Hellema Analytics).
- 11. UV/Vis spectrophotometer, mod. UV-1800 (Shimadzu).
- 12. 2,2-Diphenyl-1-picrylhydrazyl reagent (60 μM): Dissolve 0.0059 g of 2,2-diphenyl-1-picrylhydrazyl in 250 mL of pure methanol under darkness (*see* **Notes 6–8**).

2.4 2,2-Diphenyl-1-Picrylhydrazyl Radical (DPPH) Scavenging Assay 2.5 2,2'-Azinobis-(3-Ethylbenzothiazoline-6-Sulfonic Acid) Radical Cation (ABTS) Scavenging Assay

#### Trolox stock solution (1.2 mM): Dissolve 0.0030 g of trolox in 100 mL of pure methanol protected from light using a volumetric flask (*see* Notes 6, 7 and 9).

- 1. Analytical balance, with a resolution of 0.0001 g, mod. ME 614S (Sartorius).
  - 2. Borosilicate glass weighing funnels (see Note 7).
  - 3. Volumetric flasks and beakers of borosilicate glass.
  - 4. Aluminum foil (see Note 4).
  - 5. Microtubes of 1.5 mL.
  - 6. Opaque rack for microtubes.
  - 7. Magnetic stirrer bars and magnetic stirrer plate.
  - 8. Eppendorf Research Micropipette of 10–100, 20–200, and 100–1000  $\mu L.$
  - 9. Quartzglass cuvettes of 1000 µL (Hellema Analytics).
- 10. UV/Vis spectrophotometer, mod. 115 VAC (Shimadzu).
- 11. 2,2'-Azinobis-(3-ethylbenzothiazoline-6-sulfonic acid) reagent: Generated from the dilution with distilled water of the initial mixture of 2,2'-azinobis-(3-ethylbenzothiazoline-6-sulfonic acid) diammonium salt and potassium persulfate (initial concentrations for ABTS and potassium persulfate: 7 and 2.45 mM, respectively) (*see* Notes 6 and 10).
- 12. Ascorbic acid stock solution (1.7 mM): Dissolve 0.0030 g of ascorbic acid in 100 mL of distilled water under darkness (*see* **Notes 6** and **11**).

#### 3 Methods

Carry out all procedures at room temperature unless otherwise specified.

3.1 Sample The first step that we must take is to mince the whole sample and homogenize it correctly [19]. Firstly, we must cut the sample into small blocks and then, mince and homogenize them. Both minced and homogenization are carried out with a professional mill KN 295 Knifetec (Foss) or a Homogeniser HM 294 (Foss) depending on the type/size of the sample. This step is very important to achieve a good and representative result.
3.2 Extraction of Prior to the determination of the meat antioxidant capacity, an

# 3.2 Extraction ofPrior to the determination of the meat antioxidant capacity, anAntioxidantextraction of the antioxidant compounds must be carried out. ForCompoundsthis, a solid-liquid extraction was performed following the proce-<br/>dure of Santos et al. [20], with modifications. Figure 1 shows in a



Fig. 1 Schematic representation of the steps of the antioxidant compounds extraction

simplified and schematic way the steps of the antioxidant's extraction.

- 1. Firstly, weigh  $3.0000 \pm 0.0100$  g of sample (*see* Note 12) in 40 mL polypropylene tubes protected from light.
- 2. Then, add 20 mL of 80% methanol (*see* **Notes 13** and **14**) to the polypropylene tube, close it, and homogenize for 10–20 s in a vortex-type stirrer (*see* **Note 2**).
- 3. After initial vortex homogenization, place tubes on a balance shaker in a rack at 50 rpm, for 15 min.
- 4. Next, take the tubes in the same rack in an ultrasound water bath for 15 min, applying the sonic mode (*see* **Note 15**).
- 5. Following that, you must centrifuge the tubes at  $10,500 \times g$  for 10 min, maintaining a temperature of  $4 \pm 1$  °C.
- 6. Subsequently, filter the supernatant through qualitative filter paper (*see* **Note 3**) on 20 mL polypropylene tubes under darkness (*see* **Note 16**).
- 7. Finally, store at refrigeration temperatures  $(3 \pm 1 \text{ °C})$  if your analysis is going to be shortly, or on the contrary, store frozen at temperatures below  $-30 \pm 1 \text{ °C}$ .

#### 3.3 Ferric Ion Reducing Antioxidant Power (FRAP) Assay

The FRAP method was carried out by spectrophotometry according to the procedure previously described by Benzie and Strain [14], with brief modifications. Figure 2 displays in a simplified and schematic way the steps of the FRAP determination.

- Initially, you must prepare in microtubes the seven dilutions of the calibration curve (ranging from 0 to 2 mM) from the 5 mM ferrous sulfate stock solution, under darkness (*see* Note 17). You can see the volumes of 5 mM ferrous stock solution and distilled water needed to prepare the seven points of the calibration curve in Table 1.
- 2. You must transfer 30  $\mu$ L of each point of the calibration curve of ferrous sulfate to a microtube, protected from light. Identically, pipet 30  $\mu$ L of the sample extract into its respective microtube (*see* Note 18).
- 3. Next, add 90  $\mu$ L of distilled water to each previous microtube.
- 4. Subsequently, you must add 900 µL of FRAP reagent.
- 5. Then, homogenize each microtube in a vortex.



Fig. 2 Schematic representation of the steps of the FRAP determination

## Table 1 Volumes required for the preparation of the ferrous sulfate calibration curve

Point	5 mM ferrous sulfate solution ( $\mu\text{L})$	Distilled water ( $\mu$ L)	Ferrous sulfate concentration ( $\mu \text{M}$ )
1	0	1000	0
2	25	975	125
3	50	950	250
4	100	900	500
5	200	800	1000
6	300	700	1500
7	400	600	2000

- 6. Afterwards, incubate all the microtubes in a water bath during 20 min at  $37 \pm 1$  °C using a rack and protecting the microtubes from light.
- 7. Finally, you must measure the absorbance of each point of the calibration curve and sample extract in the spectrophotometer, at a wavelength of 593 nm (*see* **Note 19**).
- 8. Expression of results:

The results of antioxidant capacity through the FRAP method can be expressed in  $\mu$ mol Fe<sup>2+</sup>/100 g of sample following the next steps:

(a) Calculation of the equation of the ferrous sulfate calibration curve (*see* **Note 20**), where the absorbance is related to the  $Fe^{2+}$  concentration as follows:

Absorbance = 
$$a \cdot [Fe^{2+}] + b$$

where: *a* is the slope value.  $[Fe^{2+}]$  is the known concentration of  $Fe^{2+}$  ( $\mu M$ ). *b* is the y-intercept value.

(b) Next, the absorbance value obtained for the sample extracts is substituted in the previous equation and the  $Fe^{2+}$  concentration is calculated in  $\mu M$  as follows:

$$\left[\mathrm{Fe}^{2+}\right] = \frac{\mathrm{Absorbance} - b}{a}$$

where: *a* is the slope value, previously calculated.  $[Fe^{2+}]$  is the concentration of Fe<sup>2+</sup> calculated for the sample extract ( $\mu$ M). *b* is the y-intercept value, previously calculated.

(c) Lastly, the results can be transformed into  $\mu$ mol Fe<sup>2+</sup>/ 100 g of sample following the next formula:

$$\frac{\mu \text{mol Fe}^{2+}}{100 \text{ g of sample}} = \left[\text{Fe}^{2+}\right] \cdot \frac{V}{\mathcal{J}} \cdot 100$$

where:  $[Fe^{2+}]$  is the concentration of  $Fe^{2+}$  ( $\mu$ M), previously calculated. *V* is the volume used in the prior extraction of antioxidant compounds (L).  $\mathcal{G}$  is the sample weight used in the extraction of antioxidant compounds (g).

The DPPH assay was performed by spectrophotometry following the method previously reported by Brand-Williams [21], with slight changes. Figure 3 displays in a simplified and schematic way the steps of the DPPH determination.

 Initially, you must prepare in test tubes the seven dilutions of the calibration curve (ranging from 0 to 1.2 mM) from the 1.2 mM trolox stock solution, protected from light (*see* Note 17). You can see the volumes of 1.2 mM trolox stock solution and methanol needed to prepare the seven points of the calibration curve in Table 2.

3.4 2,2-Diphenyl-1-Picrylhydrazyl Radical (DPPH) Scavenging Assay



Fig. 3 Schematic representation of the steps of the DPPH determination

Point	1.2 mM trolox solution (mL)	Methanol (mL)	Trolox concentration (mM)
1	0	4.000	0
2	1.000	3.000	0.30
3	2.000	2.000	0.60
4	2.500	1.500	0.75
5	3.000	1.000	0.90
6	3.500	0.500	1.05
7	4.000	0	1.20

# Table 2Volumes required for the preparation of the trolox calibration curve

- 2. You must pipet 100  $\mu$ L of each point of the calibration curve of trolox in its respective test tube, under darkness. Identically, transfer 100  $\mu$ L of the sample extract into a test tube (*see* **Note 18**).
- 3. Next, you must add 3900  $\mu L$  of DPPH to each previous test tube.
- 4. Then, homogenize the contents of the test tubes in a vortex.
- 5. Subsequently, incubate all the microtubes in a water bath at  $37 \pm 1$  °C for 10 min, employing a rack and protecting the microtubes from light.
- 6. After the incubation, you must measure the absorbance of the DPPH reagent and the absorbance of each point of the calibration curve and sample extract in the spectrophotometer, at a wavelength of 515 nm (*see* **Note 21**).
- 7. Expression of results:

The results of antioxidant capacity through the DPPH assay can be expressed in  $\mu$ g trolox/g of sample following the next steps:

(a) Calculation of the inhibition percentage of the calibration curve and the sample extract by the following formula:

Inhibition (%) = 
$$\frac{\text{Abs(blank)} - \text{Abs(calibration curve/sample extract)}}{\text{Abs(blank)}} \cdot 100$$

where: Abs(blank) is the absorbance at 515 nm of the DPPH reagent. Abs(calibration curve/sample extract) is the absorbance at 515 nm of points 1–7 of the calibration curve (Table 2) and of the sample extracts, respectively.

(b) Calculation of the equation of the trolox calibration curve (*see* **Note 20**), where the percentage of inhibition calculated in the previous step is related with the trolox concentration as follows:

Inhibition (%) =  $a \cdot [\text{Trolox}] + b$ 

where: a is the slope value. [Trolox] is the known concentration of trolox (mM). b is the y-intercept value.

(c) Next, the percentage of inhibition calculated in the previous step for sample extracts is substituted in the trolox calibration curve equation and the trolox concentration is calculated in mM as follows:

$$[\text{Trolox}] = \frac{\text{Inhibition } (\%) - b}{\alpha}$$

where: a is the slope value, previously calculated. [Trolox] is the concentration of trolox calculated for the sample extract (mM). b is the y-intercept value, previously calculated.

(d) Then, the concentration of trolox must be changed from mM to  $\mu$ g/L, using the molecular mass of trolox as follows:

$$\frac{\mu g \text{ trolox}}{L} = mM \text{ trolox} \cdot PM \cdot 10^3$$

where: PM is the molecular weight of trolox (250.29 g/mol).

(e) Lastly, the results can be transformed into  $\mu$ g trolox/g of sample following the next formula:

$$\frac{\mu g \text{ trolox}}{g \text{ of sample}} = [\text{Trolox}] \cdot \frac{V}{\mathscr{G}} \cdot 100$$

where: [Trolox] is the concentration of trolox ( $\mu$ g/L), previously calculated. *V* is the volume used in the prior extraction of antioxidant compounds (L). *g* is the sample weight used in the extraction of antioxidant compounds (g).



Calibration curve (ascorbic acid)



Table 3								
Volumes	required	for the	preparation	of the	ascorbic	acid	calibration	curve

Point	1.7 mM ascorbic acid solution ( $\mu\text{L})$	Distilled water ( $\mu$ L)	Ascorbic acid concentration ( $\mu$ M)
1	0	1000	0
2	100	900	170.34
3	200	800	340.68
4	300	700	511.02
5	400	600	681.35
6	500	500	851.69

3.5 2,2'-Azinobis-(3-Ethylbenzothiazoline-6-Sulfonic Acid) Radical Cation (ABTS) Scavenging Assay The ABTS method was carried out by spectrophotometry following the procedure previously described by Re et al. [22], with some modifications. Figure 4 displays in a simplified and schematic way the steps of the ABTS assay.

- 1. Firstly, you must prepare in microtubes the six dilutions of the calibration curve (between 0 and 851.69  $\mu$ M) from the 1.7 mM ascorbic acid stock solution, protected from light (*see* **Note 17**). You can see the volumes of 1.7 mM ascorbic acid stock solution and distiller water needed to prepare the six points of the calibration curve in Table 3.
- 2. Then, you must transfer 20  $\mu$ L of each point of the calibration curve to a microtube, protected from light. Identically, pipet 20  $\mu$ L of the sample extract into its respective microtube (*see* Note 18).
- 3. Next, you must add 980  $\mu L$  of the ABTS solution to each previous microtube.
- 4. Immediately, homogenize the content in a vortex.
- 5. Then, incubate all the microtubes protected from light at room temperature (21  $\pm$  1  $^{\circ}{\rm C})$  during 10 min.

- 6. After the incubation, you must measure the absorbance of each point of the calibration curve and sample extract in the spectrophotometer, at a wavelength of 734 nm (*see* Note 22).
- 7. Expression of results:

The results of antioxidant capacity through the ABTS assay can be expressed in mg ascorbic acid/100 g of sample following the next steps:

(a) Calculation of the equation of the ascorbic acid calibration curve (*see* **Note 20**), where the absorbance is related to the ascorbic acid concentration as follows:

Absorbance =  $a \cdot [AA] + b$ 

where: a is the slope value. [AA] is the known concentration of ascorbic acid ( $\mu$ M). *b* is the y-intercept value.

(b) Next, the absorbance value obtained for the sample extracts is substituted in the previous equation and the ascorbic acid concentration is calculated in  $\mu$ M as follows:

$$[AA] = \frac{Absorbance - b}{a}$$

where: a is the slope value, previously calculated. [AA] is the concentration of ascorbic acid calculated for the sample extract ( $\mu$ M). *b* is the y-intercept value, previously calculated.

(c) Then, the concentration of ascorbic acid must be changed from  $\mu$ M to mg/L, using the molecular mass of ascorbic acid as follows:

$$\frac{\text{mg ascorbic acid}}{L} = \mu M \text{ ascorbic acic } \cdot \frac{PM}{10^3} \cdot 100$$

where: PM is the molecular weight of ascorbic acid (176.12 g/mol).

(d) Lastly, the results can be transformed into mg ascorbic acid/100 g of sample following the next formula:

$$\frac{\text{mg ascorbic acid}}{100 \text{ g of sample}} = [\text{AA}] \cdot \frac{V}{\mathscr{G}} \cdot 100$$

where: [AA] is the concentration of ascorbic acid (mg/L), previously calculated. V is the volume used in the prior extraction of antioxidant compounds (L). g is the sample weight used in the extraction of antioxidant compounds (g).

#### 4 Notes

- 1. All laboratory waste must be eliminated according to its nature following current regulations, differentiating at least acids, bases, organic solvents, halogenated solvents, and non-halogenated solvents.
- 2. Instead of vortex-type stirrers, dispersion devices can also be used in order to homogenize the sample.
- 3. Different filters can also be used at this stage according to sample needs.
- 4. Aluminum foil is not required if opaque material is used and vice versa.
- 5. The opaque rack will avoid the need to use aluminum foil or opaque microtubes, which can be more expensive.
- 6. The solution must be prepared daily and for each batch of samples to be determined.
- 7. Due to the very small mass that we have to use to prepare the solution, weighing must be performed in weighing funnels.
- 8. The DPPH reagent should be weighed on weighing funnels and then transferred to a beaker by dragging with methanol extra pure. After constant stirring until complete dissolution with a volume of methanol of approximately 220 mL, the volume of the solution will be adjusted to 250 mL in a volumetric flask and stored under darkness.
- 9. The trolox stock solution should be weighed on weighing funnels and then transferred to a beaker by dragging with methanol extra pure. After constant stirring until complete dissolution with a volume of methanol of approximately 80 mL, the volume of the standard solution will be adjusted to 100 mL in a volumetric flask and stored protected from light.
- 10. 0.0970 g of ABTS and 0.165 g of potassium persulfate are weighed separately on weighing funnels and transferred to a beaker by dragging with distilled water. After constant stirring until complete dissolution with a volume of distilled water of approximately 20 mL, the volume of the solution will be adjusted to 25 mL in a volumetric flask and stored protected from light during almost 16 h. After this period, the reagent is diluted with distilled water until its absorbance is adjusted to  $0.700 \pm 0.020$  at 734 nm (with a previous autozero made with distilled water). It is possible to start with an illustrative quantity of 540 µL of ABTS and 50 mL of distilled water and gradually add more ABTS or distilled water if the absorbance is less or greater than 0.700  $\pm 0.020$ , respectively.

- 11. The ascorbic acid stock solution should be weighed on weighing funnels and then transferred to a beaker by dragging with distilled water. After constant stirring until complete dissolution with a volume of distilled water of approximately 80 mL, the volume of the solution will be adjusted to 100 mL in a volumetric flask and stored under darkness.
- 12. Sometimes and because meat normally has low antioxidant capacity, instead of being used fresh, the sample is employed after being lyophilized, so that the antioxidant compounds are concentrated [12].
- The solvent used for extraction may vary according to the study in question, the usual solvents being ethanol, chloroform, phosphate buffer, and water [10–13], apart from 80% methanol [2].
- 14. It should be noted that different solvents will extract different antioxidant compounds.
- 15. Ultrasound treatment is done to help release antioxidant compounds from cells, increasing extraction performance [4, 10].
- 16. This liquid will be the extract of the sample that you will analyze through the different antioxidant capacity assays.
- 17. The calibration curve should be prepared immediately before adding the reagents, to avoid losing antioxidant capacity.
- 18. Depending on the antioxidant capacity of the sample extract, it may require prior dilution.
- 19. Before measuring absorbances at 593 nm, an autozero should be performed with distilled water.
- 20. The value of  $r^2$  obtained for the equation of the straight line must be greater than 0.900.
- 21. Before measuring absorbances at 515 nm, an autozero should be performed with methanol.
- 22. Before measuring absorbances at 734 nm, an autozero should be performed with distilled water.

#### Acknowledgments

This work was supported by Centro Tecnolóxico de la Carne de Galicia. Thanks to GAIN (Axencia Galega de Innovación) for supporting this book chapter (grant number IN607A2019/01). Authors are members of the Healthy Meat network, funded by CYTED (ref. 119RT0568). Noemí Echegaray acknowledges to Consellería de Cultura, Educación e Ordenación Universitaria (Xunta de Galicia) for granting with a predoctoral scholarship (Grant number IN606A-2018/002).

#### References

- Mukumbo FE, Descalzo AM, Collignan A et al (2020) Effect of Moringa oleifera leaf powder on drying kinetics, physico-chemical properties, ferric reducing antioxidant power, α-tocopherol, β-carotene, and lipid oxidation of dry pork sausages during processing and storage. J Food Process Preserv 44:e14300
- Ribeiro JS, Santos MJMC, Silva LKR et al (2019) Natural antioxidants used in meat products: a brief review. Meat Sci 148:181–188
- Böhm V, Müller L (2009) Methods to measure the antioxidant capacity of meat products. In: Nollet LM, Toldra F (eds) Handbook of processed meats and poultry analysis. CRC Press, Boca Raton, FL, pp 246–289
- 4. Echegaray N, Munekata PES, Centeno JA et al (2021) Total phenol content and antioxidant activity of different celta pig carcass locations as affected by the finishing diet (chestnuts or commercial feed). Antioxidants 10:5
- Lengkidworraphiphat P, Wongpoomchai R, Taya S et al (2020) Effect of genotypes on macronutrients and antioxidant capacity of chicken breast meat. Asian-Australasian J Anim Sci 33:1817–1823
- 6. Antonini E, Torri L, Piochi M et al (2019) Nutritional, antioxidant and sensory properties of functional beef burgers formulated with chia seeds and goji puree, before and after in vitro digestion. Meat Sci 161:108021
- 7. Gulcin İ (2020) Antioxidants and antioxidant methods: an updated overview. Arch Toxicol 94:651–715
- Capanoglu E, Kamiloglu S, Ozkan G et al (2018) Evaluation of antioxidant activity/ capacity measurement methods for food products. In: Apak R, Capanoglu E, Shahidi F (eds) Measurement of antioxidant activity and capacity: recent trends and applications. Wiley, Oxford, pp 273–286
- 9. Echegaray N, Pateiro M, Munekata PE et al (2021) Measurement of antioxidant capacity of meat and meat products: methods and applications. Molecules 26:3880
- 10. Simonetti A, Perna A, Gambacorta E (2019) Comparison of antioxidant compounds in pig meat from Italian autochthonous pig Suino Nero Lucano and a modern crossbred pig before and after cooking. Food Chem 292: 108–112
- 11. de Carvalho FAL, Lorenzo JM, Pateiro M et al (2019) Effect of guarana (*Paullinia cupana*) seed and pitanga (*Eugenia uniflora* L.) leaf extracts on lamb burgers with fat replacement

by chia oil emulsion during shelf life storage at 2 °C. Food Res Int 125:108554

- 12. de Carvalho FAL, Munekata PES, Lopes de Oliveira A et al (2020) Turmeric (*Curcuma longa* L.) extract on oxidative stability, physicochemical and sensory properties of fresh lamb sausage with fat replacement by tiger nut (*Cyperus esculentus* L.) oil. Food Res Int 136: 109487
- 13. Mancini S, Preziuso G, Dal Bosco A et al (2017) Modifications of fatty acids profile, lipid peroxidation and antioxidant capacity in raw and cooked rabbit burgers added with ginger. Meat Sci 133:151–158
- Benzie IFF, Strain JJ (1996) The ferric reducing ability of plasma (FRAP) as a measure of "antioxidant power": the FRAP assay. Anal Biochem 239:70–76
- 15. Nenadis N, Tsimidou MZ (2018) DPPH (2,2-di(4-tert-octylphenyl)-1-picrylhydrazyl) radical scavenging mixed-mode colorimetric assay(s). In: Apak R, Capanoglu E, Shahidi F (eds) Measurement of antioxidant activity and capacity: recent trends and applications. Wiley, Oxford, pp 141–164
- Alam MN, Bristi NJ, Rafiquzzaman M (2013) Review on in vivo and in vitro methods evaluation of antioxidant activity. Saudi Pharm J 21: 143–152
- 17. Kedare SB, Singh RP (2011) Genesis and development of DPPH method of antioxidant assay. J Food Sci Technol 48:412–422
- 18. Floegel A, Kim DO, Chung SJ et al (2011) Comparison of ABTS/DPPH assays to measure antioxidant capacity in popular antioxidant-rich US foods. J Food Compos Anal 24:1043–1048
- Presidencia del Gobierno (1979) Métodos oficiales de análisis de aceites y grasas, productos cárnicos, cereales y derivados, fertilizantes, productos fitosanitarios, productos lácteos, piensos, aguas y productos derivados de la uva. B.O.E. núm. 207 de 29 de Agosto de 1979
- 20. Santos J, Mendiola JA, Oliveira MBPP et al (2012) Sequential determination of fat- and water-soluble vitamins in green leafy vegetables during storage. J Chromatogr 1261:179–188
- Brand-Williams W, Cuvelier ME, Berset C (1995) Use of a free radical method to evaluate antioxidant activity. LWT-Food Sci Technol 28:25–30
- 22. Re R, Pellegrini N, Proteggente A et al (1999) Antioxidant activity applying an improved ABTS radical cation decolorization assay. Free Radic Biol Med 26:1231–1237

## NDEX

#### А

Amino acids	
Alanine	
Arginine	
Aspartic acid	
Cysteine	
essential amino acids	
Glutamic acid	
Glycine	
Histidine	
Hydroxyproline	
Proline	
Serine	
Taurine	
Threonine	
Tyrosine	
Amino acids derivation	
Amino acids extraction	
Ankom extractor	
Antioxidant capacity	153–155, 158, 161,
162, 165, 167	
Antioxidant capacity assays	
Antioxidants compounds	
Antioxidants extraction	
159, 161, 164, 165, 167	
Ash content	
2,2'-Azinobis-(3-ethylbenzothiazo	line-6-sulphonic acid)
radical cation scavenging pr	obe (ABTS)154,
158, 164–165	

#### В

Basic-acid methylation	
Biogenic amines	
Agmatine	
Cadaverine	
Histamine	
2-Phenylethylamine	
Putrescine	
Spermidine	
Spermine	
Tryptamine	
Tyramine	

#### С

Carbonyls	107–113
Chemical color	
Cholesterol	
Chromatographic techniques	42, 116
CIElab system	19, 21, 22
Colorimetric determination	
Colorimetric reaction	126 127
	120, 127

#### D

Derivatizing reagents	
Dienes	116–124
2,4-Dinitrophenylhydrazine	
2,2-Diphenyl-1-picrylhydrazyl radical scaver	nging assay
(DPPH)154, 155, 157, 1	61–163, 166

#### F

Fat content2, 7, 41, 50, 51,
119, 121
Fat extraction7, 8, 42, 43, 46–48, 119
Fatty acids
omega-3v
polyunsaturated fatty acids (PUFA)117
saturated fatty acids (SFA)2, 41
Fatty acids methylation47
Ferric ion reducing antioxidant power test
(FRAP)154–157, 160, 161
Food analysis108
Food safety
Free amino acids

#### G

Gas chromatography (GC)	
49, 54, 66, 98, 117, 127–130,	134, 136, 137
Gas chromatography-mass spectrometr	ry (GC-MS) 98,
117, 135, 136	

#### Н

Hardness	. 30,	36,	38
Head-space VOC extraction		1	34

José Manuel Lorenzo et al. (eds.), Methods to Assess the Quality of Meat Products, Methods and Protocols in Food Science, https://doi.org/10.1007/978-1-0716-2002-1, © The Author(s), under exclusive license to Springer Science+Business Media, LLC, part of Springer Nature 2022

# 170 METHODS TO ASSESS THE QUALITY OF MEAT PRODUCTS Index

Healthy meat	.16, 27, 40, 52, 63, 71,
82, 95, 105, 138, 167	
Heme pigments	
Hexanal	117, 127–131
Hydrolyzed amino acids extraction	on55, 56

#### L

Inductively coupled plasma optical emission	
spectroscopy (ICP-OES)	74–77
Ion chromatography	

### Κ

Kjeldahl method	5
-----------------	---

### L

Library match factor	
Lipid oxidation	
Lipids	2, 18, 42, 65, 66, 108, 118–121,
123, 128	
Liquid chromatograph	ny43, 54–56, 66, 67,
98, 100, 101	

#### М

Malondialdehyde (MDA)108, 109, 116, 117
Mass spectrometer identification 143, 149, 152
Mass spectrometry (MS) 49, 66, 75, 98, 117,
127 - 130, 134, 136, 138, 142 - 144, 148, 151, 152
Mass spectrum
Matrix-assisted laser/desorption ionization-time of flight
(MALDI-TOF) 143, 144, 152
Meat products1-3, 20, 21, 29, 30,
32, 34, 37, 38, 40–43, 54, 55, 61, 66, 67, 69,
73–75, 78, 81, 85–87, 91, 94, 98, 101, 103,
105, 115–117, 127, 128, 133–135, 142, 144,
153–155
Meat sciences
Mineral determination75, 77, 78
Minerals
calcium (Ca)74, 76-79, 81
copper (Cu)74, 76–79, 82
iron (Fe)74, 76–79,
118, 123
magnesium (Mg)74, 76–79, 81
manganese (Mn)74, 76–79, 82
phosphorous (P)74, 76–79
potassium (K)74–78
sodium (Na)74, 76-79, 81
zinc (Zn)74, 87
Moisture determination

#### Ν

Nitrate determination	87, 91–94
Nitrite determination	87–91
Nutritional determination	v, 54

#### 0

OPA derivatization	98,	100
--------------------	-----	-----

#### Ρ

Peroxide value (PV)	116–119
pH determination	
Physical color	
Primary lipid oxidation products	vi, 115–131
Protein determination	
Protein isoelectric point	
Protein oxidation	108, 111, 128, 142
Proteomics	
Proximate composition	1

#### Q

Quantifier ion	. 131,	139

#### R

#### S

Sample analysis
Sample calcination
Sample preparation3, 5, 21, 31-32, 117, 129,
156, 158
Saponification
Secondary lipid oxidation products vi, 115-131
Shear force
Sodium dodecyl sulphate polyacrylamide gel
electrophoresis (SDS-PAGE)147
Solid-phase microextraction (SPME) 117, 127,
129, 130, 134–136, 138
Spectrophotometric determinationvi

#### Т

Texture profile analysis (TPA)	
compression test	
penetration test	31, 32, 34, 37, 38
Texturometer	
Thiobarbituric acid reactive substanc	es (TBARS)117,
124–127	
Transesterification	
Trienes	
Tryptic digestion	143, 144, 151–152
Two-dimensional electrophoresis (2-	DE)142,
143, 150, 151	

#### V

Volatile compounds .	54, 117,	129, 130	,134–136,	138
Volatile organic comp	oounds (V	OC)		138

#### W

Warner-Bratzler (WB) test	30-32, 34, 35
Water holding capacity (WHC)	